

TRUCK A REVIEW AND ST

TRUCK TRAVEL SURVEYS: A REVIEW OF THE LITERATURE AND STATE-OF-THE-ART

by

Samuel W. Lau

JOSEPH P. BORT METROCENTER • 101 EIGHTH STREET • OAKLAND CALIFORNIA 94607-4700

TRUCK TRAVEL SURVEYS: A REVIEW OF THE LITERATURE AND STATE-OF-THE-ART

by

Samuel W. Lau

Planning Section Metropolitan Transportation Commission Joseph P. Bort MetroCenter 101 Eighth Street Oakland, CA 94607

January 1995

PREFACE AND ACKNOWLEDGEMENTS

This study was conducted for the Metropolitan Transportation Commission (MTC). The contents of this report reflect the opinions of the author who is responsible for the facts and accuracy of the information presented herein. The contents of this report do not necessarily reflect the official views or policies of MTC.

The author wishes to acknowledge the support and assistance of Charles Purvis, Senior Transportation Planner/Analyst, MTC, Planning Section, Chris Brittle, Manager, MTC, Planning Section, and Keith Mattson, Associate Transportation Planner/Analyst, MTC, Planning Section, for their time and valuable contributions to this report. Christine Harris, the MTC librarian, also spent countless hours searching for sources and documents.

The author would also like to recognize the following individuals from state departments of transportation, regional planning agencies, councils of government, and consulting firms, who have taken time to provide valuable information and written documents:

Keith Hall (Houston-Galveston Area Council) Gerald Rawling (Chicago Area Transportation Study) Cathy Arthur (Maricopa Association of Governments) Ron Taste (Port Authority of New York and New Jersey) Adiele Nwankwo (South East Michigan Council of Governments) Murray McLeod and Greg Little (Ontario Ministry of Transportation) Zac Graham (Texas Department of Transportation) Jack Fulcher (California Public Utilities Commission) Nancy McGuckin (Barton Aschman Associates, Inc.) Earl Ruiter (Cambridge Systematics, Inc.) Paul Roberts (Transmode Consultants) Marsha Anderson (Street Smarts) Dan Murray (American Trucking Association) Michael Arendes (Intermodal Association of North America) Clyde Woodle (Trucking Research Institute) David Pearson (Texas Transportation Institute) Daniel Blower (University of Michigan Transportation Research Institute)

Submit comments to:	Charles L. Purvis	OT	Samuel W. Lau samuell@ocf berkeley edu
	101 8th Street	01	Samuen Woer. Serverey.edu
	Oakland, CA 94607		
	chuckpurv@aol.com		

	Т	ABLE	C OF C	ONTE	NTS				
PREFACE AND ACKN	NOWLI	EDGE	MENT	S.	•	•	•	•	i
TABLE OF CONTENT	Γ.	•	•	•	•	•	•	•	ii
APPENDICES .	•	•	•	•	•	•	•	. •	vi
LIST OF TABLES	•	•	•	•	•	•	•	•	vii
LIST OF FIGURES	•	•	•	•	•	•	•	•	vii
EXECUTIVE SUMMA	RY	•	•	•	•	•	•	•	viii
Chapter One INTRODUCTION	•	•	•	•	•	•	•	•	1
TRUCK SURVEY LITERATURE	S & TR	UСК 1	TRAVE	EL DEN	MAND	FORE(CASTIN	IG	1
Chapter Two NATIONAL TRUCK J	RAVE	L SUR	VEYS	•	•	•	•	•	3
Truck Inventory an 1990 Nationwide T UMTRI National T	d Use S ruck Ac ruck Tr	urvey (ctivity a ip Info	(TIUS) and Cor rmatior	mmodi 1 Surve	ty Surv y (NT)	rey (NT FIS)	ACS)		3 3 4
FHWA Highway Si Chapter Three TRUCK SURVEYS AN	tatistics ND TRU	Report	ts . RAVE	L DEN	MAND	•	•	·	5
FORECASTING EXPI	ERIEN	CES IN	N MET	ROPC	DLITA	N ARE	AS.	•	7
TRUCK SURVEY FHWA Orig	MANU	JALS Destina	tion Su	rvey G	uide	•	•	•	8 8
Tri-State Tra Upstate New	ansporta v York I	tion Co Transpo	ommitte ortation	ee . Studie	s.	•	•	•	8 8
Indiana State	e Highw	vay Cor	mmissi	on.	•	、 •	•	•	9
CHICAGO AREA Survey Meth	TRAN	SPORT	CATIO	N STU	DY	•	•	•	9 10 11
Survey Data Survey Instra Truck Mode	ument l Output	• • t •	• • •	• •	• •	• •	• •	• •	12 13

1

- 1

ł

ſ

ſ

and services and

ſ

ii

.

PHOENIX	•	•		•	•		•	15
Survey Methodology	•			•		•	•	15
Survey Data .	•	•	•				•	16
Truck Models.	•		•		•			17
Study Findings	•	•	•	•	•	•	•	18
PORT AUTHORITY OF N	EW YO	JRK A	ND NE'	WJERS	EY			20
Truck Commodity Sur	vevs			, olice		•	•	20
Truck Cordon Surveys	3	•			•	•	•	24
EL PASO	•	•	•	•	•	•	•	26
NORTH CAROLINA	•	•	•	•	•	•	•	27
HOUSTON-GALVESTON	•	•	•	•	•	•	•	27
CALGARY	•	•	•	•	•	•	•	28
ONTARIO	•	•	•	•	•	•	•	29
Survey Methodology		•		•	•	•	•	30
Survey Data .	•	•	•	•	•	•	•	30
Data Limitations	•	•	•	•	•	•	•	30
Truck Volume by Tim	e-of-De	ay	•	• •	•	•	•	32
Truck Volume by Day	v-of-We	ek	•	•	•	•	•	32
Survey Findings	•	•	•	•	•	•	•	33
Survey Data Applicat	tions	•	•	•	•	•	•	34
VANCOUVER .		•		•		•		34
GVRD Truck Model	•	•	•	•	•	•	•	34
Chapter Four								
CALIFORNIA EXPERIENCE	S	•	•	•	•	•	•	38
STATEWIDE TRUCK STU	JDIES	•	•	•	•	•	•	38
LOS ANGELES AREA	•	•	•	•	•	•	•	40
SACRAMENTO AREA	•	•		•	•	•	•	45 ·
SAN FRANCISCO BAY A	REA			•		•	•	45
Bay Bridge Truck Tr	avel St	udy				•		45
Survey Method	lology	•		•		•	•	47
Survey Data	•	•	•	•	•	•	•	47
Study Findings	5	•	•	•		•		47

.

•

١

San Francisco Bay Area-Alan	neda Co	ounty J	ruck T	ravel Su	irvey	. 48
Survey Methodology .		•	•	•		. 48
Truck Travel Survey De	ata	-	•	•	•	. 49
Truck Model		• ·	•	•	•	. 50
Study Findings .		•	•	•	•	. 50
Chapter Five OTHER RESEARCH AND STUDIES TRAVEL CHARACTERISTICS	OF TR	UCK				. 52
		-	•	•	•	• • • •
Hourly Distribution of Internal True	k Trips	in Urb	an Area	as	•	. 52
Truck Trip Generation by Land Use	-	•	•	•	•	. 52
Methods for Estimating Truck Trip	Matrice	es				. 53
Passenger Vehicle Equivalencies (V	EQs)	•	•	•		. 54
Truck Travel Demand Forecasting a	nd Pav	ement]	Manage	ment	•	. 56
Chapter Six COMPUTERIZED DATA COLLECT TRAVEL INFORMATION	(ON O)	F TRU	CK •	•	•	. 58
Chapter Seven CONCLUSIONS AND RECOMMEND	ATIO	NS		•	•	. 60
THE NEED FOR TRUCK TRAVE	L DAT.	A	•	•	•	. 60
TRUCK SURVEYS & TRUCK TR LITERATURE	AVEL	DEMA	ND FC	RECA	STING	. 61
TYPES OF DATA COLLECTED .		•	•	•	•	. 61
USES OF TRUCK SURVEY DATA	4	•	•	•	•	. 62
TRUCK SURVEY METHODS US	ED	•	•	•	•	. 62
NUMBER OF COMPLETED SUR	VEYS.	AND F	ESPO	NSE RA	TE	. 63
SURVEY COST	1	•	• 、	•	•	. 63
COMPARISON OF SURVEY FINI	DINGS		•			. 64
Characteristics of Commercia	ıl Vehio	cles	•	•	•	. 64
Average Vehicle Weigh	ıt	•		•	•	. 64
Truck Size .	,	•	•	•	•	. 66
Characteristics of Commercia	ıl Vehio	cle Trip	os	•	•	. 66
Trips by Vehicle Type		•	•	•	•	. 66

1

н

1

1

[

L

 $\left[\right]$

ſ

[

Average Trip per Commercial V	<i>'ehicle</i>	•	•		66	
Regional vs. Through Trips .		•	•		66	
Average Trip Length			•	•	66	
Vehicle Miles Traveled .		•	•		66	
Time of First Commercial Vehic	cle Trip	•	•	•	67	
Time-of-Day Distribution .	•	•	•		68	
Truck Travel During Peak Perio	ods.	•	•	•	68	
Truck Travel During Peak Perio	ods as					
Percent of Total Vehicular Volu	ıme .	•	•	•	68	
Day-of-Week Distribution .	•	•	•	•	69	
Average Trip Time		•	•	•	69	
Truck Travel by Facility Type.	•	•		•	69	
Route Choice for Return Trips	•		•	•	69	
On-Street Stops	•		•	•	69	
Commercial Vehicle Trips and Land U	se .		•	•	70	
Trips by Land Use	•	•	•	•	70	
Facility of Trip Origin .		•	•	•	70	
Activities at Trip Ends .		•	•	•	70	
Other Truck-Related Findings .		•	•	•	70	
Truck Travel and Dangerous G	oods Ma	ovement	•	•	70	
Truck-Related Accidents .	•	•	•	•	71	
RECOMMENDATIONS	•	· •		•	71	
Survey Conduct	•	•		•	71	
Truck Travel Analysis	•	•	•	•	71	
BIBLIOGRAPHY		•	•	•	73	

~

-

_

•

APPENDICES

1

.

ſ

H

Appendix I:	NTACS Survey Form
Appendix II:	NTTIS Survey Form
Appendix III:	1986 CATS Commercial Vehicle Survey Form
Appendix IV:	1991 Phoenix Urban Truck Travel Survey Form
Appendix V:	1987 New York-New Jersey Truck Commodity Survey Form
Appendix VI:	1991 New York-New Jersey Truck Commodity Survey Form
Appendix VII:	1992 New York-New Jersey Truck Cordon Study Survey Form
Appendix VIII:	1994 El Paso Commercial Truck Travel Survey Form
Appendix IX:	1994 North Carolina-Triad Area Truck Survey Form
Appendix X:	1994 Houston-Galveston Area Council Commercial Vehicle
	Survey Form
Appendix XI:	1988 Ontario Commercial Vehicle Survey Form
Appendix XII:	1986 Caltrans Bay Bridge Truck Travel Survey Questionnaires
Appendix XIII:	1991 Caltrans-Alameda County Truck Intercept Survey &
	Classification Count Forms
Appendix XIV:	PDA Comparisons and Tandy Zoomer
Appendix XV:	Report Contact List

LIST OF TABLES

Table A-1:	Summary of Truck Travel Surveys in Urban Areas		•	•	xiii
Table A-2:	Summary of Truck Travel Survey Methods.	•	•	•	viv
Table A-3:	Types of Data Collected	•	•	•	xv
Table 3-1:	1986 CATS Commercial Vehicle Survey Sample S	ize Cor	nposite		
	Summary	•	•	•	12
Table 3-2:	Sample Rate by Day-of-Week		•		33
Table 4-1:	Comparison of Survey Methods for the Sacramento) Truck	Travel		
	and Goods Movement Study	•	•	•	46
Table 5-1:	Comparison of Commercial Vehicle Trip Rates	•	•		54
Table 7-1:	Summary of Truck Travel Surveys in Urban Areas	•	•	•	64
Table 7-2:	Summary of Truck Travel Survey Methods .	•	•	•	65
Table 7-3:	Types of Data Collected	•	•	•	66

LIST OF FIGURES

Figure 3-1:	Location of Port Authority Bridges and Tunnels	•		•	21
Figure 3-2:	1992 - 1994 Truck Cordon Survey Map .	•	•	•	25
Figure 3-3:	1988 Ontario Commercial Vehicle Survey Sites			•	31

EXECUTIVE SUMMARY

This paper presents the literature on conducting truck travel surveys in the U.S. and abroad. It includes past experiences, as well as current practices in conducting truck surveys. The primary purpose of this paper is to present compiled information on truck surveys and truck travel demand forecasting experiences to the Metropolitan Transportation Commission (MTC) to help evaluate the need for new truck/freight planning tools. The paper reports truck/freight survey experiences from Councils of Government (COGs), metropolitan planning organizations (MPOs), for which MTC is one, and other state and regional transportation planning agencies, both inside and outside of the country. The paper does not attempt to review literature on freight mode-choice surveys and/or modeling efforts (competition between rail and truck, for example).

THE NEED FOR TRUCK TRAVEL DATA

Truck travel and goods movement is essential to the economic vitality of an urban area. Trucks not only act as the "supply-line" from warehouses to points of consumption (whether they be retail stores or residences), they connect intermodal freight facilities (seaports, airports, railroad and freight terminal facilities). As important as truck travel is to a region, it also has negative effects such as traffic congestion, truck-related accidents, air pollution, noise, and pavement deterioration. In order for an urban area to undertake comprehensive truck/freight planning, accurate and reliable truck travel data is needed for analysis.

In terms of metropolitan area planning needs, improved truck travel data could be used for the following purposes:

- 1. Truck travel model development
 - Truck trip generation
 - Origin and destination analysis
 - Local and freeway route assignments
 - Congestion and speed simulations
 - Travel time analysis
 - Analyze impact of toll facilities
 - Spatial and temporal (time-of-day, day-of-week, and season) analyses
- 2. Corridor/Route analysis
 - Evaluate route/corridor traffic management proposals for freight impacts
 - Provide information on truck travel to formulate traffic management plans during roadway reconstructions
 - Assess impact of truck route reassignments or closures

- 3. Air quality modeling
 - Estimate truck emissions
- 4. Intermodal Freight Planning
 - Facilitate seaport planning
 - Facilitate airport planning
 - Understand competition and demand of different freight modes
 - Provide data to develop performance measures for Intermodal Management Systems as required under ISTEA
- 5. Pavement Management Planning
 - Evaluate and design road geometrics
 - Help calibrate pavement deterioration models
- 6. Truck restrictions and enforcement
 - Route restriction analysis
 - Dangerous goods movement regulation and enforcement analyses
 - Truck driver safety programs
- 7. Facilitate public-private partnership
 - Open dialog with private freight industries in gathering data
 - Provide truck travel data to public and the freight industry for research and analysis
 - Freight-economic analysis

TRUCK SURVEYS & TRUCK TRAVEL DEMAND FORECASTING LITERATURE

This literature and state-of-the-art review reveals that few urban areas in the country have had extensive experience in conducting truck surveys and truck travel demand forecasting. Most metropolitan planning organizations (MPOs) or regional transportation planning agencies continue to generate their truck trip estimates based on origin-destination surveys conducted in the 1960s and 70s. In the last ten years, only a few metropolitan areas, namely Chicago (1970 and 1986), Ontario (1978, 1983, and 1988), Vancouver (1988), Phoenix (1991), Alameda County, California (1991), New York-New Jersey (1991-94), El Paso, Texas (1994), and Houston-Galveston (1994) have undertaken significant efforts to collect truck travel data or develop new techniques in forecasting truck traffic. Out of the eight urban areas, only Chicago and Phoenix have had their truck model development and forecasting methodologies documented in detail, and only Ontario and the Port Authority of New York & New Jersey (PANYNJ) have systematically collected truck travel data. This report documents the experiences of different urban areas in the U.S. and Canada. The following is a summary of the results.

TYPES OF DATA COLLECTED

- The eight most recent truck travel surveys all collected origin-destination information (see **Table A-3**).
- With the exception of roadside surveys conducted in New York and New Jersey, most truck surveys requested land use at destination and truck odometer readings from respondents.
- Most surveys classified trucks by weight, number of axles, or by truck type.
- With the exception of roadside surveys, all other survey types included trip diaries (Chicago, Phoenix, El Paso, Houston-Galveston, and Alameda County).
- The commodity data collected ranged from a simple classification of commodity by type to detailed description of the actual commodites being carried.
- The 1988 Ontario survey is the only commercial vehicle survey that gathered information on truck driver characteristics.
- The 1994 El Paso commercial truck survey was the only survey that collected route choice information for the surveyed trip.

USES OF TRUCK SURVEY DATA

- The most common uses of truck data are for regional truck travel model development and corridor/route analysis. Chicago, Phoenix, El Paso, and Vancouver have used their truck survey data to develop regional truck travel demand models.
- Ontario has seen the most use of its truck survey information. The truck data have been used for time series comparisons, evaluation of road design and geometrics, pavement management planning, truck-related accident analysis, dangerous goods movement regulation and enforcement, understanding truck driver characteristics and for planning truck driver education programs.
- El Paso has mainly used its truck data for regional travel and truck emissions modeling.
- The Port Authority of New York and New Jersey has used its truck data for traffic management purposes during highway and bridge/tunnel reconstructions and freight-economic analysis.
- Chicago has used its truck data to generate truck activity maps of the Greater Chicago region; truck speed simulation; and modeling the effects of toll facilities on truck route choices within the context of the Chicago regional travel model.
- The Southern California Association of Governments (SCAG) has used its truck travel data to estimate heavy truck VMT and model truck emissions in the South Coast Air Basin (SCAB) for the Los Angeles area. It has also used truck data to conduct analysis of truck traffic to the Port of Hueneme in Ventura County.

- Caltrans and Alameda County has used its truck survey data to estimate truck traffic entering and leaving the County, as well as seaport planning for the Port of Oakland.

TRUCK TRAVEL SURVEY METHODS USED

- The most common survey method for conducting truck travel surveys in urban areas was the *combined telephone-mailout-mailback* method. Three urban areas in the country Phoenix, Arizona; Alameda County, California; and Houston-Galveston, Texas have recently conducted truck travel surveys using the combined telephone-mailout-mailback method.
- The combined telephone-mailout-mailback survey method is more costeffective and yields a reasonably high response rate.
- The second most used survey method was the *roadside interview* method. The Province of Ontario, Canada and the Port Authority of New York & New Jersey have conducted numerous roadside interviews.
- Roadside interviews produce very high response rates with complete information. They are ideal for cordon surveys or surveying trucks traveling in from outside the survey area.
- The most common source for drawing the survey sample is the Department of Motor Vehicle (DMV) registration files. Other sample sources include lists of truck registration files available from commercial vendors (R.L. Polk, Texas Vehicle Information and Computer Services, Inc., etc.).

A summary of different survey characteristics for eight urban truck travel surveys is found in Table A-1, and a summary of different truck travel survey methods (typical response rate, advantages, and disadvantages, etc.) is found in Table A-2.

NUMBER OF COMPLETED SURVEYS AND RESPONSE RATE

- The approximate number of completed surveys from the eight urban truck surveys varied from 188 to 19,225.
- Roadside surveys produced the highest number of completed surveys and the best response rate (nearly 100 percent).
- Mailout-mailback surveys produced the lowest overall and item response rates.
- Combined-telephone-mailout-mailback surveys produced improved response rates over mailout-mailback or telephone surveys alone (See Table A-1 and Table A-2).

Survey	Survey	Survey	Approx. Number of Completed	Response Rate	Data Applications	Total Survey Cost	\$/Survey
LUCATION			Surveys			,	
Chicago	1986	Mailout- Mailback	3,506	25.3%	 Truck travel model development Corridor/Route analysis Effects of toll on trucks Truck speed simulation model Truck activity mapping 	\$200,000	\$57/survey
Ontario	1988	Roadside Interview	19,225	96.5%	 Time series comparison Evaluate & design road geometrics Pavement management planning Truck accident analysis Dangerous goods regulation & enforcement analysis Truck driver characteristics Driver education program 	NA	NA
Phoenix	1991	Combined Telephone- Mailout- Mailback	720	30.0%	Truck travel model development	\$90,000(1)	\$125/survey(1)
N.Y. & N.J.	1991	Roadside Interview	4,500	NA	 Evaluate dedicated route/corridor proposal Traffic management for highway reconstruction Time series freight analysis Freight-economic analysis 	NA	NA
Alameda County, Calif.	1991	Combined Telephone- Mailout- Mailback & Roadside	2,200 over 8,000	79% NA	 I-880 corridor analysis Create truck travel submodel for corridor analysis Generate 24-hour & PM peak volumes by axle 	NA	NA
N.Y. &	1992-94	Roadside	14,671	37.8%(2)	NA	\$312,000(3)	\$21/survey
El Paso	1994	Telephone Interview	188	42.6%	 Truck travel model development Part of regional travel study Truck emissions analysis 	\$65,000(4)	\$345/survey(5)
Houston- Galveston	1994	Combined Telephone- Mailout- Mailback	900	35%-40%	Truck travel model development	\$150,000	\$167/survey

F 1 11 .

Metropolitan Transportation Council (NYMTC), and the Port Authority of New York and New Jersey. The survey was conducted at 18 locations with 3 interviewers per toll plaza for 24 hours. Cost included sample design, survey design, data collection, coding, reporting, survey analysis, and model development.

(4) (5) The higher cost was due to a high number of incomplete surveys.

	Table A-2: Summary of Truck Travel Survey Methods												
Survey Methods	Place (year) of Surveys	Typical Completed Surveys (% of Total Pop.)	Typical Response Rate	Advantages Disadvantages									
Telephone Interview	N.Y.(1964) Calgary(71) El Paso(94)	4%-15%	40%-50%	 High response rate Easy to follow-up Can only call during business hours "Phone-tagging" problem Limited time on phone is respondent is busy Requires access to vehicle registration file 									
Mailout- Mailback	Chicago(86)	1%-5%	10%- 45%(1)	 Less costly Good response rate w/ certified mail Only follow-up of non- responses is necessary Low overall & item response rate Possible bias due to bette response from some drivers/owners Low response from small truck owners Low response from out-of-state trucks Need to follow-up on non responses Difficult to ensure that the drive will fill out the form, instead o the owner or fleet manager who receives the survey forms Requires access to registration file 									
Combined Telephone -Mailout- Mailback	Phoenix(91) Houston(94) Alameda, CA(91)	3%-10%	30%- 80%(2)	 Improved response rate over mailout-mailback alone Early identification of owners who agree to participate & potential non-responses through phone contact Phone contact may help adjust sample size for mailout-mailback Same disadvantages as telephone survey method above High cost of telephone follow-up: Need phone reminders for trip diary More costly than above methods 									
Roadside Intercept/ Interview	Calgary(71) Ontario(78, 83, 88) N.Y. & N.J. (74, 82, 85, 91-94) Alameda County, CA (91)	8%-35%(3)	95%- 100%	 Complete information High response rate Better sampling control Good representative sample of trucks entering or leaving a cordon line Easy comparison with mainstream traffic through field counts at survey location Data Survey affected by weather, lighting Hazardous to survey crew Time constraint No follow-up possible Enforcement problems Drivers avoiding the survey station Only represent trucks traveling or road along survey station, no entire region 									

(2) The higher response rate was due to an employer survey conducted in California (1991 Caltrans-Alameda County Survey).
 (3) The higher percentage is from the 1988 Ontario survey which surveyed 57 locations over a 1,855-hour period throughout the Ontario Province.

			T٤	able A	-3:]	[ypes	of Da	ita Colle	cted				
Survey Location	Survey Year	Survey Method	Sample Source	Weight	Axle	Truck Type	O-D	Odometer	Commodity	Land Use	Driver Info	Route Info	Develop Model
Chicago	1986	Mailout- Mailback	DMV	٠		٠	•	•		•			Yes
Ontario	1988	Roadside Interview	Road- side(1)	•	•	٠	•	•	•		٠		
Phoenix	1991	Combined Telephone -Mailout- Mailback	DMV	•	•	•	•	•		•			Yes
N.Y. & N.J.	1991	Roadside Interview	Toll Plaza		•	•	•		•				
Alameda County, Calif.	1991	Combined Telephone -Mailout- Mailback & Roadside Interview	DMV, Port of Oakland Road- side(1)		•		•	•	•	•			Yes
N.Y. & N.J.	1992- 1994	Roadside Interview	Road- side(1)		•	•	•		•				
El Paso, Texas	1994	Telephone Interview	TVICS(2)		•	•	•	٠	•	•		•	Yes
Houston- Galveston	1994	Combined Telephone -Mailout- Mailback	DMV			•	•	•	•	•			

ž

SURVEY COST

- Telephone interviews are the most costly to conduct. They require a large number of staff and time for data collection.
- Combined telephone-mailout-mailback surveys are the most cost-effective to conduct. They yield reasonably high response rates over mailout-mailbacks alone. The phone contact portion of the survey can help assess non-response biases when analyzing and weighting the mailback survey samples.

COMPARISON OF SURVEY FINDINGS

A summary of the general findings from various truck travel surveys are as follows:

Characteristics of Commercial Vehicles

• Average Vehicle Weight: The only survey that reported average vehicle weight was the 1991 Phoenix Commercial Vehicle Survey. The average vehicle weight per commercial trip was 11,870 lbs.

• *Truck Size*: The share of different truck sizes used varied from urban area to urban area.

Characteristics of Commercial Vehicle Trips

- Average Trip per Commercial Vehicle: Light trucks have a higher average trip frequency than for heavy trucks.
- **Regional vs. Through Trips**: Most truck trips serve local regional needs. Of the few through trips (usually less than 10 percent), most are made by heavy trucks.
- Average Trip Length: Heavy trucks make longer trips than lighter trucks.
- *Vehicle Miles Traveled*: Heavy trucks log a higher VMT per day than light trucks.
- *Time of First Commercial Vehicle Trip*: Most "first" truck trips occur early in the morning (between 6:00 a.m. to 9:00 a.m.). This pattern, however, varies by weight category. Light trucks were more likely to start their first trip between 6:00 a.m. and 9:00 a.m. Heavy trucks, however, started their first trip before 6:00 a.m.
- *Time-of-Day Distribution*: Most truck trips seem to occur during the midday period between 9:00 a.m. and 3:00 p.m. Truck "through" traffic seems to avoid peak periods and tend to travel at night.
- *Truck Travel During Peak Periods*: The results vary by urban area and by individual locations. In New York and New Jersey, over 35 percent of trucks made trips during the morning peak period (6:00 a.m. to 10:00 a.m.). In comparison with AM and PM peaks for private vehicle travel, the results found that the AM peak period travel was as important for commercial vehicles as for private vehicles.
- Truck Travel During Peak Periods as Percent of Total Vehicular Volume: Truck traffic range from less than 9 percent to as high as 17 percent of the total vehicular volume during peak periods.
- *Day-of-Week Distribution*: Truck traffic typically occurs on weekdays and decreases significantly on the weekends.
- Average Trip Duration: Trip time generally increases with vehicle weight. The 1991 Phoenix survey recorded that the overall average trip time for truck travel was 28.1 minutes.
- Truck Travel by Facility Type: Few surveys or studies have attempted to analyze truck trips based on facility types used. Only the Canadians used facility types to classify their truck trips. A 1991 Barton Aschman Study of Alameda County truck trips found that many of the approximately 5,000 daily truck trips in the Port of Oakland area are local trips that never access a freeway.
- *Route Choice for Return Trips*: The only survey that analyzed route choice for return trips was the 1991 New York and New Jersey Truck Commodity Survey. It found that 73 percent of the truck drivers interviewed in the toll direction indicated that they would use the same route for the reverse trip.

• **On-Street Stops:** The 1991 Phoenix survey was the only to report the number of on-street stops made by trucks. The results found that over one-third of all commercial vehicles stops were made on-street. Light vehicles made half of their stops on-street.

Commercial Vehicle Trips and Land Use

- Trips by Land Use: Light trucks make more residential trips than any other truck category. Retail attracted many more light and medium truck trips. Heavy trucks dominated terminal/warehouse land uses.
- Activities at Trip Ends: Light trucks are heavily used for service delivery and personal business. Heavy trucks are most used for loading and unloading cargo at their trip ends.

Other Truck-Related Findings

- *Truck Travel and Dangerous Goods Movement*: The Ontario survey (1988) was the only survey that obtained information on dangerous goods movement. It found that a total of about 5 to 6 percent of all truck trips surveyed involved the carrying of dangerous goods. Flammable liquids (47 percent) were the most frequently transported dangerous goods, followed by compressed gases (24 percent), and corrosive substances (20 percent).
- *Truck-Related Accidents*: The 1988 Caltrans Urban Freeway Gridlock Study found that 5 to 10 percent of all truck-related incidents were found to cause major incidents which closed two or more freeway lanes for at least two hours.

RECOMMENDATIONS

This report recommends the following for conducting a regional truck travel survey and truck travel demand forecasting model if MTC should be interested in developing new truck data and tools:

Survey Conduct

- For internal-to-internal or internal-to-external truck trips, draw the survey sample from the DMV registration file or regional truck registration files (PUC, or private truck registration databases). Conduct either a telephone or mailout-mailback survey, or a combination of both to obtain a better response rate.
- For external-to-internal or external-to-external truck trips, conduct roadside intercept surveys at various roadway facilities and links in the network. The best places to conduct them are "weigh-in-motion" stations. This would minimize traffic delay for the mainline and would be safer for the survey crews compared to conducting the survey at the roadside.

- Consider conducting intercept surveys at bridge toll plazas. For a better explanation of how to conduct roadside surveys at toll plazas, review the experiences in New York and New Jersey.
- For roadside interviews or cordon surveys, conduct vehicle classification counts at the same time and at the same location where the actual survey/interview is conducted. This will provide the basic information for sample expansion and analysis.
- For obtaining trip diaries, using a combination of fleet-employer samples and truck unit samples is desirable. Sub-sampling fleet employers will provide better sample control and reduce the problem of oversampling large fleet operators.
- Oversample smaller or individual truck operators. The 1986 CATS survey has shown that large fleet operators tend to respond better (more manpower, time, or incentive to reply to surveys) and smaller operators tend to yield higher non-responses.
- To reduce the cost of conducting a full-scale truck survey, consider making the survey a multi-agency effort.
- Consider soliciting the help of private freight/trucking agencies or organizations. Open a dialog with interested parties to facilitate cooperation and to request assistance, especially in the design of the survey.

Truck Travel Analysis

- Time-of-day (24-hour), day-of-week, and seasonal variations in truck travel should be examined.
- Analyze trips by facility types used (include questions that obtain facility type information for each trip).
- Conduct further analysis on the impact trucks have on peak period congestion. Several surveys (New York, New Jersey, and Ontario) have found that in comparison with AM and PM peaks for private vehicle travel, AM peak period travel was as important for commercial vehicles as for private vehicles.
- Estimate total truck hours of delay by facility to help reduce truck operating cost.
- Conduct further analysis on the impact of truck traffic on pavement, especially the impact of waste-refuse trucks and buses (considered as "passenger-carrying trucks") on residential arterials and streets.
- The origins and destinations of trips that begin and end within the study area should be geocoded to the transportation analysis zones (TAZs) rather than at the city or zip code level. This would improve the accuracy of truck trip generation models based on zonal socioeconomic attributes.
- Exercise extreme caution when using or applying vehicle equivalency factors (VEQs) in truck travel analysis.

Chapter One INTRODUCTION

TRUCK SURVEYS & TRUCK TRAVEL DEMAND FORECASTING LITERATURE

To date, there has been limited experience in truck surveys and truck travel demand forecasting in urban areas. In most states and metropolitan areas, the collection of truck traffic data and development of truck travel demand forecasts are treated as an appendage to similar data collection and modeling efforts being done for passenger vehicles. There have been few efforts in treating truck data collection and forecasting as a separate issue. Moreover, a survey conducted by the Indian Nation Council of Governments (INCOG) has found that many MPOs continue to generate their truck trip estimates based on origin-destination surveys conducted in the 1960s and 70s (INCOG, 1990). Only a few metropolitan areas, namely Chicago (1970 and 1986), Ontario (1978, 1983, and 1988), Phoenix (1991), Vancouver (1988), Alameda County, California (1991), New York-New Jersey (1991-94), El Paso, Texas (1994), and Houston-Galveston (1994) have recently undertaken Herculean efforts to collect truck travel data or develop new techniques in forecasting truck travel. Moreover, recent technological advances, such as Weigh-in-Motion (WIM) count machines, Automated Vehicle Classification (AVC) devices, and Automated Vehicle Identification (AVI) systems, have made gathering truck data more accurate and less costly and cumbersome (National Cooperative Highway Research Program, 1988). Truck activity mapping, using geographic information systems (GIS), has also enhanced the ability to analyze truck and goods movement spatially. The cited studies varied in context, methodology and results. They reflect a limited but sizable literature from which to draw some lessons and conclusions.

This paper presents the literature on conducting truck travel surveys in the U.S. and abroad. It includes past experiences, as well as current practices in conducting truck surveys. The primary purpose of this paper is to present compiled information on truck surveys and truck travel demand forecasting experiences to the Metropolitan Transportation Commission (MTC). The objective of this paper is to present truck/freight survey experiences from Councils of Government (COGs), metropolitan planning organizations (MPOs), for which MTC is one, and other state and regional transportation planning agencies, both inside and outside of the country. The purpose, however, is not to present literature on freight mode-choice surveys and modeling efforts (competition between rail and truck, for example).

This report is organized into seven chapters. The following is a brief description of each chapter:

Chapter One is the introduction to the report. It presents a background and context to which the report is based.

1

Chapter Two gives a brief description of four recent national truck travel surveys and data collection efforts.

Chapter Three presents truck travel survey experiences in urban areas. It includes a description of truck manuals produced in the 1960s and early 70s. Truck travel survey experiences from Chicago, Phoenix, New York and New Jersey, Texas, North Carolina, Calgary, Ontario, and Vancouver are presented.

Chapter Four discusses truck travel data collection experiences in California. It includes a detail discussion of experiences from Los Angeles and the San Francisco Bay Area.

Chapter Five presents other research and studies of truck travel characteristcs. It includes topics such as hourly distribution of internal truck trips in urban areas, truck trip generation by land use, passenger vehicle equivalencies (VEQs), and truck travel demand forecasting and pavement management issues.

Chapter Six gives a brief discussion of computerized data collection techniques for gathering truck travel data. It presents the results of evaluating handheld computers for truck travel data collection.

Chapter Seven presents the conclusions from the analysis of the various surveys and studies mentioned in this report. It includes a list of recommendations for conducting future regional truck travel surveys.

Chapter Two NATIONAL TRUCK TRAVEL SURVEYS

There are various national efforts to collect large truck data. The purpose is to estimate truck miles traveled mainly for accident risk assessment and energy consumption estimations. Currently, there are four main national truck travel data sources available: the Truck Inventory and Use Survey (TIUS) conducted by the U.S. Bureau of the Census; the 1990 Nationwide Truck Activity and Commodity Survey (NTACS) conducted by the Bureau of the Census for the U.S. Department of Transportation; the National Truck Trip Information Survey (NTTIS) conducted by the University of Michigan Transportation Research Institute (UMTRI); and annual estimates published in *Highway Statistics* by the Federal Highway Administration (FHWA).

Truck Inventory and Use Survey (TIUS)

The Bureau of Census conducts the TIUS every 5 years as part of the Census of Transportation. The samples for the survey are drawn from the R.L. Polk vehicle registration files. The survey is conducted via questionaires mailed to a random sample of truck owners (Massie, Campbell, and Blower, 1993). The questionaires solicits information on the "typical" configuration and operation of all trucks over a 1-year period. Owners are asked to estimate the number of miles traveled with respect to the sampled vehicle, as well as information on the number of trailers usually hauled, type of cargo usually carried, typical weight of a load, etc. As a result, TIUS produces aggregate (state and national level) truck travel estimates and cannot be broken down to reflect operating environments or specific truck configurations. For the 1987 TIUS, a total of 104,606 trucks, including all truck weights²⁻¹, were sampled. Massie compared the 1982 and 1987 TIUS results with the NTTIS data (Massie *et al*, 1993). The comparisons covered power unit type, gross vehicle weight rating (GVWR) class, cab style, carrier type, and owner-reported annual mileage. Overall, Massie found that there was a good overall agreement between the two sets of survey data.

1990 Nationwide Truck Activity and Commodity Survey (NTACS)

The NTACS was a detailed annual and daily truck activity database for a sample of trucks covered in the 1987 TIUS. The data was collected at randomly selected days over a 12-month period ending in October 1990. The results of the survey were presented in a summary report (Oak Ridge National Laboratory, 1993). Unlike the TIUS which reported only annual data, the 1990 NTACS asked truck operators to report daily, as well as annual data. The purpose was to capture temporal and geographic variations in truck travel and other trucking characteristics beyond the scope of the TIUS. The 1990

²⁻¹ TIUS and NTTIS excludes ambulances, open utility vehicles, motor homes, buses, farm tractors, and government-owned vehicles.

NTACS sampled 44,002 trucks, a subsample of the approximately 105,000 trucks from the 1987 TIUS. The trucks were stratified by geographic division, types of haul, and truck classification. The survey was a mailout-mailback type survey (the survey instrument can be found in Appendix I). Respondents were asked to report trip activities for 2 selected days in a 4-week period. The survey response rate was around 77.9 percent, with 104,601 trucks responded, representing 9,794 operators. Because of initial low item response rate, two mail and one telephone follow-ups were conducted. Due to the high cost of telephone follow-ups, the use of certified mail responses was implemented. Certified mail proved to be the most cost effective method and improved not only response rate, but item non-response as well. The survey analysis included the estimated population of four year and older trucks, with an adjusted estimate which included trucks under four years old. It also included estimated annual miles traveled, daily miles traveled, and distribution by truck type. The survey excluded vehicles owned by Federal, State, or local governments, ambulances, buses, mobile homes, farm tractors, unpowered trailers, and trucks sold prior to 1987 and disposed of prior to July 1, 1986. The NTACS, however, suffered from high item nonresponse and data inconsistency problems. The report warns that it should be used with caution.

UMTRI National Truck Trip Information Survey (NTTIS)

From November 1985 to February 1987, the Center for National Truck Statistics of the University of Michigan Transportation Research Center (UMTRI) conducted a 15month National Truck Trip Information Survey (NTTIS) of medium and large trucks. The results were summarized in three separate TRB articles (Campbell, 1986; Massie et al., 1993; Massie et al., 1993). Like TIUS, the NTTIS drew its sample from the R.L. Polk vehicle registration files and obtained most of its information through telephone interviews with truck owners. Mail versions of the interview forms were used only if the interview could not be completed by phone (see Appendix II for sample of questionnaire). In contrast, however, NTTIS was based on actual trips made by truck drivers, not a characteristic of "typical" trips as in the case of TIUS. Only commercial vehicles with a gross vehicle weight rating (GVWR) greater than 10,000 lb. were sampled.²⁻² The sampling frame was stratified by state, and within each state, straight trucks were sampled separately from tractors. A total of 8,144 trucks were selected from the R.L. Polk registration list to form the survey sample. Out of the 8,144 trucks, 564 of the samples were determined to be unusabled. Of the resulting 7,580 cases, 6,305 interviews were completed, with a response rate of 83.2 percent and a refusal rate of 3 percent. NTTIS included truck information by road type, area type, and time-of-day. Moreover, information on truck configuration, number of trailers, carrier type, cab style, fuel type, cargo body style, cargo type and weight, number of truck axles, power unit, driver age and experience were all obtained from the survey. The data were organized into three separate data files: a truck file, a truck-tractor trip file, and a straight truck trip file (Blower and Pettis, 1988). The truck file contains information on the vehicle,

²⁻² TIUS and NTTIS excludes ambulances, open utility vehicles, motor homes, buses, farm tractors, and government-owned vehicles.

company (owner), and annual mileage, with one record per vehicle. The tractor and straight truck files contain trip information, one record per trip for each trip taken by the surveyed vehicle.

A sub-sample of 2,511 straight trucks and 2,501 tractors were drawn to collect detailed 24-hour truck travel information for a designated day. Phone calls were made, typically the day after the survey day, to the drivers or owners to obtain detailed trip information. The trips were traced on specially prepared maps based on the Rand McNally *Road Atlas* maps, and mileage was broken down by road type, rural/urban, and day/night. The overall response rate for the 24-hour truck trip survey was 93.7 percent (4,789 out of 5,112 truck surveyed, including vehicles not in-use). The response rate for vehicles in-use, however, were 88.2 percent for straight trucks and 84.6 percent for tractors.

The NTTIS data found that straight trucks outnumbered tractors in the national large-truck population by about 70 percent to 30 percent. The annual miles of travel, however, was nearly the reverse. Tractors logged in 68 percent of the total miles traveled and straight trucks only 32 percent. The survey also found that the average annual mileage of a tractor was 5 times that of a straight truck (41,176 miles compared to 8,231 miles). Trucks by configuration found that single tractors logged in 63 percent of the total annual mileage, straight trucks with no trailers 30 percent, and bobtails, doubles, and straight trucks pulling trailers together accounted for only 7 percent of the total large-truck travel. Truck travel by road type revealed that less than 20 percent of straight trucks to more than 72 percent of doubles traveled on limited access roads. Conversely, travel on arterials dropped from 42 percent for straight trucks to 20 percent for doubles. Truck travel by time-of-day showed that all truck configurations log in far more miles during the day than at night. The distribution, however, varied. Straight trucks made less than 3 percent of their trips at night, and doubles made less than 40 percent for the same period.

In summary, Massie concludes that truck travel varies by truck configuration, road class, time-of-day, and population area. Secondly, large trucks vary widely in their physical configuration, which in turn affects their risk of accident involvement. The authors state that the diversity of trucking operations underscores the importance of reliable truck travel data in any analysis that seeks to determine the relative safety of one class type versus another.

FHWA Highway Statistics Reports

Each year, the Federal Highway Administration (FHWA) compiles all state transportation statistics and publishes an annual national *Highway Statistics* report. The report categorizes truck travel by vehicle class and road type. The *Highway Statistics* report includes data on government-owned vehicles and commercial vehicles registered in Alaska and Hawaii. The data are provided by state DOTs and are mainly aggregate data. Mileage data submitted by the states are based on traffic counts of 13 vehicle classes on 12 selected types of road. Counts are often conducted manually or by automated vehicle counters. Massie cited that manual and automated counts are both problematic due to human counting error and defective automated detectors (Massie *et al*, 1993). Because trucks represent a small portion of overall traffic, counting errors can lead to large percentage errors in vehicle classification estimates. Second, in a recent discussion of *Highway Statistics* estimates for large trucks, Mingo cited several indications that the estimates are too high (Mingo, 1991). Third, state classification systems tend to vary. This makes it especially difficult for cross-referencing and comparisons. Fourth, state counts are conducted only on weekdays. They do not take into account day-of-week variations in truck travel. Overall, Massie concluded that the FHWA *Highway Statistics* systematically overestimates truck travel. Moreover, it is plagued by data collection and reporting errors.

Chapter Three TRUCK SURVEYS AND TRUCK TRAVEL DEMAND FORECASTING EXPERIENCES IN METROPOLITAN AREAS

During the 1960s, there was a major national effort to collect truck travel data through Federal sponsorship of roadside surveys. In the past two decades, however, Federal support has waned because of high data collection costs and safety problems associated with protecting survey crews during the data collection process. Consequently, commercial vehicle trip data have been downplayed in many metropolitan areas and have become less important and more difficult to obtain. Recent estimates, however, reveal that commercial vehicle trips account for 10 to 30 percent of the total trips, with many of the trips originating from or destined to outside the metropolitan areas (INCOG, 1990; Schlappi, Marshall, and Itamura, 1993). Furthermore, many of the heavier trucks have been found to account for over 30 percent of the total vehicle miles traveled (VMT) in metropolitan areas (Schlappi et al, 1993). Very little is known about the 10 to 30 percent of commercial trips traveling within a metropolitan area. Traditionally, truck trip estimations were merged with total vehicle trip estimations and simply represented as a certain percentage of the total trip estimates. The percent of truck trip estimates used for forecasting may be based on a rate derived from vehicle classification counts, or a percentage selected and based on a planner's best judgment or intuition. Only recently has there been a national trend to collect new truck trip data and develop new techniques for forecasting truck travel in metropolitan areas.

There are numerous papers and studies on urban goods and freight movement. Most are related to specific problems or issues, such as cost, congestion in downtown, inter-city or inter-state goods movement, etc. Few, however, deal with developing regional travel models to forecast truck traffic in metropolitan areas. During the 1960s and 70s, several truck surveys were conducted, and survey procedure manuals were produced with funding from the U.S. Department of Transportation, Bureau of Public Roads. The metropolitan areas that have conducted truck surveys in the 1960s and 70s were the Tri-State Transportation Committee of Connecticut (1964), New Jersey, and New York, the Upstate New York Transportation Studies (1964), the Indiana State Highway Commission (1970), the Chicago Area Transportation Study (1970), and the Bay Area Transportation Study Commission (1965). Unfortunately, few of the survey results have been published. Chicago and Phoenix seem to be in the forefront of metropolitan areas that have published their survey results and developed models and methodologies in forecasting truck travel. Outside of the U.S., Canada - Calgary (1971), Vancouver (1988), and Ontario (1978, 1983, and 1988) - and Australia (1977-78) has also conducted numerous commercial vehicle surveys.

TRUCK SURVEY MANUALS

FHWA Urban Origin-Destination Survey Guide

In 1944, the Bureau of Public Roads (now Federal Highway Administration) published the first procedural guide for conducting origin-destination surveys. Revised editions were published in 1946, 1954, and 1973. The latest revision *Urban Origin-Destination Surveys* was reprinted in 1975 (FHWA, 1975). Chapter 5 of the 1975 survey guide included guidelines on conducting internal truck and taxi surveys. It discussed sampling frame, sampling procedures, interviewing techniques, and some survey management issues. The guide included a sample of typical survey questionnaire. A sample set of instructions is included in the appendix. This document is merely a guide and is not intended to serve as a manual.

Tri-State Transportation Committee

The Tri-State Transportation Committee, in cooperation with the three States of New York, New Jersey and Connecticut, conducted a goods movement - external truck survey in 1963-64. The study involved studying the movement of people and goods within the 8,000 square mile Tri-State area. A truck survey procedure manual was produced (Tri-State Transportation Committee, 1964). The manual documented the data collection procedures for the survey interviewers. The truck survey involved conducting roadside interviews of trucks registered within the region, as well as commercial vehicles crossing the cordon line. Interview stations were selected at 22 routes. The types of information collected from the survey include commodity, weight, length-of-haul, and origin-destination information. Approximately 480 interview hours and at least 3,000 interviews were planned. During the roadside interview, a separate vehicle classification count was also conducted at the location of the interview station.

Upstate New York Transportation Studies

In 1964, the Upstate New York Transportation Studies produced a truck and taxi survey manual (Upstate New York Transportation Studies, 1964). The manual was written for telephone interviewers. The vehicle owners, drivers, or fleet dispatchers were contacted by phone and asked to participate in the survey. The interviewers were encouraged to directly survey the driver of the vehicle being sampled. The survey was conducted to collect trip information on origin and destination, beginning and ending times, trip purpose, land use at trip ends, volume and description of the cargo hauled, whether expressways were used, whether the vehicle crossed the screen line, and whether intermediate stops, if applicable to a particular trip, were made. The survey was conducted over a 3-month period.

Indiana State Highway Commission

In 1970, the Indiana State Highway Commission produced a truck-taxi survey manual for planning areas within the state (Indiana State Highway Commission, 1970). The manual included a sample ratio table based on the population of the study area to be surveyed. The manual recommended obtaining the sample population from the Bureau of Motor Vehicles. From this list, vehicles garaged outside of the study were to be eliminated and publicly owned vehicles included. In general, it recommended that for a study population of over 1 million, a minimum of 2 out of every 25 trucks (an 8 percent sample), should be surveyed. Each selected vehicle owner should be contacted by mail to inform them of the survey and to solicit cooperation. The next step is a face-to-face interview with the respondent. The preliminary interview would include the delivery of the trip log and a full explanation of how to complete it. Next, the actual interview is scheduled with the respondent in which the trip log is collected and reviewed with the respondent. The survey data collection would include vehicle type, classification, origin and destination information, trip beginning and ending times, cargo description, and approximate daily mileage.

CHICAGO AREA TRANSPORTATION STUDY

In the 1970s, the Chicago Area Transportation Study (CATS) began developing separate trip forecasting models for passenger vehicles and commercial vehicles. It was one of the first of such efforts in the country. There was a recognition that commercial vehicles had very different travel characteristics than passenger vehicles and that their impact on the urban roadways necessitated a separate approach and technique in forecasting future trips. In addition to the 1970 Origin-Destination Household Survey, a separate commercial vehicle survey was conducted to collect data on truck travel characteristics. The 1970 survey consisted of field interviews of owners or operators of vehicles drawn at random from the state's motor vehicle registration database. The respondents reported data on origin-destination, commodity and goods movement by weight. The data provided CATS with the ability to model passenger and commercial vehicle trips through the 1970s and the early 1980s. As a result of CATS's modeling efforts, two reports based on the 1970 data were published. One was written by Southworth and Chon (1981) and another by Southworth, Lee, Griffin, and Zavattero The former looked at vehicle emissions of trucks, and the latter was on (1983). truck/freight planning for the Chicago region. The Southworth and Chon report documented the methods used to estimate regionwide truck emissions in the Chicago area The analysis relied heavily on the 1970 CATS (Southworth and Chon, 1981). Commercial Vehicle Survey data to produce truck trip estimates for the emissions model input.

In 1986, CATS began to update their commercial vehicle data. The 1970 data on commercial vehicles were determined to be dated and did not represent commercial vehicle travel at the time. Changes in commercial vehicle activity, coupled with major

revisions to rules and regulations (Motor Carrier Act of 1980) convinced CATS that new data were needed. In the early 1980s, CATS experimented with statistical techniques to create an acceptable formula to indirectly estimate truck travel. The effort, to conserve time and resources, was unsuccessful. As a result, there was a recognition that new data was needed. However, the scale of resources committed to the 1970 surveys was not available. As a result, a more cost-conscious commercial vehicle survey was conducted in 1986 to provide updated truck data. The survey process and results are documented in two reports. The first report presents the details of administering the survey (Rawling and Reilly, 1987), while the second documents the results (Reilly, Rosenbluh, and Rawling, 1987).

Survey Methodology

The 1986 CATS Commercial Vehicle Survey was based on a random sample of registered commercial vehicles in the state Department of Motor Vehicles (DMV) database. Vehicle counts by license plate types were obtained with the cooperation of the Secretary of State's Office. The population to be surveyed was obtained from a random sample of seven counties by three license plate types: (1) weight plates, (2) mileage plates, and (3) International Registration Plan (IRP) plates. The total population for the 1986 Commercial Vehicle Survey was determined to be 359,383 vehicles. There were 305,183 weight and mileage plate vehicles and 50,400 IRP plate vehicles registered in northeast Illinois. An addition to the 1986 survey, not included in the 1970 survey, was the sampling of the utility vehicle fleet. This included a sizeable number of vehicles that had fixed assignments and routes and made fixed number of trips (postal service vehicles, gas company trucks, etc.) The 1986 CATS survey included, for the first time in a commercial vehicle survey in the U.S., school buses and U.S. Postal Service (USPS) vehicles. The school buses survey results were summarized in a separate report and will be discussed in the section on *methodological issues* (Rawling and Reilly, 1987). The USPS operated 3,800 vehicles in the northeast Illinois area, the largest fleet of any operator. The following groups of vehicles, however, were not included in the survey:

- M-plates (municipal and other local government fleets);
- U-plates (State of Illinois vehicle fleet);
- federal and military vehicles;
- taxis, commuter vans, dealership vehicles;
- ambulances and tow trucks;
- vehicles registered outside the six northeast Illinois counties.

The weight, mileage and IRP plate vehicles were grouped into the following weight categories:

(1) B license plates (vans, pick-ups and panel trucks of 8,000 lbs. or less of gross vehicle weight);

(2) Light vehicles (D to J, MD to MJ plates, 8,001-28,000 lbs. GVW);

(3) Medium vehicles (K to T, MK to MT plates, 28,001-64,000 lbs. GVW);

(4) Heavy vehicles (over 64,000 lbs GVW).

To provide a profile of trips generated and destined to different land uses, survey respondents were asked to provide land use data. The eleven land use categories coded from the survey results are:

- Residential
- Retail
- Manufacturing
- Terminal/Warehouse
- Public/Government
- Office/Service
- Construction
- In Transit
- Landfill
- Agricultural
- Other/Missing

The survey was administered in January and February of 1986. January 21, 1986 was the target date set for weight and mileage plates. February 25, 1986 was the target date for IRP plates. The survey spanned fifteen months and cost slightly over \$200,000.

Survey Data

The sample was weighted to ensure that each weight group was represented. The B weight group was sampled at 2 percent because B trucks represented 66.9 percent of the total commercial vehicle population. The *light, medium,* and *heavy* vehicles were sampled at 6 percent, and all IRP trucks (approximately 85 percent are heavy trucks) were sampled at 10 percent. This was to ensure a usable return of sample vehicles working in the area. For the USPS fleet, 129, or 3.4 percent of the 3,800 vehicle fleet, was sampled. An effort was made to find an equal distribution of USPS vehicles according to vehicle types and geographic areas for which the USPS fleet serves. After a month-long letter and telephone campaign to major IRP operators, CATS failed to reach 34 companies (operating 881 vehicles) and was refused by seven companies (operating 176 vehicles). A composite summary of the sample size distribution and return rates is shown in (**Table 3-1**).

The data was coded by trip frequency, distance, purpose, land use, etc. The trip end and the addresses were geocoded to the section (square mile) level of the range-township system and the 1,542 CATS zone system. Information for the USPS vehicle fleet was taken directly from the USPS operations management and converted into the standard trip diary format.

Table 3-1. 1986 CATS Commercial Vehicle Survey Sample Size Composite Summary											
Vehicle <u>Class</u>	Class <u>Total</u>	% <u>Total</u>	No. of <u>Samples</u>	% <u>Sampled</u>	Samples <u>Returned</u>	% <u>Returned</u>					
B trucks	240,600	66.9	4,748	2.0	310	6.5					
Light Medium Heavy	68,383	19.0	4,067	5.9	811	19.9					
IRP	5,038	14.0	5,038	10.0	2,256	44.8					
Total Source: CATS	359,383 1986 Commercial V	100.0 Vehicle Surve	13,853 y (Rawling and Re	3.9 illy. 1987)	3,377	24.4					

Prior to estimating the number of commercial trips for the study area, the CATS researchers felt it important to exclude, for travel estimating purposes, two groups of vehicles: (1) those solely in personal use, and (2) vehicles in commercial service but not in service on the survey date. First, B plate vehicles for personal use were removed from the analysis and assumed to have been represented in the home interviews as personal use vehicles. Second, the "working population" of commercial vehicles were identified for the analysis. "Working population" is defined as "vehicles reported to have no work on the survey date [including] those out of service, sold or for sale, in for repair, only used on site, in seasonal use, or registered to a company which was out of business" (Reilly, Rosenbluh and Rawling, 1987). The study was unclear, however, on whether vehicles which had "no work", but had the potential to make trips, were included in the analysis. In the report (Reilly *et al*, 1987), the graphs, showing the distribution of trip frequency by vehicle class, did not present data for trucks that made no trips on the survey day. Omitting trucks that did not, but had the potential to make trips on the survey day would skew the analysis and overestimate the number of trips generated.

Survey Instrument

The CATS Commercial Vehicle survey instrument was designed to meet three criteria:

- (1) A physically manageable and convenient product;
- (2) A device that made the least personal demand on the respondent (considered most likely to be the driver), and

(3) A format that would obtain the desired data in way that could be coded and processed efficiently.

A small portable logbook format, postage-paid mailback provision, and self-adhesive seals were designed to make the survey more manageable and easier to administer. A total of 17,834 surveys were mailed or hand-delivered. A sample of the survey instrument used can be found in **Appendix III**.

Truck Model Output

The data from the surveys were used to calibrate the truck models for *present* and *future* scenarios. The following were the types of outputs that were created from the truck models:

- Separate and combined absolute 24-hour volumes for heavy, medium and light commercial vehicles;
- Separate and combined volumes for truck weight types, with VEQs applied;
- Separate and combined volumes for truck weight types, as a percentage of all vehicles (with or without VEQs applied);
- Volume/capacity (v/c) ratio.

Study Findings

The 1986 CATS Commercial Vehicle Survey produced the following conclusions:

- Twenty-six percent of the usable returns showed trips originating in a county different from the county of registration. The study also found a high level of reciprocity, in which a truck trip found originating in a county different from the county of registration was offset by another trip originating from within the county and registered in another.
- B trucks made more residential trips than any other truck category (probably due to the close association between B trucks and personal use). Of the B plate vehicles reported, 21 percent were in personal use. The state vehicle registration files also showed that a large number of B vehicles registered to names of individuals as opposed to companies or corporations.
- Almost three-quarters, or 73.8 percent, of all IRP vehicles were from outside the six-county study area; 7.5 percent were non-operational on the survey day. Only 18.7 percent of IRP vehicles surveyed were in use within the study area during the survey day.
- With the exception of heavy trucks, trip frequency increased with vehicle weight. B trucks made on average 6.9 trips per 24-hour period, and medium trucks made 9.3 trips for the same time period.
- Heavy trucks made much longer trips (24.9 miles), compared to light trucks (9.6 miles); 7.5 percent of heavy trucks had an average trip length of over 60 miles.

- Weighing the average trip frequency and trip length by the number of working vehicles in each class, the study found that almost two-thirds, or 65.2 percent, of all commercial vehicle miles of travel were made by B trucks. When weighing the average trip frequency and trip length by the number of working vehicles in each class **and** by the appropriate vehicle equivalency (VEQ) factors, the results showed that almost half, or 49.1 percent, of all commercial vehicle equivalent miles of travel were made by B trucks. Comparatively, heavy trucks, commonly perceived as having the major impact on VMT, accounted for 28.1 percent of the VEQ miles traveled and only 12.4 percent of the actual VMT.
- B class truck trip ends dominated all land uses, except for the landfill and intransit categories.
- Retail attracted many more B class, light, and medium truck trips. This is perhaps due to the fact that light or medium trucks are more suitable for delivery than heavy trucks.
- A strong relationship existed between heavy trucks and terminal/warehouse land uses; 42.6 percent of heavy trucks were destined to terminal/warehouse facilities.
- Heavy trucks were also found to dominate the "in transit" category. Of all the heavy trucks, almost 9 percent were found to be "in transit", i.e., leaving the region. Of all the commercial vehicle trips in the "in transit" category, 44.8 percent were made by heavy trucks.
- There was a greater concentration of trip ends in the Chicago area. Trips decreased as distance increased from the Chicago core area. Heavy truck activity concentrations can be found in west Cook County and northwest Cook County in the vicinity of the O'Hare Airport.
- Commercial vehicle activity concentrated in certain satellite centers radiating outward from the Chicago area.
- There was a forecasted increase of 52 percent of commercial B plate vehicles and a 10 percent increase for heavy trucks from 1986 to 2010. This means that there would be an overwhelming increase in real vehicle numbers, from 240,000 to 365,700, of B plate vehicles. Chicago Freight Advisory Committee members and freight consultants revealed that there was a growing proliferation of small commercial vehicles. This was because smaller vehicles were more maneuverable, cheaper, disposable after a few years of use, did not require special driver training, could be driven home with the driver, and did not require special garaging. They were more suitable for the growing service, air freight, fast food, electronics, security, laundry and dry cleaning industries. Small commercial vehicles were advantageous for a wide range of quick-response. just-in-time businesses, with high value/weight products such as the financial and legal records delivery industries. The projected steady increase in heavy truck registration, on the other hand, was attributed to a steady trend for truckload activity (economy of scale).
- A need exists for the development of a coding scheme on restricted links for trucks in the modeling network. Since commercial vehicles are converted to

14

passenger vehicle equivalents, it was difficult to identify which trips were truck trips and exclude them from being assigned to some routes where, in practice, they were forbidden by law.

Over the last two decades, CATS has developed and refined the truck travel forecasting effort. Many studies since have cited CATS's pioneering efforts in truck travel forecasting. Since the 1986 CATS survey, several reports have been written based on the data and findings of the commercial vehicle survey (Englund and Ryan, 1986; Stenzel, 1986; Reilly, Rosenbluh and Rawling, 1987; Rawling, 1987; Rawling 1988; Rawling and DuBoe, 1991). These analyses ranged from corridor/route analysis, the effects of toll facilities on truck route choice, creating separate truck origin-destination matrix tables, and truck speed simulation models to truck trip mapping (i.e., the representation of truck activity on a regional or subregional network). CATS argued that having a proper profile of the motor freight industry was essential for planning purposes. As an example, CATS was able to target the trucking industry as part of a public information program/traffic mitigation project associated with the I-94 Dan Ryan reconstruction project (Rawling, 1988). CATS contended that having a good profile of the freight industry was useful in maintaining the necessary public-private dialog that helped minimize construction impacts for the project.

PHOENIX

In 1991, the Arizona Department of Transportation, Transportation Research Center, funded a commercial vehicle travel survey within the Phoenix metropolitan area. The study was conducted by Cambridge Systematics, Inc., Cambridge, Massachusetts. The primary objectives of the study were to collect truck travel data to develop commercial vehicle trip generation, trip distribution, and traffic assignment models. The models were developed to be incorporated into the Urban Transportation Planning System-based travel model maintained by the Maricopa Association of Governments (MAG). The methodology for data collection, the types of information provided by the survey, the transferability of the study findings to other urban areas, and model development possibilities are summarized in a final report to the Arizona Department of Transportation (Ruiter, 1992).

Survey Methodology

The survey only included commercial vehicles registered within the MAG study area. Commercial vehicles registered outside Maricopa County were not included in the survey. The purpose of the survey was to develop new models for internal commercial vehicle trips only. Two sources of data were used to determine the total number of commercial vehicles to be sampled. *The Department of Motor Vehicles (DMV)* produced a computerized file of 157,000 commercial vehicles registered in Maricopa County in 1989 (Ruiter, 1991; Ruiter, 1992). The compiled list of commercial vehicles were then stratified by vehicle weight. The entire DMV file was also sorted by zip code before

15

sample selection. Subsamples were obtained so that vehicle weight categories would be represented for all geographic areas. The sampling strategy was designed to obtain 40 percent sample of light vehicles (under 8,000 lbs) and 20 percent for each of the three remaining weight categories (8,000-28,000 lbs, 28,000-64,000 lbs, 64,000+ lbs). The second was a list of 2,300 vehicles garaged in Maricopa County. All of these vehicles were owned by the U.S. Postal Service (USPS) but were not registered in Arizona. The USPS vehicles were sampled by weight and by garaging location. They all fell into the two lightest vehicle weight category (2,180 in the under 8,000 lb category and 101 in the 8,000-28,000 lb. category). The selection process provided 1 in 40 postal vehicles in the light category and 1 in 10 in the next heavier category.

The data collection method used was a combined telephone and mailout-mailback survey. The telephone survey utilized screening questions, including the question of who the mailout questionnaire should be sent to. After telephone contacts, 37 percent of the sampled vehicle owners agreed to participate in the survey. The mailback questionnaire, which included a 1-day trip diary, was then mailed to the participants (see **Appendix IV** for sample of trip diary packet). The questionnaire was also mailed to owners who could not be contacted by phone. The overall response rate for the mailback survey was 30 percent with 720 responses, of which 527 vehicles (73 percent) made trips on the survey day. USPS travel forms, detailing daily itineraries for 62 selected vehicles, were also obtained from the manager of fleet operations. The information obtained were used to fill out the trip diary.

Survey Data

The telephone survey to the vehicle owners revealed that only 75.7 percent of registered vehicles from the DMV data list were available for use for commercial purposes. The mailout questionnaires used for the truck survey was patterned after the 1986 Chicago Area Transportation Study (CATS) Commercial Vehicle Survey (Rawling and Ryan, 1987). It was designed to obtain the following data for each sampled commercial vehicle:

- Starting and ending addresses for all trips on the survey day;
- Vehicle type based on number of axles and body style;
- Estimated gross weight;
- Vehicle usage for home-based work and work-related trip purposes;
- Total number of one-way trips on the survey day.

In addition to the above information, the travel diary requested the following information on the first 10 one-way trips made by each vehicle on the survey day:

- Start and stop times;
- Stop odometer readings;

- Name and address of each stop;
- Driver and vehicle activity of each stop;

1

41

- Land use at each stop; and
- Vehicle type and total axles for each trip (to determine trailer pick-up and dropoff locations.)

ŧ

Truck Models

Γ

Since the survey included information on land uses at trip ends and the MAG zonal data included the number of residents and employment by land use category, it was possible to analyze trip generation rates by land use categories. The five land use categories used in the analysis were retail, industrial, public, office, and other. The equation for the trip rates is:

truck trip rate for land use category $i = \underline{study area trips to land use category } i$ study area employment at land use category i

An additional land use category - residential land - was also included in the survey. Trip rates for trips to and from this land use category was defined as:

total study area trips to residential land total study area households

The equation for the trip generation models for each vehicle weight category is:

 $\text{TRIPS}_{wi} =$

t*TOTHH_i + u*RETEMP_i + v*INDEMP_i + w*PUBEMP_i + x*OFFEMP_i + y*OTHEMP_i + z*RESHH_i

where

$TRIPS_{wi} =$	total average weekday commercial vehicle trips for vehicle weight
	category w originated from and destined for zone or district i;
$TOTHH_i =$	total households in zone or district <i>i</i> ;
$RETEMP_i =$	total retail employees in zone or district <i>i</i> ;
$INDEMP_i =$	total industrial employees in zone or district <i>i</i> ;
$PUBEMP_i =$	total public employees in zone or district <i>i</i> ;
$OFFEMP_i =$	total office employees in zone or district <i>i</i> ;
OTHEMP _{i} =	total other employees in zone or district <i>i</i> ;
$\text{RESHH}_i =$	total resident (non group quarters, non temporary, and non seasonal)
	households in zone or district <i>i</i> .
For consistency with the MAG's models, a gravity model was used to simulate trip distribution. The iterative application of the TRANPLAN gravity model calculation program was used to calibrate the model, supplemented by a spreadsheet to help make manual friction factor adjustments. The model calibration process involved reestimating each friction factor using a correction term equal to the desired fraction of trips in a travel time range divided by the previously estimated fraction in this range. The travel time ranges were selected to ensure that the resulting friction factors would always decrease as travel times increase. Furthermore, a two step adjustment was made to the calibration process to account for vehicle registration under reporting in Maricopa County:

- (1) The commercial vehicle trips were expanded by weight class to account for the average number of axles per vehicle in each weight class.
- (2) The total commercial vehicle trips were expanded by VMT so as to match the estimated with the observed VMT for the Phoenix region.

Study Findings

The Phoenix Commercial Vehicle Survey produced the following conclusions:

- Vehicles in the lighter weight categories made more trips; 96.6 percent of all commercial trips were made by the two lightest weight categories. The average trips per vehicle for the 8,000-28,000 lb category, for example, was 9.6 trips; whereas, vehicles in the 64,000+ lb category made only 4.0 trips per vehicle.
- Vehicles in the heaviest category made few but long trips. The average VMT per vehicle for the 64,000+ lb category was 156.8 miles, compared to 56.2 miles for the 8,000-28,000 lb category. Vehicles in the heaviest weight category averaged 33.4 miles per trip, compared to 11 or less miles per trip for the lighter weight categories.
- A total of 79.4 percent of the surveyed vehicles were used for commercial purposes.
- Of the total number of vehicles used for commercial purposes, 41.9 percent were also used for travel between home and work.
- Most trucks started their first trip between 6:00 a.m. and 9:00 a.m. This pattern, however, varies by weight category. Light trucks were more likely to start their first trip between 6:00 a.m. and 9:00 a.m. Heavy trucks (51.8 percent), however, started their first trip before 6:00 a.m.
- The peak period for truck travel occurred between 9:00 a.m. and 2:00 p.m. Heavy trucks, however, have a shorter peak period (11:00 a.m. to 2:00 p.m.). During both of these periods, 13 percent of daily commercial vehicle travel occurred.

- In comparison with AM and PM peaks for private vehicle travel, the results found that the AM peak period travel was as important for commercial vehicles as for private vehicles.
- The trucks surveyed made on average 7.7 trips per day. Light trucks made on average more trips (12.1 trips) than heavy trucks (4.7 trips).
- The average vehicle weight per commercial trip was 11,870 lbs.
- Light trucks (8-28,000 lbs) were found to be heavily used for service delivery and personal business. Heavy trucks were most used for loading and unloading cargo at their trip ends.
- Heavy trucks were found to make high proportion of trips (26.7 percent) to residential land uses. Analysis showed that the reason could be that heavy trucks, to a large extent, were used to delivery construction materials, including lumber and ready-mixed concrete to residential construction sites.
- Trucks in the medium weight category (28-64,000 lbs) were also found to make significant proportion of trips to residential areas. These trips were speculated to be largely solid waste pick-ups.
- Over one-third of all commercial vehicles stops were made on-street. Light vehicles made half of their stops on-street.
- The overall average trip time was 28.1 minutes. Trip time generally increased with vehicle weight.

The Phoenix commercial vehicle survey experience also produced a list of requirements and suggestions for model transferability to other urban areas. These requirements and suggestions include:

- A file from the state vehicle registration agency of all commercial vehicles registered to owners in the study area;
- The ability to geocode street addresses to traffic analysis zones (TAZs);
- Current zonal data on households and employment by type of vehicles and land area;
- A matrix of zone-to-zone off-peak highway travel times in the year of the truck travel survey;
- An existing model system to which truck travel models can be added or replaced;
- Estimates of regional VMT by commercial vehicle type and by private automobiles.

In telephone conversations with the planning staff at the Maricopa Association of Governments, sources revealed that the survey data has been used mainly for truck modeling purposes only.³⁻¹ In the future, the data could potentially be used for intermodal freight planning and to conduct separate truck assignments independent of passenger vehicle assignments.

³⁻¹ Telephone conversation with Cathy Arthur, Maricopa Association of Governments, Phoenix, Arizona, November 30, 1994.

Although the models developed for Phoenix have done a relatively reasonable job of forecasting truck travel, the study does cite inherent trade-offs between the cost of conducting a truck survey and the precision of the models in forecasting truck travel. It recommends that growing cities in the South and West whose current or expected future levels of truck travel are similar to those in Phoenix can implement a complete transfer of the Phoenix models. It also suggests that adjustments reflective of local conditions is encouraged to improve model accuracy. These changes include:

- Revisions to the trip generation models implemented in FORTRAN to reflect coefficient changes required to match local measures of vehicle registrations or VMT, or both;
- Revisions of the friction factors inputted to the trip distribution models to reflect changes to match local data on average trip lengths.

PORT AUTHORITY OF NEW YORK AND NEW JERSEY

Truck Commodity Surveys

In 1974, 1982, 1985, and 1991 the Port Authority of New York and New Jersey (PANYNJ) conducted Truck Commodity Surveys of trucks traveling eastbound³⁻² on six toll bridge/tunnel crossing facilties. The location of the six crossings - George Washington Bridge, Lincoln Tunnel, Holland Tunnel, Bayonne Bridge, Goethals and Outerbridge Crossing - are shown in **Figure 3-1**. The data collection was part of an ongoing effort to monitor truck traffic passing the toll facilities. Results for the 1974 and 1982 surveys were not available. A final report for the 1985 survey was obtained from the Port Authority, and it summarized the survey methodology and findings (The Port Authority of New York and New Jersey, 1987). A copy of the 1987 Commodity Survey form can be found in **Appendix V**.

Some 15,000 trucks were intercepted at toll queues between November 1984 and November 1985. Trucks with two or three-axles were classified as small trucks, and four or more axles were defined as large trucks. Vans were not included in the survey because the existing toll classification system grouped vans with automobiles. The survey period covered the hours between 4:00 a.m. and 8:00 p.m. Truck drivers were interviewed to obtain information on commodity, trip origin and destination, and the type of facility they were going to. Each interview was completed within 30 seconds to minimize traffic disruption to toll facility operations. A copy of the sample survey form can be found in **Appendix VI**.

³⁻² Westbound data was not collected due to difficulty in stopping trucks and disruption to the traffic flow. Vehicular movement is free-flowing on the westbound direction, with no toll plazas which vehicles can form queues. For these reasons, the survey was limited to the toll direction.



purce: Port Authority of NY & NJ / Interstate Transportation Department / Goods Movement Division

The 1991 Truck Commodity Survey produced the following findings:

- *Trip frequency*: Most trucks used the Port Authority crossings frequently on a typical weekday. Almost half used one of the crossings at least once a day, and almost 30 percent more than once a week.
- *Geographic Origin and Destinations*: Two-thirds of the eastbound trucks using the Port Authority crossings began and ended theirs trips within the region, with 85 percent of the total eastbound truck trips terminating within the region.
- *Facility Origins*: Warehouses were the predominant points of origin 45 percent of trucks originanted from warehouses. Of the nearly one quarter that originated from factories, nearly 60 percent came from factories within the region.
- *Facility Destinations*: Most trucks were traveling within and to other warehouses or points of consumption within the region (retail stores, residences). Nearly 40 percent of drivers indicated they were going to warehouses.
- *Regional vs. Through Traffic*: The majority of the truck traffic served the region. Only about 7 percent of the eastbound truck traffic at the 6 crossings were passing through the region.
- *Type of Movement*: The most commom type of origin-destination movement was warehouse to warehouse.
- *Commodity Carried*: Food was the major commodity carried eastbound over the crossings between 4:00 a.m. and 8:00 p.m.
- *Truck Size*: Truck size was evenly split between small and large trucks.
- *Peak Period Traffic*: Trucks at the 6 crossings represented an average of 11 percent of the total peak period vehicular traffic. Large trucks accounted for 45 percent of the peak period truck traffic.
- Morning Peak Period Traffic: Over 35 percent of trucks made trips during the morning peak period (6:00 a.m. to 10:00 a.m.). The results, however, vary by individual crossings.
- Close to 80 percent of the trucks were full or partially loaded. Many of the 20 percent "empty" trucks were carrying empty containers.

In summary, the survey findings indicated that each crossing serves specific markets and displays distinct characteristics in terms of eastbound truck traffic. The survey displayed an important recognition that truck travel plays a vital role in supporting the region's economy - supplying manufacturing and service firms, linking ports-of-entry to their inland customers, and supplying consumers in the region. The truck survey was important in analyzing regional economic competitiveness of truck freight. Taste (1994) noted that the critical portions of the area's transportation infrastructure were built before interstate highway standards were developed. Thus, the functional adequacy of the infrastructure to handle modern trucks and tractor-trailers, he maintained, has been a key element in most analyses, especially in terms of regional economic competitiveness.

In 1991, the Port Authority conducted its fourth truck commodity survey. Four reports were planned. The survey, however, produced only one report on the analysis of the George Washingon Bridge (The Port Authority of New York and New Jersey, 199?) due to the lack of funding to complete the other three. The purpose was to understand the goods movement role of the six interstate vehicular crossings in the region's businesses, consumers and import/export traffic. The specific objectives were to use the data to help portray the economic functions performed by the trucks that use the facilities and general goods movement flow within the region. The information that was obtained were used to develop strategies for alleviating peak period congestion and facilitate overall freight movement in the New York - New Jersey area. The most recent commodity survey data was used by the Interstate Transportation Department's Goods Movement Division to evaluate freight initiatives such as the Circumferential Commercial Corridor and the proposed Northern Corridor, a dedicated commercial route. The data was also used by other departments and outside agencies to analyze regional highway network changes such as the opening of I-287 and the Gowanus reconstruction project. Moreover, the data has been incorportated into New York and New Jersey's State Intermodal Plans as required under the Intermodal Surface Transportation Efficiency Act (ISTEA).

11 - U.

The survey was conducted on December 10, 1991, and over 4,500 truck drivers were interviewed. The sample represented 35 percent of the total truck toll transactions recorded for that day. Each interview was completed in less than 1 minute and 15 seconds, minimizing disruption to traffic on the bridge. The data were considered typical of weekday traffic. For comparison purposes, the 1991 survey employed the same methodologies used in the 1985 Truck Commodity Survey. Similar types of information were gathered via interviews with truck drivers. The only difference was the incorporation of a series of "reverse trip questions" during the interview process. Since westbound information was not collected for the 1985 survey, the 1991 survey attempted to collect information for return trips. The analysis included truck traffic through and within the region, marine terminal and airport traffic, trucks to Manhattan, Long Island, etc.

The preliminary findings from the George Washington Bridge survey seemed to suggest a shift towards using more large trucks. This was found especially true for the food industry. The report cited that other freight research have shown the shift because of more cargo carrying capacity of larger and wider trucks. The reverse trip findings revealed that nearly half of the trucks interviewed returned empty. Moreover, 73 percent of the truck drivers interviewed in the toll direction indicated that they used the George Washington Bridge on the reverse trip. The survey also found that "through" traffic seemed to avoid the peak period. The overnight, off-peak period (8:00 p.m. - 6:00 a.m.) had the largest percent of "through" traffic. Only 8.7 percent of the morning peak truck traffic was "through" traffic, and 13 percent for midday.

Truck Cordon Surveys

From 1992 to 1994, the Port Authority also conducted three Truck Cordon Surveys at 18 locations (See Figure 3-2). The surveys were separated into two phases:

- Phase I (1992): at 3 locations; and
- Phase II (1993/1994): at 15 locations.

The latest survey, being completed in Spring 1994, was a bi-state, multi-agency funded effort. The survey cost was \$312,000³⁻³ with partnership from the New Jersey Department of Transportation (NJDOT), the New York Metropolitan Transportation Council (NYMTC), and the Port Authority of New York and New Jersey. According to the Port Authority, the latest surveys were conducted at 18 locations with 3 interviewers per toll plaza for 24 hours.³⁻⁴ Out of a total of 38,823 trucks that crossed the Port Authority bridges during the survey period, 14,671 were surveyed. The sampling rate was 37.8 percent. The cost per completed survey came to around \$21. The lower cost is a result of higher interviewer productivity. Because the surveys were relatively short to complete (around a minute), each interviewer can survey as many as 40 to 50 surveys per hour. Thus, the number of surveys an interviewer is able to complete is a function truck volume passing through the interview site. The greater number of trucks passing through the interview site, the greater number of surveys the interviewers are able to complete.

In May 1994, a paper was presented at the Transportation Management Conference/Workshop at the State University of New York (SUNY), Graduate Program and International Transportation Research Center (Taste, 1994). The paper detailed the basic issues of a truck driver interview surveys from experiences learned from the Port Authority of New York and New Jersey. The paper stated that good truck data on origin and destination, vehicle type, and commodity flow, will be essential in establishing quantifiable performance measures in formulating Intermodal Management Systems as required under ISTEA.

The paper presented a few issues to be resolved before conducting a field truck driver interview at toll plazas. The following is a summary of a few important points:

- Obtain official permission to conduct survey.
- Determine manpower estimates (typically by examining toll transaction data).
- Field trip to site and meet toll plaza operations supervisor and staff.

Two types of survey instruments were presented (see **Appendix VII**). The first dealt with "reverse trip" questions for interviews conducted for one traffic direction only

³⁻³ The cost only included survey design, data collection and reporting. No analysis was included in the survey cost.

³⁻⁴ Information from Ron Taste, Transportation Planner, Goods Movement Division, Interstate Transportation, Port Authority of New York and New Jersey, December 1994.



.

H |

(typically the toll direction). The second dealt with "facility type" questions for intermodal information. In summary, the Port Authority suggests keeping the field interview at toll plazas under one minute to minimize traffic disruption. Furthermore, because of time constraints, the sequence of questions is also important. The survey form should be well designed to work in the dark and in poor weather conditions. The questions should be precise and clear to solicit the proper responses without further explanation from the interviewer.

EL PASO

The Texas Department of Transportation (TxDOT) and the City of El Paso Metropolitan Planning Organization conducted a Travel Study of the El Paso County area in the summer of 1994. The Travel Study was comprised of six separate surveys: the External Travel Survey, the Home Interview Survey, the Workplace Survey, the Special Generator Survey, the Commercial Truck Survey, and the Air Quality Survey (Barton-Aschman Associates, Inc., 1994). Because El Paso County has been identified as a serious non-attainment area for ozone and a moderate non-attainment area for carbon monoxide by standards established by the 1990 Clean Air Act Amendments, and because trucks can be major emitters of pollutants, the Commercial Truck Survey has been a major element of the El Paso Travel Study. The collected data were intended to meet the data requirements for both travel demand and air quality modeling within the El Paso area. A draft report has been produced which includes a brief description of the survey methodology, the survey instructions and instrument.

The Commercial Truck Survey only included trucks registered in El Paso County. Trucks registered outside of the County would have been surveyed in the External Survey. Furthermore, non-commercial light-duty trucks were also excluded from the survey. The survey sample was drawn from the database of Texas Vehicle Information and Computer Services, Inc. (TVICS). This private firm provided the vehicle list by address of owner, type of vehicle, and the gross vehicle weight. Only commercial vehicles registered in the County with a GVW of 8,500 lbs or greater were selected. The TVICS list provided 5,000 commercial of which 2,511 vehicles met the weight criteria. Out of the 2,511 vehicles, 441 were selected to participate in the survey. To avoid oversampling large owners/operators, only 1 out of every 5 trucks were included in the survey for multiple truck ownership. Each selected truck owner/operator was telephoned to solicit participation and requested to record trip information for a specific 24-hour period. Reminder calls were made to the participants one or two days prior to the appointed survey day. Telephone interviews were conducted the day after the assigned travel day to collected the requested information. The actual survey was conducted between June and July 1994. The survey obtained truck travel information on trip origin and destination, trip start and end times, trip purpose, land use at trip ends, truck weight and size, odometer readings, fuel type used, and the business type of the owner or operator. Moreover, "key facility" route information were collected for each trip to

provide basic routing information that can be used to validate truck assignment procedures. A copy of the survey instrument can be found in **Appendix VIII**.

The survey produced a response rate of 42.6 percent with 188 completed and usable questionnaires. The draft report cited that some of the owners said they were too busy to return the questionnaires (Barton-Aschman Associates, 1994). Many, however, preferred to return the forms by mail rather than spend time on the phone to report the requested information. Furthermore, the consultants speculate that the lower response rate might be the fact that the person who agreed to participate was generally not the truck driver. At the time of this report, the results from the survey were not yet available.

NORTH CAROLINA

At the time of this report, the North Carolina Department of Transportation is currently planning for a truck survey in the Triad Area (Greensboro-Winston Salem-High Point). A survey pretest has been completed, which yielded travel information on 500 commercial vehicles. The total cost of the pretest was approximately \$92,400.³⁻⁵ The analysis has not be completed, and no information was available. Only a copy of the survey instrument was obtained (See Appendix IX).

HOUSTON-GALVESTON

The Houston-Galveston Area Council (H-GAC) has just completed a commercial vehicle survey in November 1994. The survey was conducted by Wilbur Smith Associates. The survey obtained travel information for commercial vehicles with a gross vehicle weight (GVW) of 6,000 lbs or greater and for-hire passenger vehicles such as taxis, limousines, and vans (Wilbur Smith Associates, 1994). The data will be used to estimate truck travel in the region as an input to the regional travel demand model. A random sample of commercial vehicles with GVW of 6,000 lbs or greater were drawn from the Department of Motor Vehicles' registration list.³⁻⁶ A separate database of for-hire passenger carriers were used to obtain the sample for the for-hire passenger vehicle survey. Public transit operators and private fixed-route operators were excluded from the for-hire passenger vehicle survey. The selected sample was designed to obtain 500 useable truck surveys from appropriate commercial vehicles categories, including light-duty gas trucks and heavy-duty trucks. Another 400 usable for-hire passenger vehicle surveys were also completed. The estimated response rate from the survey was between 35 and 40 percent. Survey participants were asked to record 24-hour trip logs during the

³⁻⁵ Information according to Nancy McGuckin, Senior Associate, Barton-Aschman Associates, Inc., Washington, D.C., January 1995.

³⁻⁶ The El Paso Commercial Truck Survey drew the sample frame from the TVICS database. There was no discussion in the study report of whether the same sample frame was considered for the Houston-Galveston survey.

assigned weekday (Monday through Friday) and during Saturday or Sunday. A sample of the survey instrument is included in **Appendix X**. Respondents were requested to provide information on trip origin and destination, departure and arrival times, daily odometer readings, type of fuel used, model year of truck, cargo carried, and land use category of each trip end, etc. The data collection, coding, and reporting of the truck and for-hire passenger vehicle survey cost approximately \$150,000 (Wilbur Smith Associates, 1994). Presently, the data coding and analysis is still ongoing. No information is available on the results of the survey. The Texas Transportation Institute (TTI) is expected to perform the expansion and analysis of the data at a later date.

CALGARY

In 1971, the City of Calgary, Calgary Transportation Study conducted home, truck, taxi and external origin-destination surveys. The surveys were part of a larger study called the Urban Evaluation System. The purpose was to evaluate the inter-relationship between the configuration of transportation and utility systems and the geographical distribution of activities in the study area. The survey procedures for the truck interviews were summarized in a report (Calgary Transportation Study, 1974). Out of the total population of trucks registered in the City of Calgary, 20 percent were sampled to obtain travel information. Every fifth truck from the motor vehicle registration file was randomly selected by the computer. Personal-use trucks, discovered from telephone contacts with owners, were eliminated from the sample. As a result, 3,252 trucks from a total population of 16,540 trucks were selected. Furthermore, 42 of the 450 trucks not registered but were based in Calgary and 22 buses were included in the sample. The survey was conducted between August 16 and the end of September 1971.

The truck survey involved two separate methods. Non-fleet truck owners were surveyed via telephone interviews, and fleet owners were telephoned and followed up with a face-to-face interview. Telephone calls were made to non-fleet truck owners the day prior to the survey day. They were requested to make note of specific information related to the travel of the sampled truck. Another telephone call was then made immediately following the survey day, and the requested information was obtained. Fleet owners were contacted by the phone just like the non-fleet owners but were followed up by a face-to-face interview to obtain the requested truck travel information. The two methods were documented to be successful. Truck travel information such as industry or business type, vehicle type, trip start and end times, commodity carried, and origin and destination information were obtained. A temporary staff of 5 interviewers were hired to conduct the truck interviews. The total personnel cost of conducting the survey was approximately \$4,600 and the coding of the survey was approximately \$2,000.

Aside from an internal survey of registered commercial vehicles in the study area, an external cordon survey of all vehicles, including commercial vehicles, was conducted

during the daylight hours between 6:00 a.m. and 8:00 p.m. The primary objectives of the external cordon survey were:

- 1) To determine the total number and classification of all vehicles entering and leaving the City of Calgary; and
- 2) To obtain origin and destination information from drivers of these vehicles.

In theory, the sampling design estimated that a 50 percent sampling rate would be satisfactory. In practice, however, the field interview achieved close to a 100 percent sampling rate, with the exception of the peak hour periods. Sampling rates for the peak hour periods dropped to 80 percent in order to keep vehicular queues within an acceptable limit. Occassionally, the sampling rate for peak hour periods dropped to approximately 50 percent due to vehicles stalling while waiting in line for interviews. Furthermore, the sampling rate was dependent on the number of interview lanes available. The report estimated that a maximum of 360 interviews per hour per lane was achieved with 9 interviewers per lane. Each interview took approximately 90 seconds to complete. Vehicle classification counts were also conducted at the same time period of the surveys. The external cordon survey employed 37 temporary staff and 4 supervisors.

ONTARIO

The Ontario Ministry of Transportation periodically conducts surveys of commercial vehicles based on 5-year intervals for planning and operational purposes. Roadside intercept surveys were completed in 1978, 1983, and 1988. At other times, ad hoc surveys are conducted at selected locations (Babin, Junor, Little, and Rhone, 1993; Junor, 1993). These ad hoc surveys are designed to address specific needs. For the 1988 Ontario Commercial Vehicle Survey, 19,000 trucks were surveyed to gather truck travel information (Gorys, 1991). The primary purpose of the 1988 commercial vehicle survey was to provide a current profile of trucking activity in the Province for the planning, delivery, and evaluation of Ministry programs. Other than the primary purpose, the survey was also conducted to gather information on the following:

- Nature and extent of dangerous goods movement;
- Structure of the industry between private and for-hire carriers;
- Transborder movement of goods, particularly the degree and nature of traffic made by other provincial and U.S. carriers;
- Seasonal variations in transportation and commodity movements;
- Profile of commercial vehicle drivers (demographics and other characteristics);
- Commodity and load characteristics; and
- Measures of efficiency and productivity on the basis of empty truck movements and ton-miles transported.

Questions were asked in regards to the driver's employment characteristics, such as age, sex, years of experience, recent training, the number of hours expected to work on the particular trip being surveyed, carrier type, union affiliation, method of renumeration, and employee category type. Other information such as the type of commodity hauled, the degree of utilization level of the vehicle, and the origin and destination of the vehicle were also collected. A copy of the sample questionnaire is in **Appendix XI**. The information was cross-checked, whenever possible, through a review of photocopied waybills.

Survey Methodology

The survey was conducted during a 23-week period, from March to November 1988. Interviews were undertaken on 100 days, but not necessarily for the entire day, depending on proper lighting, weather, and safety conditions. It was carried out at 57 locations along principal intercity highway inspection stations, rest stops, and at a few border crossing points (see **figure 3-3**). Where possible, surveys were conducted at the identical 1983 survey locations. The survey consisted of 29 questions, and 8 questions specifically pertained to the driver. The data collection included carrier information, area of registration, load utilization, commodity type and weight carried, and trip origin and destination information. The station inspectors recorded truck body type, number of axles, vehicle weight, etc. Interviewers recorded vehicle plate numbers and then approached truckers to request approval to commence with the survey. Fully completed interviews lasted between 8 to 12 minutes; it was slightly longer when portable scales were deployed. During the duration of the survey, a vehicle classification count of vehicles passing by the interview location was also undertaken. The purpose was to expand the sample data to represent the daily average traffic for the survey location.

Survey Data

A total of 19,225, or 8.6 percent, of the total population of commercial vehicles were interviewed over a 1,855 hour period. The overall refusal rate for the survey was 3.5 percent. Another 1,363 hour period was used to collect information on vehicle type. No surveys were conducted during this period of time because of weather problems, absence of enforcement staff, equipment, etc. The 7-day, 24-hour vehicle classification counts at two selected inspection stations west of Toronto were conducted to obtain data on daily and hourly variations in truck travel. The sampling rate varied by location, time-of-day, day-of-week, and the degree of traffic passing by the survey locations.

Data Limitations

The survey was not without its limitations. The prime limitation was that the data collected was expanded to reflect a yearly flow of truck traffic. This practice presented questionable results, and was not encouraged by the author, given possible seasonal variations in truck travel. Second, the placement of the interview sites presented a bias towards trucks traveling on the major intercity routes. Since the interview was conducted

30



Note: Arrows indicate direction of traffic surveyed.

Figure 3-3: 1988 Ontario Commercial Vehicle Survey Sites

at inspection stations along intercity routes, only trucks traveling along these routes would be surveyed. But because the principal purpose of the survey was to sample major intercity traffic, this was not a major concern. Moreover, because enforcement was being practiced during the duration of the survey, there were concerns that vehicles not incompliance would bypass an open inspection station (and hence avoid the chance of being interviewed). However, selected classification counts on known by-pass routes near inspection stations where the surveys were undertaken found this not to be true. Third, the survey does not represent all trucking activity. Since only the major intercity routes were sampled, truck traffic on secondary routes and rural highways, with considerable urban origin or destination, were not adequately represented in the sample. Fourth, only a small proportion of weekend truck travel, on Saturday, was captured. Week-long classification counts at two inspection stations, however, showed that weekend truck traffic was quite low. Saturday travel represented 6.4 percent of weeklong travel, while Sunday represented only 1.1 percent. Lastly, nonresponses, provision of false information, etc., resulted in missing values for incomplete interviews. Nonresponses varied considerably by question type. Typically, they were between 9 to 14 percent for each question. They were a function of language translation problems, refusals, misinformation, and in some instances, unawareness. Whenever possible, missing information were filled in through a review of other surveyed information.

Truck Volume by Time-of-Day

The sampling rate also varied by time-of-day. In general, the results found that it was easier to sample trucks during periods when volumes were lower. Truck volumes were found to be highest during the 4:00 p.m. to midnight period. An analysis of truck volume by hour in proportion to the total traffic volume, however, revealed that trucks accounted for between 10 and 58 percent of the total traffic, depending on the hour surveyed. The greatest proportions were found to constitute 17 percent of the total vehicular traffic. Combination vehicles (tractor-trailer combinations, for example) were most evident during the early morning hours, comprising of as much as 85 percent of all trucks. Straight trucks, on the other hand, were most visible during normal business hours, accounting for as much as 22 percent of all trucks.

Truck Volume by Day-of-Week

The 7-day classification count at the two selected inspection stations found that truck traffic was highest on Wednesdays, and Sundays produced the least amount of truck traffic (**Table 3-2**). The results also revealed that the greatest number of surveys obtained and the highest truck traffic volumes were found during the mid-week (Wednesdays or Thursdays). Despite an effort to minimize sampling on Mondays and Fridays, the survey produced a higher sampling rate on those two days. The author attributed the results to a greater ability to capture truck traffic (given lower volumes generally) on those days.

Day-of-	Surveys	%	Trucks	%	Sample
<u>Neek</u>	Collected	of Total	Classified	of Total	<u>Rate</u>
Sun	0	0.0%	0	0.0%	
I on	1,616	8.4	7,078	6.6	22.8%
ſue	3,722	19.4	17,643	16.5	21.1
Wed	5,084	26.5	26,386	24.7	19.3
ſhu	4,542	23.6	33,111	31.0	13.7
ri	3,623	18.8	20,394	19.1	17.8
Sat .	638	3.3	2,219	2.1	28.8
[otal	19,225	100.0	106,831	100.0	18.0

Survey Findings

The 1988 Ontario Commercial Vehicle Survey produced the following findings:

- Fully completed interviews lasted between 8 to 12 minutes; it was slightly longer when portable scales were deployed.
- Nonresponses varied between 9 to 14 percent for each question.
- The 7-day classification count found that truck traffic was highest on Wednesdays, and Sundays produced the least amount of truck traffic.
- Truck volume in proportion to the total traffic volume was found to be greatest during the early morning hours (3:00 a.m. to 5:00 a.m.). Overall, trucks were found to constitute 17 percent of the total vehicular traffic.
- Combination vehicles were most evident during early morning hours, comprising of as much as 85 percent of all trucks. Straight trucks were most visible during normal business hours, accounting for 22 percent of all trucks.
- Tractor-trailer units were the predominant type of truck, at 77 percent of the surveyed population.
- Straight trucks (16 percent of the surveyed population) were the second largest truck type group.
- Van body style trucks accounted for about 60 percent of the truck-trailer units. This was a reflection of the operational flexibility of van body style trucks, especially in the transportation of general freight.
- Five-axle tractor-trailer units were the largest group of vehicles, at 57 percent of the total truck population. The 2-axle straight trucks (10.7 percent) and the 6-axle tractor trailer (10.3 percent) were the two next categories.
- Of the vehicles surveyed, 38 percent were equipped with energy-saving devices such as roof-mounted air deflectors, side fairings, and bubble-type devices.

- Private carriers used lighter vehicles than for-hire carriers. The for-hire carrier industry preferred larger vehicles with 6 or more axles.
- The survey of on-board monitoring devices found that logbooks were the principal trip recording devices (53 percent of all vehicles), followed by tachographs (32 percent), and on-board computers (4 percent).
- A total of about 5 to 6 percent of all truck trips surveyed involved the carrying of dangerous goods. Flammable liquids (47 percent) were the most frequently transported dangerous goods, followed by compressed gases (24 percent), and corrosive substances (20 percent).

Survey Data Applications

According to the Ontario Ministry of Transportation, the data collected from the 1988 commercial vehicle survey were of valuable use. They were used to evaluate the geometric design of roads, highways and ramps, pavement and structure rehabilitation planning and scheduling, reciprocity arrangements with adjacent jurisdictions, dangerous goods regulation and enforcement efforts, and driver education programs. They provided useful information for private sector interests on market opportunities, fleet construction and disposition. The 1988 survey also provided time series statistics on commercial vehicle drivers for policy development and analysis purposes. The types of analysis include hours-of-service regulations, evaluation of driver educational resources and recruitment, and comparison of truck-related accidents. Furthermore, data on methods of driver payment, hours-of-service restrictions, and work-related association membership provide insight into issues affecting owners and unions.

VANCOUVER

In 1988, the City of Vancouver and the Greater Vancouver Regional District (GVRD) conducted a truck survey. A detailed description of the survey is documented in the City of Vancouver/GVRD's Truck Study report (Vancouver City Engineering Department and the Greater Vancouver Regional District, 1990).³⁻⁷ A GVRD Truck Model was developed. It was part of a regionwide TRANSPORT 2021 project to recommend a long-range transportation plan for Greater Vancouver with associated policies, demand management measures, and priorities for transportation investment.

GVRD Truck Model

The GVRD Truck Model was developed to estimate 24-hour light and heavy truck travel demand for current and future years. Light trucks are classified as having a gross vehicle weight (GVW) of 4,500-20,000 kilograms (kg). Trucks over 20,000 kg are classified as heavy trucks. The 1988 Truck Survey origin and destination data was used to calibrate

 $[\]overline{}^{3-7}$ At the time of this report, the Truck Study report was not available for review.

the 1989 GVRD Truck Model. The model was subsequently validated to 1991 conditions using truck screenline data. The model is comprised of three main components:

- 1. A traffic zone system: comprised of 445 traffic zones. The size of the zones varies according to population and employment densities. There are 11 external zones (at six border crossings, airport and ferry terminals) at entry points to the region to account for traffic entering and leaving the region.
- 2. A regional light and heavy truck network: The network is comprised of freeway, arterial and collector facilities. Each roadway link contains information on the number of lanes, posted speed limits, capacity, and turning restrictions.
- 3. A truck demand modeling procedure: This is a procedure that predicts the number of 24-hour light and heavy truck trips.

The truck demand modeling procedure is a three-step procedure that includes: 1) trip generation, 2) trip distribution, and 3) trip assignment. The trip generation stage estimates the number of truck trips produced and attracted by each traffic zone based on population, wholesale, manufacturing, and non-wholesale employment for that zone. The trip generation equations for light and heavy trucks are:

$$Light_i = 0.327Wh_i + 0.0213NWh_i + 0.0103Pop_i$$

where,

Light _i	=	24-hour light truck trips produced by zone i
Wh _i	=	wholesale employment in zone i
NWh _i	=	non-wholesale employment in zone i
Pop,	=	population in zone i

and,

 $Heavy_i = 0.164Wh_i + 0.0665Man_i$

where,

Heavy _i	=	24-hour heavy truck trips produced by zone i
Whi		wholesale employment in zone i
Man		manufacturing employment in zone i

The trip distribution stage is applied using the "Fratar" modeling technique. Truck trips between origins and destinations are allocated based on the observed heavy and light truck trip distribution patterns. This stage produces a set of 24-hour trip tables for light and heavy trucks. External truck trips are subsequently added to these trip tables.

The final step involves trip assignment - allocating light and heavy truck trips to the computerized network. The network assignment is based on the link travel times derived from the 1991 automobile assignment. The three-step modeling process, together with the traffic zone system and computerized network system, produces estimates of 24-hour light and heavy truck link volumes. These 24-hour link volumes can be factored down to represent travel demands for different time periods during the day.

The GVRD Truck Model results produced the following findings for the base model year 1991:

- Light truck trips outnumbered heavy truck trips by two to one in the Vancouver region.
- The number of daily truck trips in the GVRD exceeded 100,000 trips, and about 15 percent of all truck traffic in the region had an origin or destination outside the region.
- Truck traffic accounted for 3 percent of total daily vehicular traffic, with almost 85 percent of the truck traffic operating within Greater Vancouver.
- External truck traffic accounted for 15 percent of the total goods movement in the region by volume.
- The modeling effort identified regional roadways that complement primary highways and existing goods movement corridors and that are vital to port and industrial activities.
- Significant increase in population and employment along regional roadways were identified. This assumed growth prompted a need for new and upgraded transportation infrastructure for the movement of goods and passengers within these areas.

The GVRD Truck Model results produced the following projections for the model year 2021:

- Total daily truck trips were forecasted to increase by approximately 85 percent from 1991 to 2021. Light trucks were forecasted to increase with a faster rate than heavy trucks.
- The number of heavy and light trucks entering Vancouver on a 24-hour basis was forecasted to increase by approximately 50 percent.
- Average truck speed was forecasted to decrease by 8 percent.
- Average trip distance was forecasted to increase by 6 percent.
- All the above forecasts indicated that trucks would experience higher levels of congestion in 2021.

The GVRD Truck Model effort produced the following policy recommendations:

• Since trucking is the most competitive mode of goods transport, it should be exempted from the same constraints as automobiles.

- Truck traffic should be permitted to escape auto congestion wherever feasible by separating truck flows from auto flows, consistent with cost-effectiveness.
- Consider opening HOV lanes to truck traffic during the off-peak period, provide that the performance and safety of transit is not compromised.
- The use of inter-regional, long haul roads for commuting should be limited, in order to provide trucks with access to long haul facilities from outside the region.
- Provide adequate off-street loading facilities for both goods vehicle and courier functions.
- Develop fuel taxation policies with commercial vehicles in mind.

чi

• Preserve and protect water/rail access to minimize the need for additional truck trips.

Chapter Four CALIFORNIA EXPERIENCES

STATEWIDE TRUCK STUDIES

Between 1985 and 1987, the California Department of Transportation (Headquarters) conducted a study to determine the impact of large trucks (3 or more axles and gross vehicle weight of 26,000 lb. or more) during the morning peak, midday off-peak, and evening peak hours on the urban freeways in the Los Angeles, San Francisco, and San Diego metropolitan areas. The lead consultant for the study was Cambridge Systematics, Inc., and a separate technical report and summary report was submitted to Caltrans (Caltrans, 1988; Caltrans, 1988s). The study was conducted to address the following three questions:

- 1. What are the impacts of large trucks on peak-period freeway congestion?
- 2. Can freeway and truck management techniques reduce congestion?
- 3. What are the economic impacts of these techniques?

The study involved videotaping freeway traffic flows at 78 urban freeway sites. The sites were selected according to the following criteria:

- Congested freeway segments with high annual average daily truck traffic reported in the Caltrans reports;
- Freeway segments identified by Caltrans, the California Highway Patrol (CHP), city and county transportation officials, and motor carrier managers as congested, high volume truck routes;
- Freeway segments where Caltrans maintains truck weight stations or has longterm truck counts that could be correlated to the results of the research; and
- Freeway segments needed to improve systematic coverage of the regional freeway networks.

Of the 78 sites, 40 were in the Los Angeles area, 25 in the San Francisco area, and 13 in the San Diego area. The videotapes were analyzed to determine the number and body type of large trucks during the morning peak, midday off-peak, and the evening peak periods. Ninety minutes of videotape were recorded for each site: 30 minutes during the morning peak period (7:00 a.m. - 9:00 a.m.), 30 minutes during the midday off-peak period (11:00 a.m. - 1:00 p.m.), and 30 minutes during the evening peak period (4:00 p.m. - 6:00 p.m.). Each site was videotaped at one direction; paired sites were used to cover both directions. The trucks videotaped were classified according to the number of axles. Trucks with three or more axles were further classified by configuration (straight, single trailer, double trailer, tractor-only/other) and by body type (van, refrigerated van, flat bed, tanker, hopper, trash hauler, dump truck, auto rack, cattle rack,

trailer on flat car, container on flat car, construction equipment, and other). The truck configurations were used to deduce the type of motor carrier and commodity being carried.

The study evaluated four freeway and truck management strategies: 1) traffic management, 2) incident management, 3) night shipping and receiving, and 4) peakperiod truck bans. It produced the following conclusions about urban freeway truck travel in the Los Angeles, San Francisco, and San Diego metropolitan areas:

- Large trucks accounted for three-quarters of all truck travel (except travel by pick-ups and panel trucks) in the three metropolitan areas.
- Large truck travel was highest during the midday off-peak period, comprising 5.5 percent of all vehicles in the Los Angeles and San Francisco areas, and 2.5 percent of all vehicles in the San Diego area. In the San Francisco area, truck volumes on I-580, east of I-880, were highest during midday. Truck flow on I-880/I-80, south of SR-24, also had the heaviest volume during the midday, from 6.8 to12.0 percent of all vehicles. Truck traffic for this segment of the freeway was heaviest northbound in the morning and southbound during the evening.
- Large trucks have an impact on the freeway equivalent to 1.8 to 2.8 passengers cars. An MTC internal memorandum, however, suggested that the Bay Area freeway terrain, often with grades of 3 to 4 percent for distances of at least one-half mile, causes large trucks to have a greater impact on traffic. The MTC memorandum suggested that the "6 to 10 passenger car equivalent" should be used in analyzing truck impacts in the Bay Area (MTC, 1989).
- Wholesale trade, durable goods manufacturing, and non-durable goods manufacturing industries generate the most miles of large truck travel. Together, they generate almost 90 percent of all truck miles of travel.
- Private truck fleets owned by businesses accounted for about half of all truck miles of travel. Most of their trips are short-haul trips (less than 200 miles). Common (for hire) carriers accounted for the other half of all truck miles of travel. About one-third of their trips are short-haul, and two-thirds are long-haul (over 200 miles).
- Few highly congested freeways have a high proportion (more than 10 percent) of large trucks in the traffic stream. At 90 percent of the 78 sites sampled, large trucks were 9 percent or less of all vehicles. However, I-5, I-605, I-710, and SR-60 in Los Angeles and I-80, I-580, and I-880 in San Francisco were identified to be highly congested with more than 10 percent of the traffic comprised of large trucks.
- As a general pattern, highly congested freeway segments tend to have lower truck volumes than moderately congested freeway segments.
- The impact of large trucks on freeway traffic flow is modest; the perceived impact, however, is greater. Large trucks create a psychological, if not an actual, barrier to passenger vehicle drivers entering or exiting the freeway. They create visibility problems for passenger vehicle drivers. Moreover, their lack of

maneuverability relative to passenger cars in congested and unstable traffic flows contributes to accidents.

- Truck accidents and incidents were estimated to account for 19 million vehicle hours of delay per year at a cost of over \$200 million per year.
- Five to ten percent of all truck incidents were found to cause major incidents which closed two or more freeway lanes for at least two hours.
- The sites selected represent a broad range of freeway conditions, but the results should <u>not</u> be considered as a statistically significant sample of all freeway segments and time periods for the study areas.

LOS ANGELES AREA

The Southern California Association of Governments (SCAG) and Caltrans conducted an urban goods movement study in the late 1970s. A working paper (number VI) was published as part of a series of reports on urban goods movement in the Los Angeles area (SCAG and Caltrans, 1979). The study is based on a sample of freight revenue bills of every regulated (for-hire) truck carrier who moved freight within the state on specified sample days. The study data was obtained from the California Public Utilities Commission (PUC) 1976 survey data. The PUC conducts annual freight surveys. The freight bill data used for the study included month of shipment, minimum rate tariff category, shipment mileage, shipment weight, freight charges, and origin and destination data by city and by the 62 metropolitan zones of Los Angeles and Orange counties.

The sample consisted of a total of 65,346 freight bills for the State of California, of which 37,301 freight bills originated from the six-county SCAG region and 27,413 destined for the region. Of the total samples in 1976, 65 percent had at least one trip end in the SCAG region, and 34 percent had both ends in the SCAG region. The results also found that 35 percent of the freight trips were just passing through the SCAG region. The analysis showed that for-hire truck activity was greatest in Central Los Angeles and declined with distance from the Central Los Angeles area. The core industrial regions (City of Vernon, Southgate, etc.) accounted for one-fourth the value of Los Angeles county freight bills and close to 5 percent of the value of all California freight bills. Moreover, Los Angeles and Imperial counties had the highest number of outflowing trips (49 percent and 56 percent) than any of the other counties. Freight flow from Los Angeles County represented 75 percent of the total outflow from within the SCAG region to outside the region. A possible explanation could be that Los Angeles County is the largest producer of manufactured products, and Imperial County produces the largest amount of agricultural products in the SCAG region.

SCAG also conducted a truck movement study for the Los Angeles region in 1988 (SCAG, 1988). The report contained compilations and discussion of local, state, and federal regulations and legislation that would affect restrictions on truck delivery hours.

It discussed issues that would affect heavy truck delivery scheduling, such as time-of-day patterns, truck volumes and level-of-service on the state highway system, the delivery needs of selected high-volume goods movement facilities, delivery needs of selected economic sectors, and factors limiting shifts of truck delivery hours. Another truck route study conducted by SCAG documented commercial vehicles regulations in the Los Angeles six county area (Imperial, Los Angeles, Orange, Riverside, San Bernardino, and Ventura counties) by weight class (SCAG, 198?). The study found that commercial vehicles with gross weight exceeding 3 tons were restricted by the largest number of local jurisdictions in the region.

11

(1, 2)

In 1988, SCAG also published a report on how to manage the increased truck traffic in and out of the Port of Hueneme in Ventura county (SCAG, 1988). The study was prompted over concerns of increased truck traffic and their associated impacts on traffic congestion, air pollution, noise, and pavement conditions. The Port truck trips were projected to increase by 118 percent by the year 2010, from 286 trips (1986) to 623 trips per day. As part of the study, truck drivers were interviewed at the principal entry/exit gate to the port. The intercept surveys were conducted for seven consecutive days, 18 hours per day from 6:00 a.m. to 12:00 midnight. Time-of-day information, trip origin and destination, type of commodity carried, type of business or industry, and route information were collected. The survey produced 691 responses.

• The following is a summary of the main findings:

- Most truck trips occurred between 8:00 a.m. and 4:00 p.m., with fairly even distribution within those hours.
- The largest share of truck trip destinations, 43 percent, were local (Ventura county). Another 26 percent of truck trips had destinations in the Los Angeles county; 21 percent were within California, and 10 percent were destined for other states. Trips destined to Canada represented 31 percent of all out-of-state destinations.
- The distribution for trip origins was similar to the distribution for trip destinations: 46 percent from local, 22 percent from Los Angeles county, 24 percent within California, and 8 percent from out-of-state.

In 1989, SCAG published a report, documenting new methods of estimating heavy truck VMT, both spatially (by location) and temporally (by time-of-day) (SCAG, 1989). Short of conducting a truck travel survey, SCAG set out to develop a new methodology for estimating truck travel based on existing data on trucks. The purpose was to gather detailed travel information on heavy trucks and to apply the information to regional travel demand, fuel consumption, air quality and emissions models. Although the then current method for estimating truck travel gave reasonable results for emissions produced by heavy trucks in the air basin, it was not adequate in providing locational or temporal information about emissions in the region. Trucks were implicitly included in the trip generation phase for the regional travel demand model as passenger vehicle equivalents. Light-duty and medium-duty trucks were assumed to represent 82 percent of the registered commercial vehicles in the region, and heavy trucks (over 8,500 lbs. GVW) were excluded from the models. Moreover, the 82 percent of commercial vehicles represented in the regional model were assumed to have similar travel characteristics to passenger vehicles. Truck travel estimates were derived using factors developed from other truck travel information. Oftentimes, the source of information can often be dated or provided on an aggregate level. Furthermore, these factors varied according to each link's facility type, area type classification, and by the time-of-day. As a result, truck data for the SCAG region was collected from various sources to come up with a better method of estimating truck travel. The major source of information for the study was obtained from Caltrans. Other commercial vehicle information were obtained from DMV files, the Los Angeles Department of Transportation (LADOT) Downtown Cordon Count Study, and numerous reports from other agencies.

Data obtained from Caltrans included master truck annual average daily truck (AADT) information, truck miles of travel, highway performance and monitoring system (HPMS) information, and truck counts at weigh stations (Caltrans 1987). The truck AADT information was collected from traffic volume counts at 970 count locations at the Caltrans district level. The counts were conducted from 9:00 a.m. to 3:00 p.m. (District 7) or 9:30 a.m. to 11:30 a.m. and 1:00 p.m. to 3:00 p.m. (District 8). The data were then expanded to reflect 24 hour volumes based on factors developed from previously collected 24-hour counts. The estimated ADT volumes are compiled and sent to Caltrans Headquarters to be published. The truck miles of travel data was obtained by axle type (2, 3, 4, and 5+) (Caltrans, 1987). The VMT were estimated from truck AADT information. The HPMS information from the Federal Highway Administration (FHWA) was obtained from Caltrans, which publishes an annual California HPMS report (Caltrans, 1987). Data such as county, functional class, area type, facility type, truck designations, AADT, speed limit, percent trucks, and capacity were extracted from the report. The truck information were for the peak and off-peak periods. Truck weigh-inmotion (WIM) machine counts were obtained from Caltrans for six locations. The data obtained were classified into 15 vehicle classifications, and truck data were grouped into axle groups. In general, the WIM truck data showed that the 2-axle and 5 or more-axle trucks represented the largest share of truck travel in the SCAG region and that truck traffic occurred in the midday off-peak period. There were no evidence of directional differences; however, the distribution of truck traffic was significantly different from passenger vehicle traffic.

Data from the DMV were obtained for the number of vehicles registered by vehicle types for each county (California DMV, 1987). Statistical records were also obtained on motive power for vehicles by body type and weight divisions, stratified by axle groups. Los Angeles Downtown cordon counts were also obtained from LADOT's Bureau of Transportation Planning (LADOT, 1984). The data included machine and manual vehicular and pedestrian counts entering and leaving downtown Los Angeles by location and by half-hour periods. The truck information, however, did not include truck counts by types or weight class. Moreover, large truck peak hour population data were obtained

from the Caltrans Urban Freeway Gridlock Study for reference and cross-checking (Caltrans, 1988). Truck travel distribution by truck classification, fuel type, area and road type traveled were also obtained from the University of Michigan Transportation Research Institute's Analysis of Heavy-Duty Truck Urban Use Data for the SCAG region.

Based on the analyses of the collected data, various assumptions were made on truck travel in the SCAG region. These assumptions were used to develop the new methodology for estimating truck travel. First, routes with high truck volumes were assumed to display insignificant variations over time and that the routes will continue to be the major truck network in the future. Second, the information used for the models assumed that regional truck travel movements were mainly on freeways, not arterials. Heavy trucks will continue to carry goods and commodities using the same established routes. Third, the new methodology assumed that most manufacturing, heavy industrial, and other employment sites using heavy trucks will be located near freeways. Fourth, the gasoline truck fleet is assumed to be less polluting and equipped with some type of catalytic control by the year 2010. Diesel trucks were assumed to have a 74 percent increase in miles traveled and an increase of 38 percent for TOG, 23 percent for CO, and 20 percent for NOX. The majority of diesel trucks were also assumed to comprise of 4 or more axles and make a large share of their trip on freeways.

The new methodology was based on 1984 data and applied to 1987 commercial vehicle information as they became available. The base year used for model calibration was set at 1987. The modeling zone system was consisted of 1,555 zones, based on the 1980 census tracts. The new highway network was built from scratch, converting the links from UTPS HNET format to TRANPLAN HWYNET format, utilizing a new node number system, facility type designation (freeway, major arterial, primary arterial, secondary arterial, HOV, and centroid connector), and a different coordinate system. The highway network produced 13,000 two-way links. TRANPLAN's graphic Highway Network Edit and Display System (HNEDS) software was used to display the freeway network on a graphic terminal monitor and coded in the proper freeway route number for each link. The TRANPLAN network was converted to ASCII format and a freeway route file was created. The same procedures were used to code state route numbers to the arterial links. Freeway ramps, however, were not coded in the network, and cordon stations were re-assigned from the low order of zone numbers to higher order of zone numbers. All the travel demand model data were then merged with Caltran's 1987 truck count data to develop heavy duty vehicle to light duty vehicle (HDV to LDV) factors. The factors were calculated by dividing the daily truck volume by the daily model volume (AMPK + PMPK + OFFPK). Three sets of factors were developed:

1. <u>RTE*FT*RSA</u>. This factor was developed to identify highway links as either intra or inter-state routes (RTE). It represented weighted mean truck percentages for each route by facility type (FT), and by regional statistical analysis (RSA) district. There were 808 observations used to develop 259 factors. Comparison of the sum of the mean values showed that the model estimates were within -3 percent of the counted vehicles.

- <u>FT*RSA</u>. This factor was applied to links with no route number coded. It represented weighted mean truck percentages for each facility type and the RSA location where counts were taken. A factor was developed for each facility type at the RSA level. The 808 observations were used to develop 119 factors. Comparison of the sum of the mean values found that the model estimates were within 1.8 percent of the counted vehicles.
- 3. <u>FT*CO</u>. This factor was applied to links with no route number coded and no truck percentage number for that link's RSA code. It represented weighted mean truck percentages for each county by facility type. A factor was developed for each facility type at the county (CO) level. The 808 observations were used to develop 20 factors. Comparison of the sum of the mean values showed that the model compared within -0.4 percent of the counted vehicles.

In applying the above factors for estimating heavy truck VMT, the following assumptions were made:

- Assume zero heavy truck volumes on secondary arterials, centroid connector links, and HOV links. Due to the lack of travel information, truck traffic was not estimated for the connector links and secondary arterials.
- Assume 50/50 split by direction of travel for truck volumes for all time periods. The assumption was based on 24-hour volume count data from 3 sites.
- Assume that trucks travel at the same speed as other vehicle types.

After the LDV to HDV factors (TRKPCT) were calculated, the total daily truck volumes (TRKADT) were estimated. The TRKADT were calculated by applying the TRKPCT factor to the modeled light-duty vehicle ADT (AMPK + PMPK + OFFPK), such that:

TRKADT = ADT * TRKPCT

The daily truck volumes were then allocated to the three time periods based on factors developed using the 1988 Weigh-in-Motion (WIM) 24-hour machine count data from five sites in the region.

The VMTSUM program was then used to calculate the VMT by county, by the area inside or outside of the SCAG region, and by the area representing the South Coast Air Basin (SCAB). The data were also used to provide information for input into the BURDEN and DTIM air quality models maintained by the California Air Resources Board (CARB). The results found that the 1987 model run overestimated truck counts by 2 percent compared to an over estimation of 18 percent for the 1984 model run. A

comparison of the heavy-duty vehicle VMT as a percentage of the total VMT varied from 5.8 percent to 6.1 percent. A comparison of the speed ranges by county found that the median speed for HDVs were significantly higher than for LDVs.

H

SACRAMENTO AREA

In 1992, Caltrans conducted a Truck Travel and Goods Movement Study in the Sacramento metropolitan region. The study was aimed at understanding truck travel patterns and goods movement across the Sacramento area. Barton Aschman Associates produced a proposal to conduct a truck travel survey in the Sacramento area. A work plan was produced (Barton Aschman Associates, Inc., 1992).⁴⁻¹ The work plan included sampling frame, sample design, survey design, and a discussion of field survey safety issues. Seven potential data collection techniques were examined and an alternative approach was recommended. The seven potential approaches included:

- 1. Mail-back survey based on license plates
- 2. Video surveillance
- 3. Manual license plate survey
- 4. Automated/Electronic data collection
- 5. CB radio/Cellular phone
- 6. Mail-back survey form/roadside distribution
- 7. Roadside interviews

The roadside interview method combined with automated classification counters were judged the best survey approach to meet the study objectives. A summary of the evaluation (advantages and disadvantages) is presented in **table 4-1**.

SAN FRANCISCO BAY AREA

Bay Bridge Truck Travel Study

In 1985, the Caltrans conducted a San Francisco Bay Bridge Truck Travel Study. The purpose was to improve truck and/or vehicular travel approaching the Bay Bridge. The information produced from the study were used to evaluated proposals to eliminate the weave for trucks coming from I-880 and a Caltrans proposal to install automated vehicle identification (AVI) systems on trucks accessing the toll plaza. Truck travel patterns to the Port of San Francisco were also studied.

⁴⁻¹ At the time of this study, no reports were available, indicating that any truck surveys were conducted in the Sacramento area.

Table 4-1: Comparison of Survey Methods for the Sacramento Truck Travel andGoods Movement Study

Survey Methods	Advantages	Disadvantages
Mail-back Survey Based on License Plates	 No disruption to traffic flow Limited exposure of survey personnel Minimal personnel requirement(2 persons per direction) Minimal euipment required(1 tape recorder per person) 	 Potentially low response rate Time lag between trip date and receipt of postcard survey form; loss of details Difficulty identifying actual driver Time consuming Bias toward owner/operator Possible difficulty reading license plates
Video Surveillance	 No disruption to traffic Limited exposure of survey personnel Permanent data record 	 High equipment cost and requirements High personnel requirements Potential for technical or equipment problems No info regarding O-D, trip purpose, freight/goods type carried. etc. Possible difficulty reading license plate from videotape
Manual License Plate Survey	 No disruption to traffic Limited exposure of survey personnel Permanent data record 	 High personnel requirements High equipment requirements No info regarding O-D, trip purpose, freight/goods type carried etc. Possible difficulty reading license plate Lots of room for human error
Automated/Elec tronic Data Collection	 No disruption to traffic Limited exposure of survey personnel Minimal equipment requirements 	 No info regarding O-D, trip purpose, freight/goods type carried. etc. Capability to obtain info of any type extremely limited(inadequate sample size likely due to limited number of active transponders) Validity/accuracy of classification info questionable Possible difficulty identifying vehicle owner for follow-up
CB Radio/ Cellular Phone	 No disruption to traffic Little or no exposure of survey personnel Moderate equipment and personnel requirements 	 No control over sample characteristics Difficulty obtaining cellular phone numbers for trucks Potentially low response rate Safety problem with drivers responding to the survey and driving at the same time
Mail-back Survey/Roadsid e Distribution	 Survey form could be designed to obtain all desired information Moderate personnel requirement(2- 4 persons per station) 	 Potentially low response rate Potential disruption to traffic Relatively high exposure of survey personnel Relatively high equipment requirements(cones, signs, etc.)
Roadside Interviews	 Able to obtain all desired information Moderate personnel requirement Better statistical control 	 Potential disruption to traffic Relatively high exposure of survey personnel Relatively high equipment requirements(cones, signs, etc.)

46

.

Survey Methodology

The study was based on data collected from travel time studies and truck surveys. Travel time studies were performed by following vehicles with cars equipped with devices to record time, speed, and distance. Most of the data collected focused on the westbound approach, although eastbound data was also collected. The data collected included truck volumes, travel times, and travel patterns for the westbound Bay Bridge and approaches. The eastbound data collection was limited to truck travel times. Truck surveys were conducted on November 20, 1985 at the westbound toll plazas (for truck lanes 13 through 17). Approximately 1,600 questionnaires were handed out between 5:00 a.m. and 7:00 p.m. for trucks with three or more axles. The questionnaire consisted of 11 questions on a prepaid postcard. Interviews were also conducted at the Port of San Francisco container terminals at Pier 80 and 94/96 between November 18 and 22, 1985. Trucks were stopped at the two locations, and drivers were interviewed outside the entry gates either on or adjacent to the weighing scales. The interviews were conducted during Port operating hours (8:00 a.m. to 5:00 p.m.). A copy of the postcard and Port questionnaires is in **Appendix XII**.

1 1 1

Survey Data

Out of the 1,600 questionnaires distributed at the Bay Bridge toll plaza, 430 were returned. This yielded a response rate of 27 percent. The Port interviews produced 1,538 samples, representing close to a 100 percent sample.

Study Findings

The San Francisco Bay Bridge Truck Travel Study produced the following findings:

- The Bay Bridge data only represent trucks with three or more axles.
- During the AM peak period, trucks are expected to encounter 15 to 20 minutes of delay at the I-80/580/880 interchange and the Bay Bridge metering signals.
- Minor delays of about 3 minutes were expected for westbound traffic during the PM peak period on the San Francisco end of the Bridge.
- and I-880 are the two major truck routes used by westbound trucks destined for the Bay Bridge. The two routes carry 49 and 42 percent of the total truck traffic, respectively.
- About 58 percent of the trucks that serve the Port of San Francisco use the Bay Bridge. However, trucks accessing the Port make up only 8 percent of the total Bay Bridge truck volume.

San Francisco Bay Area-Alameda County Truck Travel Survey

In 1991, Caltrans District 4 conducted a study to obtain information to develop a travel demand model for trucks. The study was conducted by Barton-Aschman Associates, Inc. (Schlappi, Marshall, and Itamura, 1993). A truck intercept survey procedure manual was produced (Barton Aschman Associates, Inc., 1991). The report documents and summarizes the findings and conclusions by Barton-Aschman Associates of truck travel in the Alameda County and adjacent counties in the San Francisco Bay Area. The report is part of the State Route 61/I-880 corridor study sponsored by Caltrans and the Federal Highway Administration (FHWA). The objective was to develop and evaluate alternative strategies to alleviate congestion on I-880 and improve access to the Oakland Airport and Alameda Island. Although the study focused on the SR 61/I-880 corridor in Alameda County, the model and database included the entire nine-county San Francisco Bay Area in order to consider the many truck trips that had one trip end originating, destinating, or passing through Alameda County. Using the already operating Alameda Countywide Transportation Model, the purpose of the study was to create a submodel to forecast truck travel demand to enhance the ability to monitor congestion in the study corridor by forecasting future travel demand. Two areas were identified as critical gaps in the knowledge of truck travel in Alameda county: (1) time of day patterns and (2) origin and destination data.

Survey Methodology

Four travel surveys were conducted to obtain information regarding travel patterns of trucks operating within Alameda County:

- *Truck classification counts* at 11 freeway locations along the I-80 and I-880 corridor for a five to seven-day period.
- *Truck-intercept surveys* at nine California Highway Patrol (CHP) weigh stations. Over 8,000 interviews, at four toll bridge crossings, and almost 700 postcard surveys were completed. See Appendix XIII for a sample of survey and classification count forms.
- *Employer surveys* of truck trips. A combined telephone and mail-back survey contacted 550 Alameda County employers representing 36,000 employees. The response rate was over 79 percent and produced detailed truck trip data for over 2,200 truck trips.
- *Surveys and interviews* at the Port of Oakland with terminal operators and Port planning staff. Data for almost 5,000 daily truck trips were produced.

The surveys were designed to obtain three truck travel patterns for two, three, and four or more-axle trucks:

• External conternal trips - refers to an origin or destination outside the ninecounty Bay ... region.

- Internal-external trips refers to trips that have either an external origin or destination.
- Internal-internal trips are further subdivided into garage-based trips and linked trips. Garage-based trips are trips where the truck travels from its origin to a destination and returns to its origin. Linked trips involve departure from the origin and travel to several destinations before returning to the point of origin.

The truck intercept surveys and employer interviews also asked what type of goods were being hauled. Goods were classified into ten categories:

- Agriculture
- Chemicals
- Construction
- Construction Materials
- Empty
- Manufacturing
- Miscellaneous Freight
- Retail
- Service
- Waste

The purpose was to examine freight movement and its relationship to truck travel. However, no detailed commodity data (weight or volume of goods carried) were collected.

Truck Travel Survey Data

The employer survey data represented all trucks garaged in Alameda County and was used to create the internal-to-internal trips model. Consequently, trips recorded in the employer survey with an external origin or destination was assumed to be accounted for in the intercept and bridge crossing survey data and were not included in the internal-internal model. The employer survey results found that most trips (59 percent) were *linked* trips, and 35 percent and 6 percent of the trips were *garage-based* and *internal-external* trips. Although daily internal-external trips were only 14 percent of the total truck trips, they constituted 32 percent of the total vehicle-hours traveled (VHT). The survey results also found that two-axle trucks had shorter trip lengths than four or more-axle trucks. Moreover, linked trips were approximately 30 to 50 percent shorter than garage-based trips. Among the ten goods categories carried, the study found that for internal-to-internal trips for 3 or more-axle trucks, *empty* trucks represented a large proportion of trips (40 percent), followed by retail-related trips (24 percent). For 2-axle trucks, however, retail-related trips predominated.

Truck Model

As part of the SR 61/I-880 corridor study, the Barton-Aschman study developed a truck travel demand model to forecast 24-hour and P.M. peak-hour estimated volumes for two-axle, three-axle, and four or more-axle trucks. Peak-hour volume estimates were generated using factors derived from survey data. Goods movement forecasts were not developed because of their complexity in urban areas, with multiple origins and destinations.

The travel demand forecasting process consisted of four components or submodels:

- (1) Trip generation
- (2) Trip distribution
- (3) Peak-hour factoring
- (4) Trip assignment

The model was developed using existing Bay Area highway networks, 1990 Metropolitan Transportation Commission (MTC) zonal socioeconomic data, and results of various surveys conducted in 1991 as part of the overall truck study. Truck volume counts from 1991 were used to validate the travel model.

Study Findings

The study findings produced the following conclusions:

- The peak period for truck travel occurs in the midday. The findings were consistent with other Bay Area and national studies (California Department of Transportation, 1988; Grenzeback, Reilly, Roberts and Stowers, 1990).
- Most truck trips in the San Francisco Bay Area (nine counties) are local or regional. At five CHP weigh stations and four bridge crossings, 98 percent of the truck trips surveyed had either their origin or destination in one of the nine Bay Area counties.
- Many of the approximately 5,000 daily truck trips in the Port of Oakland area are local trips that never access a freeway. Furthermore, 59 percent of truck trips to the Port were found to originate in the nine-county Bay Area. The San Joaquin Valley accounted for 19 percent of trips originating to the Port.
- To collect truck trip log data, a sample of employers is more effective than a sample of vehicles. Vehicle registration data has several disadvantages. *First*, firms with large fleets are more likely than firms with small fleets to be selected for the survey sample. This is important, because studies have found that truck travel for small firms vary widely, and firms with small truck fleets were found to make frequent short trips (Rawling and Reilly, 1987). *Second*, because many owners lease their trucks, a sample of employers showed that they are more capable than truck owners to complete the trip log. The study found that a

sample of employers was more likely to be representative of truck usage than a sample of registered truck owners.

- Most employers (68 percent) did not own or leased trucks. A survey of *business services* employers found that 97 percent of those contacted did not own or lease trucks.
- The Alameda County employer survey found that almost 30 percent of the daily truck trips were retail-related trips. Retail-related truck trips per truck was the highest. Each retail-related truck made an average of 5.3 trips per day.
- Overall, there was an average of 12.6 daily truck trips per 100 employees. Truck trip generation, however, was found to vary greatly among employment categories. Daily truck trips per 100 employees ranged from 0.9 for *business services* to 33.2 for *other employment* (which included trucking and warehousing).
- Three-axle trucks accounted for the smallest percentage of total travel (7.5 percent of trips) and the smallest portion of VHT (6.9 percent). Based on the results, the report suggested that three-axle trucks should perhaps be grouped with two-axle trucks.
- The truck model results found that daily internal-to-external trips were 14 percent of the total truck trips, yet they constituted 32 percent of the total VMT.

Inlight of the study findings, the report recommended the following:

- The origins and destinations of trips that begin and end within the study area should be geocoded to the transportation analysis zones (TAZs) rather than at the city or zip code level. This would improve the accuracy of truck trip generation models based on zonal socioeconomic attributes.
- A large sample of employers (about 1,500) or a sub-sample of employers with three or more-axle trucks would be desirable.
- The employer survey should include wholesale employers as a separate category, since wholesales generate a relatively higher number of freight trips.
- Time-of-day (24-hour), day-of-week, and seasonal variations in truck travel should be examined.

Chapter Five OTHER RESEARCH AND STUDIES OF TRUCK TRAVEL CHARACTERISTICS

Hourly Distribution of Internal Truck Trips in Urban Areas

Many state departments of transportation (DOTs) collect vehicle volume and classification data on urban highways. Typically, the short-count technique is widely used by highway and traffic engineers. This procedure utilizes information collected from short durations, say, 3-hour counts, and are expanded to represent 24-hour totals. Extensive databases exist for deriving expansion and adjustment factors for short-counts to reflect weekly or annual totals. However, little data exist for the hourly distribution of truck trips for urban areas. Moreover, little, if any, data exist for hourly distribution of truck trips on arterials.

It is known that the hourly distribution of truck trips in urban areas are significantly different than that of total vehicular traffic (Wilbur Smith and Associates, 1966). Chatterjee also believed that the hourly distribution of truck traffic varies according to the functional classification of roadway and size of trucks (Chatterjee, Wegmann, Brogan, and Phiu-Nual, 1979). Furthermore, it is reasonable to believe that truck traffic varies by geographic area within a metropolitan area. Hourly distribution of truck data from 12 urban areas in the country were examined (Chatterjee *et al.*, 1979). The results found that a relatively high degree of stability existed. The 24-hour distribution found that truck traffic was generally highest during the midday period and decreases as the evening peak period approaches.

Truck Trip Generation by Land Use

Currently, some metropolitan areas estimate commercial vehicle trips as a percentage of non-home-based trips produced and attracted in a zone. The 1961 Puget Sound Regional Transit Study (PSRTS), for example, found commercial vehicle trips to be 37.8 percent of non-home-based trips estimated in a zone and used the estimate as the regional model strategy (DKS Associates, 1994). Most metropolitan areas assume that truck trips generated in a zone is a function of employment and commercial land use activities in that zone. However, employment alone is not an accurate estimator of truck trips, because many commercial trips are also related to household activities, particularly the deliveries of goods. Early research in truck trip generation in metropolitan areas was summarized in studies by Wilbur Smith and Associates (Wilbur Smith and Associates, 1961; Wilbur Smith and Associates, 1969). In general, the studies found that commercial land uses generated the greatest number of truck trips per acre, and residential land uses generated the fewest. This finding was found to be true across different cities in the country. Another study for the Nashville Metropolitan Area Transportation Study looked at truck trip generation by weight class (Wilbur Smith and Associates, 1961). The study

52

also found that commercial land uses generated the greatest number of truck trips. Moreover, industrial land uses were found to produce a significant number of heavy truck trips and that a large number of trips for residential land uses were made by light commercial vehicles.

A more recent study done by CATS found that although commercial land uses represent only 3.5 percent of the developed land in the study area, they generated over 37 percent of the total truck trips (Zavattero, 1977). Manufacturing land uses accounted for another 12 percent of total truck trips and only an additional 4.4 percent of the developed land. Another study by Brogan also used truck trip origin-destination data from four case-study cities (Flint, Michigan; Columbus, Ohio; Kenosha, Wisconsin; and Racine, Wisconsin) to develop truck trip rates by land use and truck weight types (Brogan, 1977). The study found commercial and industrial land uses to be the greatest generators of truck trips. Trip rates for residential land uses, however, were found to vary for different urban areas.

Between 1983 and 1986, the Caltrans District 4 conducted several trip generation studies at hundreds of locations in the San Francisco Bay Area. These studies were initiated to develop and update trip end generation factors for various land uses. On-site observations, vehicle volume, vehicle classification, and vehicle occupancy counts were conducted. Truck volume counts included $1^{1}/_{2}$ -ton trucks with dual rear wheels, but excluded pick-ups and vans. Many of the results were used by the Institute of Transportation Engineers (ITE) in their *Trip Generation* manual. These studies provide yet another source of data for understanding truck trip generation by land use types.

The Florida Department of Transportation has also compiled its own commercial vehicle trip generation data. The Florida DOT conducted studies for Boward County, Palm Beach, and Tampa to determine the attraction variables and trip generation rates for trucks and taxis (DKS Associates, 1994). It used dwelling units and total employment as independent variables in its analysis. Moreover, the 1991 Phoenix Commercial Vehicle Survey also provided detailed truck trip generation rates by land use types. The results of the last three sources listed above are summarized in **Table 5-1**.

Methods for Estimating Truck Trip Matrices

List and Turnquist (1993) wrote a paper that describes a new truck trip matrix estimation technique that would allow wide variations in data input for estimating truck flows in urban areas. The method is linked to a GIS environment for data management and easy display of the results. The method was illustrated through a case study using truck data from the Bronx in New York City. The paper was drawn from a larger report, which included a second case study and additional material (List and Turnquist, 1993). The proposed technique allows data input for multiple vehicle classes. It includes three main assumptions: 1) each link has a "directional flag" (i.e., $i \rightarrow j$, $j \rightarrow i$, or both), 2) each link has a use label (truck classes), and 3) each link has a travel time which may vary by
VARIABLE	PHOENIX	BAY AREA	FLORIDA(1)
Trips per Household)			
Single Family	N/A	0.20	N/A
Multi-Family	N/A	0.06	N/A
Total	0.28	0.16(2)	0.20 - 0.40
(Trips per Employee)			
Retail	0.76	0.35 - 1.64	N/A
Office	0.33	0.07	N/A
Industrial	0.79	0.09 - 0.20	N/A
Public	0.32	0.15 - 0.30	N/A
Schools	0.32	0.19	N/A
Other	0 94	N/A	N/A
Total	0.55(3)	N/A	0.34 - 0.64

(2) Total rate based on applying single family and multi-family rates to Puget Sound household data. (3) Total rate based on applying rate by employment category to Puget Sound employment data.

(Source: DKS Associates, 1994)

the time-of-day. Furthermore, each zone is assumed to be non-overlapping, with a centroid where trips originate and terminate. The proposed technique also assumes that the types of data are of three basic types: 1) link volumes or classification counts, 2) partial origin-destination (O-D) estimates for various zones, time periods and truck classes, and 3) originating/terminating (OT) data indicating the total number of trucks originating and terminating in a particular zone or entry node on the network's periphery. The 1991 Truck Commodity Survey data, as well as 8 different data sources were used to test the new method. Three time periods (6-10 a.m., 10 a.m.-3 p.m., and 3-8 p.m.) and three truck classes (vans, medium, and heavy trucks) were considered. A total of 9 O-D matrices were estimated. The results found that heavy truck trips occurred in the PM peak period. This paper is one of the first that addresses the problem of using partial and fragmented data in estimating truck flow matrices.

Passenger Vehicle Equivalencies (VEQs)

The operational effects of trucks on freeways have been a subject of interest in the 1950s and 60s. One of the purpose was to determine the equivalency - in terms of the number of passenger cars - of trucks in the traffic stream. This practice of assigning VEQs was a recognition that commercial vehicles have unique operating characteristics such as slower acceleration and deceleration, longer turning times and a wider turning radius, and greater lane occupancy which affects adjacent vehicles. The first comprehensive set of values for such equivalencies were presented in the 1965 Highway Capacity Manual and later revised in 1985 (Transportation Research Board, 1985).

In 1964, the Puget Sound Regional Transportation Study attempted to apply passenger vehicle equivalency factors to trucks in order to estimate truck trips. A staff report documented the procedures of applying weight factors to truck trip ends when estimating their passenger vehicle trip equivalents (Cowan and Walker, 1964). The weight factors were divided into vehicle types and were based on highway capacity

studies and VEQs developed in other studies. In general, it suggested that 1 light truck was equivalent to 1 passenger vehicle, 1 medium truck to 2.5 passenger vehicles, and 1 heavy truck to 5 passenger vehicles. The report documented 1985 truck trip forecasts based on the 1961 base year estimates. The procedures were:

i ił

- 1. Identify the distribution of truck trips in commercial use by type for the base year and forecast year (assuming the same percent distribution of truck trips by type for the forecast year as for the base year);
- 2. Based on changes in historical trends, forecast changes in distribution of truck trips in commercial use by type for the forecast year;
- 3. Weight the two distributions (from steps 1 & 2) by the appropriate equivalency factors;
- 4. Divide the sum of weights by the total number of trucks for the forecast year. The result is the average passenger vehicle trip equivalency factor.

There are three apparent problems with the above methodology. First, the forecast assumes that the distribution of trucks by type does not change over time. This assumption is simplistic; for example, a CATS study projected that light and heavy trucks were forecasted to increase by 52 percent and 10 percent respectively from 1986 to 2010 (Rawling and DuBoe, 1991). Second, the assumption that the number of trips per truck remains constant for each truck type is unsupported. Furthermore, this assumption does not take into account changes in truck fleet composition, land use, and regional economic development over time that would likely change the number of trips, or trip length, each truck makes. Rawling cited that the industrial dispersion that has taken place since 1970 in northeastern Illinois has been significant and has probably contributed to longer trips overall (Rawling, 1988). Moreover, truck trips tend to vary by geographic location. The 1986 CATS Commercial Vehicle Survey showed that truck trips were typically concentrated at major industrial, commercial centers, and along certain corridors (Reilly, Rosenbluh, and Rawling, 1987; Rawling and DuBoe, 1991). The proposed methodology of converting truck trips to passenger vehicle trip equivalents is simply unvalidated by the lack of spatial and temporal data on truck travel.

For modeling purposes, the CATS 1970 and 1986 Commercial Vehicle Surveys also utilized the practice of "translating" commercial vehicles to passenger vehicle equivalents and then conduct one aggregate trip assignment (CATS, 1977; CATS, 1979; corridors (Reilly *et al*, 1987). Since the 1970s, CATS has also developed computer models that add commercial vehicle trips (truck trips) to separately-calculated passenger vehicle trips (person trips). The procedures used to achieve the model estimates are outlined fully in a 1986 CATS Working Paper (Englund and Ryan, 1986). CATS practice was not to do separate forecasts and trip assignments for commercial and passenger vehicles but to "translate" commercial vehicle trips into passenger equivalents (VEQs) before assigning vehicle trips to the highway network. This practice was initiated with the recognition that trucks are slower, wider, and longer than cars and that they affect traffic flow and the performance of all vehicles on the network. The purpose is to do one aggregate trip assignment, using passenger vehicle equivalents to represent commercial vehicles. The VEQ factor is as follows:

- 1 B truck or light commercial vehicle = 1 passenger vehicle;
- 1 medium commercial vehicle = 2 passenger vehicles;
- 1 heavy commercial vehicle = 3 passenger vehicles.

In general, the 1985 Highway Capacity Manual suggests that an urban freeway with 10 percent trucks in the traffic stream and grades below 2 percent has an impact equivalent to 2 to 4 passenger vehicles. The effect of heavy trucks was found to increase dramatically with the gradient of the roadway; however, as the percentage of trucks in the traffic stream grew, the effect of additional trucks declined somewhat. The Caltrans Urban Freeway Gridlock Study, however, revealed that research developed for the National Cooperative Highway Research Program (NCHRP) suggests that the 1985 Highway Capacity Manual VEQ estimates are still slightly overstated (Caltrans, 1988). More recent research for NCHRP suggests that trucks have an impact equivalent to 1.5 to 2.0 passenger cars. An MTC internal memorandum, however, suggested that the Bay Area freeway terrain, often with grades of 3 to 4 percent for distances of at least one-half mile, causes large trucks to have a greater impact on traffic. The MTC memorandum suggested that the "6 to 10 passenger car equivalent" should be used in analyzing truck impacts in the Bay Area and not the 1.8 to 2.8 estimates that Caltrans suggested (Metropolitan Transportation Commission, 1989). The literature suggests that there is not a complete agreement on passenger vehicle equivalencies on freeways. The research is limited and inconclusive. For the purpose of truck modeling, VEQs should be used with caution. Moreover, VEQs used should reflect factors such as the topography or grade of the freeway network, the length of different grade changes, congestion level, or whether a segment of the freeway is winding and narrow or straight and wide, etc. Further research and data collection is needed to analyze the affects of trucks on different freeway traffic conditions.

Truck Travel Demand Forecasting and Pavement Management

While there are numerous studies on the effects truck characteristics (weight, axle configurations, tire properties, braking, etc.) have on pavement design and management, there have been few studies in the literature concerning the use of truck travel demand forecasting for pavement design and management. Trucks have often been cited to create the most damage to pavements in the urban roadway system, yet many metropolitan areas have found that limited information on truck travel in their respective regions have limited their ability to relate traffic forecasts with pavement design and pavement rehabilitation programs. Similar concerns have focused on pavement deterioration associated with trucks in general and specific truck types in particular. Such concerns have obvious importance in, for example, developing pavement design and management practices, and in establishing weight restrictions and highway cost allocation policies in an urban area.

Three studies have attempted to discuss the role truck forecasts have in the pavement design process and highway cost allocation. The first report by the Minnesota Department of Transportation discussed the role truck forecasts have in the pavement design process (Hage, 1982). It examined the effects truck traffic have on pavement life. Furthermore, the report briefly discussed the weaknesses of the existing truck forecasting process for use in the pavement selection and design process. The report found that in Minnesota, the trend was to use more larger trucks. This was consistent with the general trend of more capital-intensive modes of economic activity. Second, more and more shippers were found to abandon railroad freight and rely more on trucks. Third, the report cited that the multi-lane, full-access interstate highway system has undoubtedly enhanced trucks' competitive position by shortening delivery times and reducing labor and vehicle operating costs. Moreover, 5-axle trucks were found to effect the most damages on pavement structure. Thus, it was determined that the forecasting process for this class of trucks, as well as for the entire truck fleet, needed further assessment. In particular, there were concerns over the reliability and accuracy of the forecast data in predicting pavement wear.

The first truck forecast reliability issue centered around truck volume forecasts. The report cited that the design and improvement of a pavement were made on the basis of a single vehicle classification count taken in the vicinity of a proposed project. Often, the count was neither current nor ideally located. Furthermore, the report cited that there were demonstrated deficiencies in load forecasting on the basis of a single traffic count, no matter how recent or well located. The report also stated that there were daily, weekly, and seasonal variations in truck travel that would affect pavement wear. It maintained that the existing forecasting procedures did not give adequate attention to these variations.

In summary, the Minnesota DOT report suggested that the current method of forecasting truck travel for pavement design and pavement management might be inadequate. Truck forecasts based on single vehicle classification counts at a limited number of locations may reduce the accuracy of the forecast. Furthermore, using the limited data to make 20-year truck traffic forecasts would be formidable. For metropolitan areas, understanding the origins and destinations of truck travel (where trucks enter or exit critical urban roadways) by vehicle classification, weight, etc., would not only improve truck forecasts, but improve the accuracy to the pavement design and Better truck travel data would help improve pavement management process. deterioration models. Improved heavy truck data, specifically, would help identify pavement "hot-spots," places that are most affected by damages caused by heavy commercial vehicles. Moreover, better truck travel data would likely help identify user costs to the pavement system. Being able to forecast truck volumes and truck types using a certain portion of a facility would help pavement managers better estimate the distribution of user costs of that facility.

Chapter Six COMPUTERIZED DATA COLLECTION OF TRUCK TRAVEL INFORMATION

Recently, the Intermodal Surface Transportation Efficiency Act (ISTEA) has called for improved data collection and management techniques of Intermodal Facilities. This has spurred the development of computerized data collection methods and related technologies. The Federal Highway Administration (FHWA) has recently funded several case studies that evaluated the use of handheld computer devices for collecting truck travel survey data. Current procedures use manual survey techniques where interviewers solicit responses and records them on a survey form or where respondents are given printed questionnaires to fill out and return. Telephone surveys, mailout-mailback questionnaires, or face-to-face interviews are the most common types of survey techniques. Recent studies have asked whether manual survey techniques themselves are accurate and useful in serving data needs. Compounded with rising costs of conducting surveys, several case studies were conducted to evaluate various computerized data collection methods.

Street Smarts, a transportation planning and engineering company based in Duluth, Georgia, recently tested the feasibility of using computerized data collection devices for collecting truck trip generation data.⁶⁻¹

Handheld computers, called Personal Digital Assistants (PDA's), were selected for analysis based on the following criteria:

- Should be small and lightweight;
- Have a sufficient battery life for daily survey needs;
- Have a sunlight readable display;
- Have means of easily transferring data to a computer for data aggregation;
- Have a time-of-day clock for stamping date and time records; and
- Have sufficient memory (RAM) for daily survery purposes.

Out of the 4 available PDAs evaluated - the Tandy Zoomer, the HP 95LX, the Apple Newton, and the Sharp Wizard - the Tandy Zoomer was selected as the data collection device best-suited for the desired data collection activity. The approximate price of the Tandy Zoomer is \$599 per unit. See **Appendix XIV** for a description of the price and characteristics of each tested PDAs and a photo of the Tandy Zoomer. Some of its features include:

- Touch screen capability
- Character recognition

⁶⁻¹ Information obtained from an information packet sent from Street Smarts to MTC staff.

- PC Compatible processor
- Big display of characters.

The software developed for the data collection survey form allows the interviewer to make observational entries as the interviewee approaches. It allows the interviewer to enter the answers to interview questions by selecting the desired answer options. The observation and interview entries were also designed to have menu selection responses as often as possible.

Recently, Street Smarts was contracted by the Federal Highway Administration (FHWA) to use handheld computer data collection devices to conduct truck origindestination and commodity surveys. The purpose of the project was to evaluate existing technologies most appropriate for computerized data collection. The project objectives were to test and develop data collection methodologies and software for computerized data collection at Intermodal Facilities. After the Tandy Zoomer was selected as the most appropriate handheld device, a survey instrument was programmed into the Zoomer as a sample data collection form and used in 3 case studies in Georgia, Florida, and Texas.

In the first case study, a truck survey was conducted at a truck weight and inspection station on I-85 in Georgia. Due to adverse weather conditions, only 131 surveys were collected. Drivers were chosen randomly and directed to park their trucks at the weight station for the interview. This first case study provided information on possible modifications of the existing survey form. Specifically, more precise selections for some commodities and some pickup and delivery choices were identified.

The second case study was at a weight station on I-4 in Plant City, Florida. At this location, 465 surveys were conducted. Four truck parking spaces were used as survey stations. The third case study was conducted at a weight station on I-10 near Houston, Texas. During a 6-hour period, 345 surveys were completed.

In comparing the computer survey technique to manual methods, several conclusions were drawn. First, computer survey tools enhance the accuracy of the data collected because of reductions in the number of internal errors in responses, the use of pick lists, and the field review of data records. Second, computer survey tools also reduce the collection time for each survey by 1) automatically recording the starting and end times of each survey, 2) calculating the data collection time, and 3) creating data files that do not have to be re-entered before it can be processed. The manual survey requires that all data be re-entered before processing. Thus, the data entry process significantly increases the chance of error. Third, computerized data collection reduces the cost of data collection by abolishing the need for data entry and transfer. Moreover, it allows the survey instrument to be changed easily without having to reprint new forms each time a change is made. Lastly, handheld computer devices replaces the need for interviewers to carry extra pencils, erasers, survey forms, or clipboards. This increases the ease of conducting the survey in the field.

Chapter Seven CONCLUSIONS

THE NEED FOR TRUCK TRAVEL DATA

Truck travel and goods movement is essential to the economic vitality of an urban area. Trucks not only act as the "supply-line" from warehouses to points of consumption (whether they be retail stores or residences), they connect intermodal freight facilities (seaports, airports, railroad and freight terminal facilities). As important as truck travel is to a region, it also has negative effects such as traffic congestion, truck-related accidents, air pollution, noise, and pavement deterioration. In order for an urban area to undertake comprehensive truck/freight planning, accurate and reliable truck travel data is needed for analysis.

Truck travel data for urban areas are important for the following reasons:

In terms of metropolitan area planning needs, improved truck travel data could be used for the following purposes:

- 1. Truck travel model development
 - Truck trip generation
 - Origin and destination analysis
 - Local and freeway route assignments
 - Congestion and speed simulations
 - Travel time analysis
 - Analyze impact of toll facilities
 - Spatial and temporal (time-of-day, day-of-week, and season) analyses
- 2. Corridor/Route analysis
 - Evaluate route/corridor traffic management proposals for freight impacts
 - Provide information on truck travel to formulate traffic management plans during roadway reconstructions
 - Assess impact of truck route reassignments or closures
- 3. Air quality modeling
 - Estimate truck emissions
- 4. Intermodal Freight Planning
 - Facilitate seaport planning
 - Facilitate airport planning
 - Understand competition and demand of different freight modes
 - Provide data to develop performance measures for Intermodal Management Systems as required under ISTEA

- 5. Pavement Management Planning
 - Evaluate and design road geometrics
 - Help calibrate pavement deterioration models
- 6. Truck restrictions and enforcement
 - Route restriction analysis
 - Dangerous goods movement regulation and enforcement analyses
 - Truck driver safety programs
- 7. Facilitate public-private partnership
 - Open dialog with private freight industries in gathering data
 - Provide truck travel data to public and the freight industry for research and analysis
 - Freight-economic analysis

TRUCK SURVEYS & TRUCK TRAVEL DEMAND FORECASTING LITERATURE

This literature and state-of-the-art review reveals that few urban areas in the country have had extensive experience in conducting truck surveys and truck travel demand forecasting. Most metropolitan planning organizations (MPOs) or regional transportation planning agencies continue to generate their truck trip estimates based on origin-destination surveys conducted in the 1960s and 70s. In the last ten years, only a few metropolitan areas, namely Chicago (1970 and 1986), Ontario (1978, 1983, and 1988), Vancouver (1988), Phoenix (1991), Alameda County, California (1991), New York-New Jersey (1991-94), El Paso, Texas (1994), and Houston-Galveston (1994) have undertaken significant efforts to collect truck travel data or develop new techniques in forecasting truck traffic. Out of the eight urban areas, only Chicago and Phoenix have had their truck model development and methodologies documented in detail, and only Ontario and the Port Authority of New York and New Jersey have systematically and continuously collected truck travel data.

TYPES OF DATA COLLECTED

The eight most recent truck travel surveys all collected origin-destination information (see **Table 7-3**). With the exception of roadside surveys conducted in New York and New Jersey, most truck surveys requested land use at destination and truck odometer readings from the respondents. Most surveys seemed to classify trucks by weight, number of axles, or by truck type. This is reflected in the collection of one or more of the above truck classification information. With the except of roadside surveys, all other survey types included trip diaries that required respondents to either report over the phone or to fill out and return. Most surveys that included trip diaries (Chicago, Phoenix, El Paso, Houston-Galveston, and Alameda County) also collected data on trip start and end times. The commodity data collected ranged from a simply classification of commodity by type to detailed description of the actual commodity being carried. The Ontario survey is the only commercial vehicle survey that gathered information on truck driver characteristics. The El Paso commercial truck survey was the only survey that collected route choice information for the surveyed trip.

USES OF TRUCK SURVEY DATA

The most common uses of truck data are for regional truck travel model development and corridor/route analysis. Chicago, Phoenix, El Paso, and Vancouver have used their truck survey data to develop regional truck travel demand models. Ontario has seen the most use of its truck survey information. The truck data have been used for time series comparisons, evaluation of road design and geometrics, pavement management planning, truck-related accident analysis, dangerous goods movement regulation and enforcement, understanding truck driver characteristics and for planning truck driver education programs. El Paso has mainly used its truck data for regional travel and truck emissions modeling. The Port Authority of New York and New Jersey has used its truck data for traffic management purposes during highway and bridge/tunnel reconstructions. It has also used the truck data for extensive freight-economic analysis. Chicago has used its truck data to generate truck activity maps of the Greater Chicago region. Truck speed simulations and modeling the effects of toll facilities on truck route choice have also been analyzed.

TRUCK TRAVEL SURVEY METHODS USED

The most common survey method for conducting truck travel surveys in urban areas was the combined telephone-mailout-mailback method. Three urban areas in the country - Phoenix, Alameda County, California, and Houston-Galveston - have recently conducted truck travel surveys using this method. The most common source for drawing the survey sample is the Department of Motor Vehicle (DMV) registration files. Other sample sources include lists of truck registration files available from commercial vendors (R.L. Polk, Texas Vehicle Information and Computer Services, Inc., etc.). However, to collect truck trip log data, a sample of employers is more effective than a sample of registered vehicles. Vehicle registration data has several disadvantages. First, firms with large fleets are more likely than firms with small fleets to be selected for the survey sample. Studies have found that truck travel for small firms vary widely, and firms with small truck fleets were found to make frequent short trips (Rawling and Reilly, 1987). Second, because many owners lease their trucks, a sample of employers showed that they are more capable than truck owners to complete the trip log. The 1991 Barton Aschman study of truck travel in Alameda County found that a sample of employers was more likely to be representative of truck usage than a sample of registered truck owners (Schlappi, 1993).

The second most used survey method was the *roadside interview* method. The Province of Ontario, Canada and the Port Authority of New York and New Jersey have

conducted numerous roadside interviews. In general, the combined telephone-mailoutmailback survey method is more cost-effective and yields a reasonably high response rate. Roadside interviews produce very high response rates with complete information. Moreover, they are ideal for cordon surveys or surveying trucks traveling in from outside the survey area. A summary of different survey characteristics for eight urban truck travel surveys is found in **Table 7-1**, and a summary of different truck travel survey methods (typical response rate, advantages, and disadvantages, etc.) is found in **Table 7-**2.

NUMBER OF COMPLETED SURVEYS AND RESPONSE RATE

The approximate number of completed surveys from the eight urban truck surveys varied from 188 to 19,225. In general, roadside surveys produced the highest number of completed surveys and the best response rate (nearly 100 percent). Mailout-mailback surveys produced the lowest overall and item response rates (See **Table 7-1** and **Table 7-2**).

SURVEY COST

Telephone interviews are the most costly to conduct. They require a large amount of staff and time for data collection. "Phone-tagging" is a common problem which is time-consuming for this survey method. Combined telephone-mailout-mailback surveys are the most cost-effective to conduct. They yield reasonably high response rates over mailout-mailbacks alone. The phone contact portion of the survey can also help to assess non-response bias when analyzing and weighting the mailback survey samples. A 1994 combined mailout-mailback survey conducted in Houston-Galveston cost \$167 per completed survey.

COMPARISON OF SURVEY FINDINGS

A summary of the general findings from various truck travel surveys are as follows:

Characteristics of Commercial Vehicles

Average Vehicle Weight

The only survey that reported average vehicle weight was the 1991 Phoenix Commercial Vehicle Survey. The average vehicle weight per commercial trip was 11,870 lbs.

	Ta	ble 7-1: S	ummary	of Truck	Travel Surveys in Urba	an Areas	
Survey Location	Survey Year	Survey Method	Approx. Number of Completed Surveys	Response Rate	Data Applications	Total Survey Cost	\$/Survey
Chicago	1986	Mailout- Mailback	3,506	25.3%	 Truck travel model development Corridor/Route analysis Effects of toll on trucks Truck speed simulation model Truck activity mapping 	\$ 200,000	\$57/survey
Ontario	1988	Roadside Interview	19,225	96.5%	 Time series comparison Evaluate & design road geometrics Pavement management planning Truck accident analysis Dangerous goods regulation & enforcement analysis Truck driver characteristics Driver education program 	NA	NA
Phoenix	1991	Combined Telephone- Mailout- Mailback	720	30.0%	 Truck travel model development 	\$90,000(1)	\$125/survey(1)
N.Y. & N.J.	1991	Roadside Interview	4,500	NA	 Evaluate dedicated route/corridor proposal Traffic management for highway reconstruction Time series freight analysis Freight-economic analysis 	NA	NA
Alameda County, Calif.	1991	Combined Telephone- Mailout- Mailback & Roadside	2,200 over 8,000	79% NA	 1-880 corridor analysis Create truck travel submodel for corridor analysis Generate 24-hour & PM peak volumes by axle 	\$285,000(2)	NA
N.Y. & N.J.	1992-94	Interview Roadside Interview	14,671	37.8%(3)	NA	\$312,000(4)	\$21/survey
El Paso	1994	Telephone Interview	188	42.6%	 Truck travel model development Part of regional travel study Truck emissions analysis 	\$65,000(5)	\$345/survey(6)
Houston- Galveston	1994	Combined Telephone- Mailout- Mailback	900	35%-40%	 Truck travel model development 	\$150,000	\$167/survey
 Cost i The c \$5,00 This v This v This v Metro 18 loc Cost i Cost i 	included data sost included 0 was also in was the samp was a multi ppolitan Trar cations with included sam igher cost w	Mailout- Mailback a collection, da I sample design neluded in the t pling rate. No r -agency effort, sportation Cou 3 interviewers pple design, sur	ta coding, and r h, survey design otal cost for con esponse rate wa with partnersh ncil (NYMTC) per toll plaza fo vey design, data number of inc	nodel develop h, data collect nducting vehic as given. hip from the 1 , and the Port r 24 hours. a collection, c omplete succe	ment. ion, coding, data reporting, and mou- le classification counts at 11 locatio New Jersey Department of Transpo Authority of New York and New Jer poding, reporting, survey analysis, an	del development ns along 1-80 an ortation (NJDO) sey. The survey d model develop	. Approximately, d I-880. (7), the New York was conducted at oment.

Survey Methods	Piace (year) of Surveys	Completed Surveys (% of Total Pop.)	Typical Response Rate		Advantages	Disadva	ntages
Telephone Interview	N.Y.(1964) Calgary(71) El Paso(94)	4%-15%	40%-50%	•	High response rate Easy to follow-up	 Can only cal hours "Phone-tagging Limited time respondent is b Requires acc registration file 	during busine "problem on phone usy cess to vehice
Mailout- Mailback	Chicago(86)	1%-5%	10%- 45%(1)	•	Less costly Good response rate w/ certified mail Only follow-up of non- responses is necessary	 Low overall & Possible bias response drivers/owners Low response owners Low response trucks Need to fol responses Difficult to ens will fill out th the owner or receives the su Requires acce file 	item response rai due to bett from sor from small tru from out-of-sta low-up on no sure that the driv e form, instead fleet manager w rvey forms ess to registrati
Combined Telephone -Mailout- Mailback	Phoenix(91) Houston(94) Alameda, CA(91)	3%-10%	30%- 80%(2)	•	Improved response rate over mailout-mailback alone Early identification of owners who agree to participate & potential non- responses through phone contact Phone contact may help adjust sample size for mailout-mailback	 Same disadvar survey method High cost of te Need phone n diary More costly the 	tages as telepho above lephone follow-u reminders for t an above method
Roadside Intercept/ Interview	Calgary(71) Ontario(78, 83, 88) N.Y. & N.J. (74, 82, 85, 91-94) Alarneda County, CA (91)	8%-35%(3)	95%- 100%	•	Complete information High response rate Better sampling control Good representative sample of trucks entering or leaving a cordon line Easy comparison with mainstream traffic through field counts at survey location	 Potential disru Quality and a affected by we Hazardous to s Time constrain No follow-up p Enforcement p Drivers avoid station Only represent road along signification 	ption to traffic conduct of surv ather, lighting urvey crew it cossible roblems ding the surv trucks traveling urvey station, t

ił

.

1

-

.

ì

• • 4.1

(2) The higher response rate was due to better survey participation non-range duck fleet operators.
 (2) The higher response rate was due to an employer survey conducted in California (1991 Caltrans-Alameda County Survey).
 (3) The higher percentage is from the 1988 Ontario survey which surveyed 57 locations over a 1,855-hour period throughout the Ontario Province.

Survey Location	Survey Year	Survey Method	Sample Source	Weight	Axle	Truck Type	O-D	Odometer	Commodity	Land Use	Driver Info	Route Info	Develop Model
Chicago	1986	Mailout- Mailback	DMV	•		٠	•	•		•			Yes
Ontario	1988	Roadside Interview	Road- side(1)	•	•	•	•	•	•		•		
Phoenix	1991	Combined Telephone -Mailout- Mailback	DMV	•	•	٠	•	•		•		-	Yes
N.Y. & N.J.	1991	Roadside Interview	Toll Plaza		•	٠	•		•				
Alameda County, Calif.	1991	Combined Telephone -Mailout- Mailback & Roadside Interview	DMV, Port of Oakland Road- side(1)		•		•	•	•	•			Yes
N.Y. & N.J.	1992- 1994	Roadside Interview	Road- side(1)		•	•	٠		•				
El Paso, Texas	1994	Telephone Interview	TVICS(2)		•	•	٠	•	•	•		•	Yes
Houston- Galveston	1994	Combined Telephone -Mailout- Mailback	DMV			•	•	•	•	•			

Truck Size

The share of different truck sizes used varied from urban area to urban area. The preliminary findings from the 1991 George Washington Bridge survey in New York-New Jersey seemed to suggest a shift towards using more large trucks. The 1991 New York-New Jersey Truck Commodity Survey, however, found that truck size was evenly split between small and large trucks.

Characteristics of Commercial Vehicle Trips

Trips by Vehicle Type

Light trucks make more trips than any other truck types. The 1991 Phoenix survey found that 96.6 percent of all commercial trips were made by the two lightest weight categories. In Los Angeles, San Francisco, and San Diego (1985-87), large trucks accounted for three-quarters of all truck travel (except travel by pick-ups and panel trucks).

Average Trip per Commercial Vehicle

Light trucks have a higher average trip frequency than for heavy trucks. In Phoenix, the trucks surveyed made on average 7.7 trips per day. Light trucks made on average more trips (12.1 trips) than heavy trucks (4.7 trips).

01 C

Regional vs. Through Trips

The 1986 CATS survey found that of all the heavy trucks, almost 9 percent were found to be "in transit", i.e., leaving the region. Of all the commercial vehicle trips in the "in transit" category, 44.8 percent were made by heavy trucks. The 1991 New York-New Jersey Truck Commodity Survey found that the majority of truck traffic served the region. Only about 7 percent of the eastbound truck traffic at the 6 crossings were passing through. In California, the 1991 Barton-Aschman study found that most truck trips in the San Francisco Bay Area (nine counties) are local or within-region. At five CHP weigh stations and four bridge crossings, 98 percent of the truck trips surveyed had either their origin or destination in one of the nine Bay Area counties.

Average Trip Length

Heavy trucks make longer trips. The 1986 CATS survey found that heavy trucks made, on average, 24.9 miles per trip, compared to 9.6 miles for light trucks. The survey also found 7.5 percent of heavy trucks made an average trip length of over 60 miles. In Phoenix, vehicles in the heaviest weight category averaged 33.4 miles per trip, compared to 11 or less miles per trip for the lighter weight categories.

Vehicle Miles Traveled

Heavy trucks log a higher VMT per day than light trucks. The 1991 Phoenix survey found that the average VMT per vehicle for the 64,000+ lb category was 156.8 miles, compared to 56.2 miles for the 8,000-28,000 lb category. The Caltrans Urban Freeway Gridlock Study (1988) found that wholesale trade, durable goods manufacturing, and non-durable goods manufacturing industries generate the most miles of large truck travel. Together, they generate almost 90 percent of all truck miles of travel. Moreover, private truck fleets owned by businesses accounted for about half of all truck miles of travel. Most of their trips are short-haul trips (less than 200 miles). Common (for hire) carriers accounted for the other half of all truck miles of travel. About one-third of their trips are short-haul, and two-thirds are long-haul (over 200 miles).

Time of First Commercial Vehicle Trip

The 1991 Phoenix survey found that most trucks started their first trip between 6:00 a.m. and 9:00 a.m. This pattern, however, varies by weight category. Light trucks were more likely to start their first trip between 6:00 a.m. and 9:00 a.m. Heavy trucks (51.8 percent), however, started their first trip before 6:00 a.m.

Time-of-Day Distribution

Most truck trips seem to occur during the midday period. In Phoenix, the peak period for truck travel occurred between 9:00 a.m. and 2 p.m. The 1991 Barton-Aschman Study of Alameda County also found truck trips converging during the midday. In Ontario, the overnight, off-peak period (8:00 p.m. - 6:00 a.m.) had the largest percent of "through" traffic. Only 8.7 percent of the morning peak truck traffic was "through" traffic, and 13 percent for midday. In California, the Caltrans Urban Freeway Gridlock Study cited that large truck travel was highest during the midday off-peak period, comprising 5.5 percent of all vehicles in the Los Angeles and San Francisco areas, and 2.5 percent of all vehicles in the San Diego area. In the San Francisco area, truck volumes on I-580, east of I-880, were highest during midday. Truck flow on I-880/I-80, south of SR-24, also had the heaviest volume during the midday, from 6.8 to12.0 percent of all vehicles. Truck traffic for this segment of the freeway was heaviest northbound in the morning and southbound during the evening.

Truck Travel During Peak Periods

The 1991 Phoenix survey found that peak period for truck travel occurred between 9:00 a.m. and 2 p.m. The results, however, vary by individual location. Heavy trucks, however, have a shorter peak period (11:00 a.m. to 2 p.m.); 13 percent of total daily commercial vehicle travel for heavy trucks occurred between this period. In Ontario, truck volume in proportion to the total traffic volume was found to be greatest during the early morning hours (3:00 to 5:00 a.m.). The survey also found that through traffic seemed to avoid the peak period. In New York and New Jersey, over 35 percent of trucks made trips during the morning peak period (6:00 a.m. to 10 a.m.). In comparison with AM and PM peaks for private vehicle travel, the results found that the AM peak period travel was as important for commercial vehicles as for private vehicles.

Truck Travel During Peak Periods as Percent of Total Vehicular Volume

In New York and New Jersey (1991), trucks at the 6 crossings represented an average of 11 percent of the total peak period vehicular traffic. Large trucks accounted for 45 percent of the peak period truck travel, and over 35 percent of trucks made trips during the morning peak period (6:00 a.m. to 10 a.m.). In Ontario (1988), trucks were found to constitute 17 percent of the total vehicular traffic. The 1988 Caltrans Urban Freeway Gridlock Study found that few highly congested freeways in California have a high proportion (more than 10 percent) of large trucks in the traffic stream. At 90 percent of the 78 sites sampled, large trucks were 9 percent or less of all vehicles. I-5, I-605, I-710, and SR-60 in Los Angeles and I-80, I-580, and I-880 in San Francisco were identified to be highly congested with more than 10 percent of the traffic comprised of large trucks.

Day-of-Week Distribution

Truck traffic typically occurs on the weekday. A 7-day classification count for the 1988 Ontario survey found that truck traffic was highest on Wednesdays, and Sundays produced the least amount of truck traffic.

1

7 1 H

Average Trip Duration

The 1991 Phoenix survey recorded that the overall average trip time for truck travel was 28.1 minutes. Trip time generally increased with vehicle weight.

Truck Travel by Facility Type

Few surveys or studies have attempted to analyze truck trips based on facility type used. Only the Canadians used facility types to classify their truck trips. The 1991 Barton-Aschman Study also attempted to collect data on facility type used for each truck trip. The result found that many of the approximately 5,000 daily truck trips in the Port of Oakland area are local trips that never access a freeway.

Route Choice for Return Trips

The only survey that analyzed route choice for return trips was the 1991 New York and New Jersey Truck Commodity Survey. It found that 73 percent of the truck drivers interviewed in the toll direction indicated that they would use the same route for the reverse trip.

On-Street Stops

The 1991 Phoenix survey was the only to report the number of on-street stops made by trucks. The results found that over one-third of all commercial vehicles stops were made on-street. Light vehicles made half of their stops on-street.

Commercial Vehicle Trips and Land Use

Trips by Land Use

Light trucks make more residential trips than any other truck category. The 1991 Phoenix survey found that trucks in the medium weight category made significant proportion of trips to residential areas. These trips were speculated to be largely solid waste pick-ups. Heavy trucks in the survey were also found to make high proportion of trips (26.7 percent) to residential land uses. Analysis showed that the reason could be that heavy trucks, to a large extent, were used to delivery construction materials, including lumber and ready-mixed concrete to residential construction sites. The 1991 Phoenix survey also found that of the total number of vehicles used for commercial purposes, 41.9 percent were also used for travel between home and work. The 1986 CATS survey found that light truck trip ends dominated all land uses, except for the landfill and in-transit categories. Retail attracted many more light and medium truck trips. The survey also found a strong relationship between heavy trucks and terminal/warehouse land uses; 42.6 percent of heavy trucks were destined to terminal/warehouse facilities. The 1991 New York-New Jersey Truck Commodity Survey found that most trucks were traveling within the region and to other warehouses or points of consumption within the region (retail stores, residences).

Facility of Trip Origin

In New York and New Jersey, warehouses were the predominant points of origin -45 percent of trucks originanted from warehouses. Of the nearly one quarter that originated from factories, nearly 60 percent came from factories within the region.

Activities at Trip Ends

The 1991 Phoenix survey found that light trucks were heavily used for service delivery and personal business. Heavy trucks were most used for loading and unloading cargo at their trip ends.

Other Truck-Related Findings

Truck Travel and Dangerous Goods Movement

The Ontario survey (1988) was the only survey that obtained information on dangerous goods movement. It found that a total of about 5 to 6 percent of all truck trips surveyed involved the carrying of dangerous goods. Flammable liquids (47 percent) were the most frequently transported dangerous goods, followed by compressed gases (24 percent), and corrosive substances (20 percent).

Truck-Related Accidents

The 1988 Caltrans Urban Freeway Gridlock Study found that 5 to 10 percent of all truck-related incidents were found to cause major incidents which closed two or more freeway lanes for at least two hours.

RECOMMENDATIONS

In light of the study findings, this report recommends the following for conducting a regional truck travel survey and truck travel demand forecasting:

Н

Survey Conduct

- For internal-to-internal or internal-to-external truck trips, draw the survey sample from the DMV registration file. Conduct either a telephone or mailout-mailback survey, or a combination of both to obtain a better response rate.
- For external-to-internal or external-to-external truck trips, conduct roadside intercept surveys at various facilities and links in the network. The best places to conduct them are "weigh-in-motion" stations. This would minimize traffic delay for the mainline and would be safer for the survey crews compared to conducting the survey at the roadside.
- Consider conducting intercept surveys at bridge toll plazas. For a better explanation of how to conduct roadside surveys at toll plazas, review the experiences in New York and New Jersey.
- For roadside interviews or cordon surveys, conduct vehicle classification counts at the same time and at the same location where the actual survey/interview is conducted. This will provide the basic information for sample expansion and analysis.
- For obtaining trip diaries, using a combination of fleet-employer samples and truck unit samples is desirable. Sub-sampling fleet employers will provide better sample control and reduce the problem of oversampling large fleet operators.
- Oversample smaller or individual truck operators. The 1986 CATS survey has shown that large fleet operators tend to respond better (more manpower, time, or incentive to reply to surveys) and smaller operators tend to yield higher non-responses.
- To reduce the cost of conducting a full-scale truck survey, consider making the survey a multi-agency effort.
- Consider soliciting the help of private freight/trucking agencies or organizations. Open a dialog with interested parties to facilitate cooperation and to request assistance. Working partnerships with the freight/trucking industry will help in designing and testing pilot surveys and in collecting data that would meet the needs of the private sector, as well as the public sector.

Truck Travel Analysis

- Time-of-day (24-hour), day-of-week, and seasonal variations in truck travel should be examined.
- Analyze trips by facility types used (include questions that obtain facility type information for each trip).

- Conduct further analysis on the impact trucks have on peak period congestion. Several surveys (New York, New Jersey, and Ontario) have found that in comparison with AM and PM peaks for private vehicle travel, AM peak period travel was as important for commercial vehicles as for private vehicles.
- Estimate total truck hours of delay by facility to help reduce truck operating cost.
- Conduct further analysis on the impact of truck traffic on pavement, especially the impact of waste-refuse trucks and buses (considered as "passenger-carrying trucks") on residential arterials and streets.
- The origins and destinations of trips that begin and end within the study area should be geocoded to the transportation analysis zones (TAZs) rather at the city or zip code level. This would improve the accuracy of truck trip generation models based on zonal socioeconomic attributes.
- Exercise extreme caution when using or applying vehicle equivalency factors (VEQs) in truck travel analysis.

BIBLIOGRAPHY

San Francisco Bay Area Studies

- 1. Barton Aschman Associates, Inc. Truck Intercept Survey Procedures Manual. Prepared for: Caltrans Alameda County. Barton Aschman Associates, Inc., March 1991.
- 2. Barton Aschman Associates, Inc. Proposal to Prepare a Truck Travel Forecasting Model, Route 61/I-880 Corridor. Prepared for Caltrans District 4. Barton Aschman Associates, Inc., January, 1991.
- 3. California Department of Transportation. San Francisco Oakland Bay Bridge Truck Travel Study. California Department of Transportation, 1986.
- 4. Metropolitan Transportation Commission. Caltran's Urban Freeway Gridlock Study: MTC Expanded Summary. Memorandum to Work Program and Plan Revision Committee from the Executive Director. Metropolitan Transportation Commission, Planning Section, March 1989.
- 5. Meyer, Isaiah and Emory Stoker. *Truck-Taxi Survey Manual: Field and Office Procedures*. Bay Area Transportation Study Commission, 1968.
- 6. Schlappi, Mark L., Roger G. Marshall, and Irene T. Itamura. "Truck Travel in the San Francisco Bay Area." Paper presented at 72nd Annual Transportation Research Board Meeting, Washington D.C., January 1993.

Los Angeles Area

- 7. City of Los Angeles, Department of Transportation. Los Angeles Downtown Cordon Counts. Bureau of Transportation Planning, LADOT, 1984.
- 8. Southern California Association of Governments. An Improved Methodology for Estimating Heavy Truck VMT. Southern California Association of Governments, December 1989.
- 9. Southern California Association of Governments. Port of Hueneme Access Study, Phase I - Final Report. SCAG, June 1988.
- 10. Southern California Association of Governments. Truck Movement Study, SCAG, 1988.

- 11. Southern California Association of Governments. Urban Goods Movement Study -Working Paper VI: For-Hire Truck Freight Bill Survey, SCAG and the California Department of Transportation, 1979.
- 12. California Department of Motor Vehicles. Statistical Record on Motive Power: Body Type and Weight Division for Automobiles, Motorcycles, Commercial Trucks and Trailers. California Department of Motor Vehicles, Sacramento, 1987.
- 13. California Department of Transportation. Urban Freeway Gridlock Study: Technical Report. Prepared by Cambridge Systematics, Inc. California Department of Transportation, 1988.

Chicago Studies

- 14. Chicago Area Transportation Study. 1970 Travel Characteristics: Commodities & Commercial Vehicles. CATS #372-49, January 1977.
- 15. Chicago Area Transportation Study. *Travel Forecasting Process*. CATS Technical Report #2344-01, April 1979.
- Englund, Dean B. and Eugene V. Ryan. "An Alternative Approach to Performing Selected Link Analysis." Working Paper 86-1, Chicago Area Transportation Study, 1986.
- 17. Rawling, Gerald F. "Information Needs for Policy, Planning and Design." *Goods Transporation in Urban Areas*. Proceedings of the Fifth Conference sponsored by the Engineering Foundation, Santa Barbara, California, March 1988.
- Rawling, Gerald F. "Reconstruction Under Traffic: A Study of Some Effects of the Dan Ryan Project on Motor Carrier Operations in Northeastern Illinois." Paper presented to the Goods Transportation in Urban Areas Conference, Santa Barbara, California, 1988.
- 19. Rawling, Gerald F. "Urban Goods, Movement Planning in the Chicago Region." Paper presented at the 67th annual meeting of the Transportation Research Board, Washington D.C., January 1988.
- 20. Rawling, Gerald F. and Robert DuBoe. "Application of Discrete Commercial Vehicle Data to CATS' Planning and Modeling Procedures." *CATS Research News*, Chicago Area Transportation Study, Spring 1991.

at

- Rawling, Gerald F. and John P. Reilly. "CATS Commercial Vehicle Survey of 1986: A Discussion of Project Management Issues." CATS Research News, Chicago Area Transportation Study, February 1987.
- 22. Rawling, Gerald F. and John P. Reilly. "School Buses Surveyed as a Discrete Vehicle Group." *CATS Research News*, Chicago Area Transportation Study, February 1987.
- 23. Rawling, Gerald F. and Eugene Ryan. "The Role of a Commercial Vehicle Survey in Transportation Planning for Northeast Illinois." CATS Research News, Chicago Area Transportation Study, February 1987.
- 24. Reilly, John P., Arnold Rosenbluh, and Gerald F. Rawling. "Factoring and Analysis of the Commercial Vehicle Survey." *CATS Research News*, Chicago Area Transportation Study, February 1987.
- 25. University of Illinois. *Motor Freight Planning for the Chicago Region*. Transportation Planning Group, Department of Civil Engineering, University of Illinois, Champaign, Urbana, 1983.
- 26. University of Illinois and the State of Illinois, Institute of Natural Resources. *Direct* and Indirect Emissions Production by Urban Truck Movement in the Chicago Region. Department of Civil Engineering, University of Illinois, Champaign, Urbana, and the State of Illinois, Institute of Natural Resources, Chicago, Illinois, 1981.
- 27. Zavatero, David A. Commercial Vehicle Trip Generation in the Chicago Region. (Unpublished report), 1977.

Other Studies

- 28. Barton-Aschman Associates, Inc. El Paso Urban Area Travel Study, Commercial Truck Travel Survey, Draft Report. Prepared for the City of El Paso Metropolitan Planning Organization and the Texas Department of Transportation, October 1994.
- 29. Blower, Daniel and Leslie C. Pettis. National Truck Trip Information Survey: UMTRI Truck Study. The University of Michigan Transportation Research Institute, March 1988.

- Brogan, James D. "Development of Truck Trip-Generation Rates by Generalized Land Use Categories." *Transportation Research Record.* Number 716, Transportation Research Board, Transportation Research Council, Washington, D.C., 1979, pp.38-43.
- 31. Cambridge Systematics, Inc. *Final Report: Phoenix Urban Truck Travel Model Projects*. Arizona Department of Transportation, 1991.
- 32. Chatterjee, Arun, Frederick J. Wegmann, James D. Brogan, and Kunchit Phiu-Nual. "Estimating Truck Traffic for Analyzing UGM Problems and Opportunities." *Institute of Transportation Engineers Journal*. ITE, May 1979, pp.24-32.
- 33. City of Calgary Transporation Department. Home, Truck, Taxi and External Cordon Interview Survey Procedures: calts update, data collection. City of Calgary Transporation Department, 1973-1975.
- 34. Cowan, Gary R. and John R. Walker. Staff Report No. 8: Conversion of Commercial Truck Trip Ends to Vehicle Trip Equivalents. Puget Sound Regional Transportation Study, Seattle, Washington, 1964.
- 35. DKS Associates. Travel Model Development and Refinement Trip Generation, Final Report. Prepared for the Puget Sound Regional Council. DKS Associates, June 1994.
- 36. Federal Highway Administration. Urban Origin and Destination Surveys. U.S. Department of Transportation, Federal Highway Administration, revised 1975.
- 37. Gillis, William R., Kenneth L. Casavant, Dolly Blankenship, and Charles E. Howard Jr. "Survey Methodology for Collecting Freight Truck Origin and Destination Data." Paper presented at the Annual Meeting of the Transportation Research Board, January 1995.
- Gorys, Julius. "1988 Ontario Commercial Vehicle Survey." *Transportation Research Record*. Number 1313, Transportation Research Board, National Research Council, Washington, D.C., 1991, pp.20-26.
- 39. Gorys, Julius and Greg Little. "Characteristics of Commercial Vehicle Drivers in Ontario." *Transportation Research Record*. Number 1376, Transportation Research Board, National Research Council, Washington, D.C., 1992, pp.19-26.

i.

a condi

- 40. Greater Vancouver Regional District. *Trucking in Greater Vancouver: Demand Forecast and Policy Implications*. Transport 2021, Technical Report 7. Greater Vancouver Regional District, Vancouver, British Columbia, August 1993.
- 41. Grenzeback, L.R., W.R. Reilly, P.O Roberts, and J.R. Stowers. "Urban Freeway Gridlock Study: Decreasing the Effects of Large Trucks on Peak-Period Urban Freeway Congestion." In *Transportation Research Record 1256*, TRB, National Research Council, Washington, D.C., 1990, pp.16-26.
- 42. Hage, Robert J. *Truck Forecasts and Pavement Design*. Minnesota Department of Transportation, Planning Division, 1982.
- 43. Indian Nation Council of Governments. Survey of Truck Travel Estimation and Simulation Methodologies. Indian Nation Council of Governments, Planning Services Division, Tulsa, Oklahoma, 1990.
- 44. Indiana State Highway Commission. Truck-Taxi Survey: Prepared for General Use, All Planning Areas, State of Indian in Cooperation with the U.S. Department of Transportation, Federal Highway Administration, Bureau of Public Roads. Indiana State Highway Commission, Division of Planning, Urban Planning Department, 1970.
- 45. List, George F. and Mark A. Turnquist. "Estimating Multi-Class Truck Flow Matrices in Urban Areas." Paper presented at the Transportation Research Board Annual Meeting, January 1994.
- 46. List, George F. and Mark A. Turnquist. Goods Movement: Regional Analysis and Database. Region II, University Transportation Research Center, New York, 1993.
- 47. Little, Arthur D. "Feasibility of a National Heavy Vehicle Monitoring System, Revised Draft, Final Report." *National Cooperative Highway Research Program*, Transportation Research Board, 1988.
- 48. Massie, Dawn L., Kenneth L. Campbell, and Daniel F. Blower. "Large-Truck Travel Estimates from the National Truck Trip Information Survey." *Transportation Research Record*. Number 1407, Transportation Research Board, Washington, D.C., 1993, pp.42-49.
- 49. Massie, Dawn L., Kenneth L. Campbell, and Daniel F. Blower. "Comparison of Large-Truck Travel Estimates from Three Data Sources." *Transportation Research Record*. Number 1407, 1993, pp.50-57.

50. Memmott, Frederick W. *Truck and Taxi Survey* Manual. Upstate New York Transportation Studies, New York State Department of Public Works, Albany, New York, 1964.

t

- 51. Meyer, Isaiah and Emory Stoker. Truck Taxi Survey Manual: Field and Office Procedures. Bay Area Transportation Study Commission, Berkeley, California, 1965.
- 52. Mingo, R.D. Evaluation of FHWA's Vehicle Miles of Travel Estimates for Heavy Vehicles. Intermodal Policy Division, Association of American Railroads, 1991.
- 53. Oak Ridge National Laboratory. 1990 Nationwide Truck Activity and Commodity Survey Summary Report. Prepared for the Federal Highway Administration, U.S. Department of Transportation, by Statistics and Data Analysis Group, Center for Transportation Analysis, Energy Division, Oak Ridge National Laboratory, Oak Ridge, Tennessee, 1993.
- 54. Ontario Ministry of Transportation. Ontario Commercial Vehicle Survey, 1988 Station Summary Report. Ontario Ministry of Transportation, Transportation Systems Planning Branch, Ontario, Canada, December 1991.
- 55. Ontario Ministry of Transportation. Windsor-Sarnia Commercial Vehicle Roadside Survey. Ontario Ministry of Transportation, Freight Transportation Policy Branch, Policy and Planninf Division, October 1993.
- 56. Ontario Ministry of Transportation. Freight Distribution Series, Volume 7, Queen Elizabeth Way (Q.E.W.) Commercial Vehicle Roadside Survey. Ontario Ministry of Transportation, Freight Transportation Policy Branch, Transportation Costing & Productivity Office, Ontario, Canada, August 1993.
- 57. Ruiter, Earl R. Development of an Urban Truck Travel Model for the Phoenix Metropolitan Area, Final Report. Prepared by Cambridge Systematics, Inc., for the Arizona Department of Transportation, Phoenix, Arizona, February 1992.
- 58. Ruiter, Earl R. "Phoenix Commercial Vehicle Survey and Travel Models." *Transportation Research Record*. Number 1364, Transportation Research Board, Washington, D.C., 1992, pp.144-151.
- 59. Southeast Michigan Council of Governments. "Recommended Truck Travel Survey Method." Memorandum to Survey Team Members from Bob Newhouser, Southeast Michigan Council of Governments, January 1993.

 $^{-1}$

- 60. Stenzel, Joyce M. "An Application of a Sub-Regional Transportation Study in the Planning Process." CATS Research News, Chicago Area Transportation Study, December 1986.
- 61. Stokes, Robert W. and T. Chira-Chavala. "Design and Implementation of Intercity Origin-Destination Surveys." *Transportation Research Record*. Number 1236, Transportation Research Board, Washington, D.C., 1989, pp.23-33.
- 62. Taste, Ron. Freight Flows: Truck Driver Interview Surveys. Paper presented at the Transportation Management Conference/Workshop, State University of New York (SUNY), Maritime College, Graduate Program and International Transportation Research Center, May 1994.
- 63. The Port Authority of New York and New Jersey. *Truck Commodity Survey: Overall Analysis and Summary.* The Port Authority of New York and New Jersey, Freight Research Section, 1987.
- 64. The Port Authority of New York and New Jersey. 1991 Interstate Truck Commodity Survey. Volume 2. The Port Authority of New York and New Jersey, 199?.
- 65. Transportation Economics & Management Systems, Inc. Survey Methodology for Intercity Trucking Activity in Ontario, Final Report. Prepared for the Ontario Ministry of Transportation. Transportation Economics & Management Systems, Inc. in association with the Gough & Gray Group, Inc., April 1993.
- 66. Transportation Research Board. *Highway Capacity Manual*. Special Report 209. Transportation Research Board, Washington, D.C., 1985.
- 67. Tri State Transportation Committee. Goods Movement External Truck Survey: Interviewing Procedure Manual. Tri State Transportation Committee, New York, 1964.
- 68. Vancouver City Engineering Department and the Greater Vancouver Regional District. *Truck Study*. Vancouver City Engineering Department and the Greater Vancouver Regional District, 1990.
- 69. Wieder, Anne S., Kyungwoo Kang, and Michael Yokel. "The Truck Commodity Survey in the New York-New Jersey Metropolitan Area." Goods Transporation in Urban Areas. Proceedings of the Fifth Conference sponsored by the Engineering Foundation, Santa Barbara, California, March 1988.

- 70. Wilbur Smith & Associates. Nashville Metropolitan Area Transportation Study: Vol. 1, Origin-Destination Survey and Major Route Plan. Wilbur Smith & Associates, New Haven, CT, 1961.
- 71. Wilbur Smith & Associates. *Transportation and Parking for Tomorrow's Cities*. Wilbur Smith & Associates, 1966.
- 72. Wilbur Smith & Associates. Motor Trucks in the Metropolis. Wilbur Smith & Associates, 1969.

APPENDIX I NTACS Survey Form (Poor copy due to poor original survey copy)

[

and the second second

9.4

THE 1990 NTACS-1 QUESTIONNAIRE

I

This section contains a copy of the 1990 NTACS-1 questionnaire. Not all NTACS reported data are included in the NTACS Public Use File mainly due to high item nonresponse rates.

Some other data items such as "current home base state" have been suppressed or aggregated for confidentiality reasons.

-	Old the 2125-0640 Approve Superver 12/21/21
time 1 All-1 vs provincest or consists a constraint of consists a constraint of constr	BOTTOL - You seen to the Conset Sorters II sendotionated to the (Fin 13 U E Contex 4 own be seen out or point Contex Sorters Contexting of the test to the Sorterget Sorters. The last two provide Sort series Context or provide the Internet Sorter Sorter Sorters.
KATIONWIDE TRUCK ACTIVITY AKD COMMODITY CURVEY	In summarisationers paralleling in Bals Super-L planas seter to this Constant Play Barrison (CPVE)
100 Eastering trees 1.00 Example day 100 Eastering day	
Duret Au or The CENSUS	
NE A PART OF AMERICA'S TRANSPORTATION PLANS POR THE 1999'S AMO ACTORNE	
• We assot your held in this heatsmunds Thuch Astrony and Commently Burvey The behaviour prevent on this quantizations will be used by the government and others planning for future transportation needs than 61. — New highways — Tomard bolines: — Tomard bolines: — Highway managements and repar	A de une produces continuer en anna pré alerre annang le Casa Trans processiones dell le ""OTES - Ann pres des 7 10HO - Conserve sont ""O Donant I) St P is General à an i D Lassant I) st P is General à an i D Lassant I) page I pré annant
Verer versen derenne fir bei Bulgebulen bedertanden besterne erten anderer wei anterere e e benen der versense ein eingenseste der anterere seiter ein der bei besterne.	b. When did your dispose of this values? Some sparse over
Indexes in the part above. • We suggest that you label this descent ward with you in the annual volume on the Semple Day, and semperes Semen C at you partern your day's accuracy. • Plastic convert this temperatures three dested plant your September Day.	e. Now did you dispose of this unbicks? "" Sais 2 tor part : Discust, :
Plates read before completing this form	d. Whe is the provent owner of this yebicit? - "" () Unknown
The parameters in our of a province of the control scapeta is the Registerious behaviored the Bandes direct, for the description parameters are an obtained base.3 controls. A good C. Samanue A and S named proved behaviored of the orbital? Substantiation and can. Social and C. Sagarange on cape A, social to be provided a state to the state of the the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the state of the stat	History (114 Bions (111 2) Code
I dray welling did not conserve on the sample day taxatled, use the publication day of the values did not beaution in their the sample day or the publication day populated, but over source darring the sample torid, and an individ an (1991) 245-1744. On not substance proving day present that is do no. The sample day the publication and the public and one to beaut other the to-	If this values is as tenger in your passacaion, places sign on page 15 and reason
rentate die an Det partanter der, eren Dereph & mer net saatt brytest to ver	the questioners and a detail of the second states

Bortlet A - Ye	Nede Information			Seatten 8 - Vel	ulaia Desuription	
14. De you contently aperate dels valdels?	·** + () ves			The lateners quarters around to the versite to use during the service do	y and the past 12 menths. If this right	nto del port operato en the 1011793~1744 pellent.
If the marcate present status	10mg			1. This trank was MORT FREQUENTLY operated as -	Barrada des	Part 12 manths
				Mark 171 andy and bee in each column	······	
	· · · · · · · · · · · · · · · · · · ·		-	A. Personal Personal State	•••()	••• 0
				h Comment can an	·•• (]	·•• C
b. Here many works during the past 12 mentes did	· • • •			a Comment of the		
And changes and angener.				4 Come bacress the	n i	••• C
28. Where is the ourself basis of the vehicle?	······································			New yeards you bast describe this vehicle as it was MORT FRIOURN TLY operand during each period? Most (2) and are been match (churn)	· · · · · · · · · · · · · · · · ·	
pertad at statemed	The Comment	11 State 1 23	Case			
	-	1	1	6. Straught truck with 4 trad withdut trader		
	111	·····		b. Stragte purs with 6 tred p.A.r. Vederisi		
during the part 12 mande?				. Straget such with a series tree without safer		
				d. Straght truck with 0 or more truck primiting the starting	·•••	
a, in here many states did this validate extents	· •		1	. Truck there to and unit putting transfel	144 ()	····
during the part 12 metho?	Stores			f. Truck tracter without their		
4. List the Seven Prome with the bighted millings during the part 12 mentils -	193			g. One - Specify in the country	······	· • • •
	(2)			3. Indeste Die bindig of Deficited public during sant perfect.	Alark IX's all that appro	Mark (X) any ana
	120			a, the wester proved		
a. Did this vehicle operate in Canada daring the	Gres - Muse Iti me prevenene an	-		b. Uniter and other traders team than 20 feet used with therefit track		··· : *
paul 12 marma?		_		(110as anto en traiter		
1		J Mannaka		(2) Ture anna a' traige		
1	This terms forme the	Adamte	•	(2) Those or more arrise on trains		,
}	1 30 Them Brokenes 1 49	Ö 9 Cause	•	a, One his tracer used with straight truth		
·	110 0.000 100	🖸 Yudan Tantur	•	(1)Tark anno an Weiter		
,	, ·)0	and the second second		12) Three pass on Water	1-1-J	,,, L
•	. , C =0			(3)Paur ar mare terret an mater		
	· · · ·			d. One sume traine	I	~
t. Did this vildels v.arets in Medice dating the	-•			(1)One ante en tradut		
	· 0+15 · 0+0			1217una antica de tratas	0	m U
	•			(B) Theat of story price on trader		334 LJ
a strang an and an and a state of the second state of the				a. Two traters, and same and one hull		-
part 12 manths was abastand from -				(1)These deale an true traders -	mü	
	1 tan			1215me arres on two traders		
a. Prives funt damp?	`			(Stiffing and an two waters	<i>"</i> "	
				14) fan ar mare antes en troe tradere	794 (]	394 LJ
b. Bas system types, stall,				P. Thrus surface, and some and two hill		D
	······································		L			mū
4. RAMPLE WEEK	Bunday (Marday (Tusatas Mudanasa)			(IT) THE AND AN PARTY PARTY I		1.0 5
For each day of the paragets areas byteres in the			-	1212ords Prine to gate and a		344 C
Representation information on page 11 chack "Yes" of the	Tres Crest Over in res !	Over Over	1.Ores	(4) Baget er mark belas en mine metare		
Courses outside traveling empty)	10mg 10mg 10mg 10mg 1	Dio jDio	-0-0	Other - Passe describe in getter the minister of Sparry and annue an those traders Spacry in the common		
		l	L			

· · ·

4

• • •

· · · · · ·

Γ	Bootlen 8 - Vehicle C	ecoristion - Continued] Seation C	-Vehicle Use	
4	. Induces the londy type that mean clearly reconcilies this us unit is 6 much means indicate the body type of the trailert	Main daring the sample portail, 1 I priorited,	t the parrier	The following questions relate to the vehicle & use DURING excepte day, use the exhibition day. If the vehicle did not	THE SAMPLE DAY the or	Alale did not operate on the sail (301) 763-1744 askest.
	Mark (2) andy and been in seath testumer.	Bumpta day	Past 12 mestes		• • • •	
	A. PLATFORM TYPES, metales foreas, mater, and foreast with solid devices, and less being spectratists		m 0	1. Bann dans und three Aparts and	Manin Day	760
	a, PICRUP		n>D	12. Whith was the advances reading of the vehicle at 12:01 a.m. an the sample day?	a Adda.a.d.	
	C. PAINA OR COMPACT VAR		m 0	The Manual manual radius and the weblack through during the	1 49.8	
	D. MINIS VAR, VTRITT, STATION WAGON ID-MAD.	m D		free elemente		temanan are accopitable
	E. VAR TYPES subsists statesed verse apen too verse, drap transports, refrigeressed, neurofrageressed, and evaluations and high subset.	H= D		b. What persent of these miles were on the MTERMTATE HIGHWAY STOTEM?	····	
	F. SPECIALIZED USE TRUCES			8, What propert of theme attes were an reade which had beer ar mean lance but were NOT on the		
	1. Automotive of boat transport	H+	1	MITINGTATE HIGHWAY SYSTEM?	· · · · · ·	
	2. Beraraga mint	3HQ -	m0	d. What parcent of these adies were affreed		
	3. Numeropass any company on varies channes		3mD			
	4. Received to be a content on trade phases		110	day?	98 ee	and of stations are summaried
	B. Other cargo servicines sheres	279 🖸	9 10	b. New much was paid in Myhmey sells during the sample day?		
	S. Compress manage		m -O	Re. What may the size of the vehicle	An 2 with the second second	
	7. Durap truth	14	D 🕫	•	an Die andere der?	wen af te manten weiginf
	E. Brun takar brakang haspara, gran barasi	344 C	m 0		111	
	S. Burbage mush	741	H1	Longon (At) Afront Inserports and of last traders	·	.n
	18. Liverteek truth, meluding buurteek mes heme				•	, , ,
	11. Pula. Ingging, at most truth	H 10	0			- <u>ħ</u>]ħ
	12, Tanà anna far dre bulk	** 0	m 0	Tare weight lengty!		**
	13. Tern weak for brands to gasse translatedies frateralis	m 0	m 0		··· ·	
	14. Turis truck für binnels at galant thesereture theteretik - menung nyop elsem persons on some	m O	- ==0	Loosed variable weight tweight of buck and carges		Re
	4. MC 387	m 0	** 0	b. How would say had deadly the woldship had	····	
	S. MC-301	D \$**	щ П _	during the sample day? If the vehicle was among the antipy day, shart the bas that typiquity applied	a C Sanard aburna	an an a truck ar traight Bass -than- durg shapharas surveildated by pthere
	e. MC-212	214 🖸	. D•••	8. Was the vehicle need to have beserdows metarlais in	····	
	4. MC 337	310 🛛	жQ	due to the Code of Federal Regulations, this 68,		
	a. MC 306	M1 ()	*** D	Transportations, dering the compte day?		
	18. Usiny music	an C	** 0	wars on heard the relates as it beit the starting place		
-	(bugs — if name of the shows attacky investment module the barts type of the vehicle of the theta vehicle attacked to it, mark the "Dirar" bar and describe the vehicle.			en the complex day? b. How commy of the " employment draws the vehicle periodice during the transfer day?	[mit,	
	8. bits - Security a Dis actions			5. Mark of heres this back of the terms during which you opened the value darks for completed any.	40 - 40 - 40 40 - 40 - 40 - 40	0 s.m. at 210 41 s.m 440 p.m. 0 s.m. at 2 441 p.m 440 p.m. 0 s.m. at 2 441 p.m 540 p.m. 0 s.m. at 2 541 p.m 1200 s.m
-	nameta in tanàn 🛛 🗛					

the state of the s

- ,

•

1

łĮ			Section C — Vehicle Use — Car	retrouved		
i	COMM		APERANCE LIST FOR USE IN ITE ME Biel. IP	R. 1 Stol. A	NO 1440	
i	PWIA - HAZARDOUS MATERIALS		Parts - PROBUCTS, SOUPSMENT,		Manada and Menal Products	
"	Fishermatic Bands	41	Agreedents and Fred Presidents		angest d'anne. ent	24
	Combusidas baunda	41	Long grands — gards, harbet gandry hage, bat, and phar mouth products of		Extracted many products and bons and many many of property of property and the second	i
1	Correspond Baunda	- 44	Frank farm products - gran arans.	1	Statistical line States	47
	Prost & subda	-4	toppens dic.	••		24
1	Person & Square	•	Propessed lands - cannot grade	1	estudiat and parts	
	Partitudes gande	-	Served mere dary products takened		Other Manufactured Products	
	Continues	47	products pri	. 1 ''	furniture immed and represents and references -	
	frammatic gas	***	Surg Preduces southed - buds of the ful away area	1.10	Testites and speaks - there testim great	
	Hanflammakis ges		Budging materials - gravel band barranes gravel, and plane, and	1.	Mansherman probars of mendecturing -	
	Person A	••	Faragery, Wash and Paper Francess		Annally and lays	- 54
	Correges and	••	Logs and lands problem - Course out		Measure of the actual and other there as	1
2	Espherica, A or B	61	Autor bates of parts and parts a		Butter gentent - fein bans, officer, out	1
:	Barring against	••	Lumber and Interested avoid products - Except Arminet	n	Mant sarps, garant fragel, paranter geneta, and	
			Paper, pressed methor and paper product	• •	Tankagents for spectrational use as in each maps of	
			Charman Province, and Adad Products	1	fores, garbage, mast	
1	Hereiter and the set for all the set		Countrasts and/or drugs - Mainting	1	trausinal many	1 87
			aw.	1.11	Other - Please describe in detail	
	Secondly of		Annalaum, parataum probats, parata, and papeal or tar someth	-		
		1	Paumas andre nabes predame		P NO LOAD CANNED - Versio anuty	-
		~	Type of Plant Codes star to these sector for comparing some Sec. 13	bi and 14	<u></u>	
	A Redress tadlig gar-sallig tarting	1	tantari battar gattar a d - Press dartadat	infan deren	0 - Constantion des	
	8 - Andread Sandar, Santar, o' and all and all and the second		E - Page and and	lading .	7 - Spylinging India, a annotic para	
	4 - Angeri Ingeneng Sealber		an, harber, gr anterent & - frangt, tant, find	farm faabbreg	B - (May balling, mineri, handris, alter sale	
	8 - Japan Ipage a delare a a diperiot	6 - Pe	riteration (1) - Engla dan state, ill		0 - Ppt a char temperat tails	
	E - Hate to part landler territer fundles	1 - 1a	a tem 🔅 Mai, galla, g		1-06s	

				Section	C - V.	edals Una -	Continued						
0. đơi	er Juli	ar Des fuite	ating data A	- CACH	TOP &	ring the parts	- CAMPLE DAY on	Y. Conto		s her houd	L Band or		
SANA	E DAY B	TOPS			-	the stup!			-				
d'annahrs a the			.				er agenty	بصب					
	any piec	N	-		1				- Ill	Wants.		all .	Constant of
	w			-			1 4 1				100		
Constant and the second second	0000 00		' ::	1	1		• (] ****	ł	{			{	
Car	County		June -	1			a Li Bray all salar a Li dhan a ta tarrat.						
				1		10	Second 2	·					
Candred enhance ort at dagar				1	<u> </u>								
Paral St Arnest tons	10			1	-		. () 000mg -	T			1		
·	المتن						a D Brug all some	 	<u> </u>				
C.41	Carrie			1			a Li Ottor - James y						
Landed vehicle art di depa		The of wate	to establish	1	1						1		
L			_		1				+		+		
Annual State	•••	Names and Am	• ::			C C Annum III	• C Bahany	1		1 ·			
Can	1			1	1	Dee .	+L) the - terms						
	<u> </u>	This of head		4	1			·	+	┨────	·	·	<u> </u>
						- College		-		1	<u> </u>	<u> </u>	ļ
Till Call Arment Street	an P				·	Linne	.130					T	
	<u>ا ا م</u>		7 21111	4	— ••				+—	+	· }	┥───	
{***	1				1		\$ C1 GRM - Showy	パ	1	1	.	l	
	<u> </u>	Per al val	ne propriet	1		10.00	-	-		1		1	
		Constraint of the			1	110000		+	+	1	╉───	+	<u>† </u>
	:=1]	-		· C Bray of same				I	1	
Cay	Frank		Liene	í	1		a [] Dear - Josefy	,		1			ł
Landard Tabletie art af Bare				·1	1	-			1	· • • • • • • • • • • • • • • • • • • •	1	1	1
				<u> </u>	1	-	······································			1	4		<u> </u>
Anternet Arrival Series		Cupaners de	• •	•		- Dhan w	• [] bet-ey	1	1	1	1	1	1
Chy	7	7	15.000	4			· · · · · · · · · · · · · · · · · · ·			1	-1		1
	1		1	4		in the second		7	-				+
Landed uplica of a dee		No of test	are granted		1		······	-1	1	1		1	1
1		l l			1.00000	· · · · · · · · · · · · · · · · · · ·					-		

and the second second

100

!

ACOTTONAL BANKLE GAY STOPS			Giuni wan dar panyoos of San onep? Mart (21) all soat appry						
	∤ •	641	141	-	1.1992	Per J		i ann	Western .
Ale Desis Tore of prove and the provided within out of desirery The of prove and the provided and the of provided and the of provided and the provided and te			Denne + Denny tean + Denny trans Den e - Denny trans Note Denny trans Note						
ACTES Dete form of privation of Departure some of the form of privation of the second source			- C Augun () - D Daber Mart () - D Daber - D Na up (- D Daber - Search - D Na up (- D Daber - Search 						
Any Control of an of a control			Denner - D Bahary - Constant - D Bahary - Constant - D Bahary - Constant - D Bahary - Ganady - T Bahary - D Pan an - D Bahary - Ganady - T Bahary - Constant - T Bahary - Constant - T Bahary - Constant - T Bahary - T					· _ · ·	
The discriment of the of the of the second o			D Anamia . D Balany Bar . D Dana at train D Anama . D Bar . South 7 D Anama . D Bar . South 7 D Anama	 					- <u>-</u>
			· [] harris · .) harri harris ·						
The provided of the second of						t			

İ	ADDITIONAL SAMPLE BAY STOPS			the step? Much SIC of the series	erel eretgete for fine fanne par E artis Stret fanterentien af Ganne entweret	ter a anter befannt a gab ang Yel
		•••			anna brend Brend	
	Case of ania a Departure time a m			- Dinne + Dinny		
I	Chry Causey Lines	1 1		· D		
I	Lander and the of a sectors the of salars prested	1 [
	Call Care I and a series and Conserver and			· Dhans		
ł	C			· Canal - a California and senser • California - a California - Spenser - Sp	┠━━━┼━━╍┼━━╼	╾┠╼╾╂╼╍╉╼╾╾┨
ł	Lasted renats un at deserves les of times anadred				┠──┼──┼──	
	Sea of a Course I former of annual Courses on annual				┟━╾┼╼╍┼╼╍╸	
ļ	;= ;=			- C) Basen ar o C) Batrary 		_
ſ				and all the second		
•	Landed vehicle on at departure					
	Carrent Dene Tone of annual and Deserve and an			· [] Person a L] Battery		
	CRy County Fund	1		· 13 Pes es . La Orier - Speerty y	<u>}──</u> {──{	
ł	Contra rotati en al depeñon - 140 el cateri encatad -	1				- -
	Contracting Dates Toma of prints and and and and			ilanan ellanar	╂──┼──┼──	╾╂╍╍╌┠╼╍╶┨
	Care (Cares) - jStore			bass ell bras et same		
	Lander where of a superior - Ity of party marries -			Line .	│ │ ┤ -	·]] -]
	Street I Count I former of annual	ļ		9499 · · · · · · · · · · · · · · · · · ·	┠───┠───┠───	
				- Channa - () baren	Ⅰ↓	
				- Chara + C Ore - Jesty		
	A of balan entering		t			

and a service of the set of t

. .

•

.....

		1		Sfins was the person of the energy block (2) of the energy	Anna Da atan	age of particular Reprint problem wellight phone		pillar the o I al cash at
ADOPTIONAL SAMPLE DAY	\$1075		-		Constant Provent	9796 9794974	11	
• @+		r		-41	1.000 Canta			+
Arata Davis Tana ar I anna an Du Anna Arata Arata Arata Yang ang ang arata Arata Arata Tana ang arata Arata Arata	Siere All	-		· () Revenue · () Determine Determine · () Determine · () Posture · () Determine · () Posture · () Determine · () Posture			 	
22312 Dury form of server 1 0 	Sectors and Sectors Sectors of bodies, priothed			· D Annen u · (1) Danen San · D Day of Sam · D An u · (1) Day of Sam · D An u · (1) Day Same · D An u	-			
An Long I and a series of a se	Bioto of protons stratters		-	Dennis a . Dennis bea . Dennis autor Den . Den . Dennis Den . Den .				
1 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0 0	Surran and a			Channe - L) Danen terr - Li Danen til Drug af varie ci i Drug Sauch ci i Drug Sauch ci i Drug Sauch	;			
Backets Control [Forme of served] 6 <	Siener af a salary principles	:		U Anne u	-			
And which at a departure pro-	5000							
	-	n	hand yo	u for your occurrentical				

a and a set of the set

The second se	, free be streen to be street to be an and the street to be street by the street pairs of the street to be st	, Wild hype of plans was bild? Court the type of plans latter and have turn 10 a house many makes from have may us the to	bline an bi ba pang a sheri da	ייינער איז	Tana una dia sekarany andro di ta sek 1969 pada se kan partang t Ulang una dia lan partang se dalama pada: Na pangka dia 2	Have many stores wants made for band, band, a			Transformer and the part of payors, hother	Name of part human lawy warding		Autorial Sectory Sectoring Sectory	Beeden C
	1 11								:	÷			
antina fuer and by Kinger fuer						1 mg		Party Lang Sand Lang are		Construction and			- Cardinad
							1	1	<u></u>	•	i	4	

- 19 and full a line have a strangeness of the stand of the address and the strangeness of the strangeness o

APPENDIX II NTTIS Survey Form

		POWE	NTTIS C ER UNIT	COMPAN DESCRI	Y & PTION		
:01	MPANY DES	CRIPTIO	N				
PEI	RATING AUTHOR	ITY:					
s th	his a daily rental	truck? YES		SKIP to Power	Unit Descrip	tion below	•
s tr cíty	y/county/state/fee	ieral)	\$			•	
	any of your t	rucks eve	er carry goo	ds interstate	(across stat	e lines)?	
0.	any or your (PRIVATE	{]1		• •		
11	I YES - Are you	FOR HIRE	1 12 / 100 1	wthorized []?)		•
	1	(Carry other people's goo	(com s(s) (zom	con/contract)	also the dri	ver7 H0 []	2
	,	PRIVATE	[]]		•		
1		(Carry own g	poods)				.,
1.	* 1 ne jou	FOR HIRE (Carry other	1 12 	(₁)5	also the dri	ver? NO 1	2
	l	people's goo	xds)				
1	9 UNKNOWN {	PRIVATE POR HIRE	() <i>1</i>			TES (32
•		FOR HINE	` 10´		also the dri	lver7 NO [12]2
	7						
o	WER UNIT D	ESCRIPT					
	Verify (the make,	model year,	and VIN, and	lask for the	model name	•
	and com	any ente	11000001				
•	Make	•		Year: 19	VIN	·	
•	Model Name			Company U	nit Number		
•	Model Name EDITOR: Code	e the bas	e state of c	Company Un	nit Number		
•	Model Name	e the bas	e state of c	Company U Operation	14 CAR STYLE		
•	Model Name EDITOR: Code POWER UNIT	e the bas TYPE []8	e state of c	Company U operation 11 13 6.	hit Number 14 CAB STYLE Cab Forward		1
	Model Name EDITOR: Code POWER UNIT 1 Tractor Straight Tr	e the bas TYPE []8 uck []]	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over	[]	1 2
•	Model Name EDITOR: Code POWER UNIT ' Tractor Straight Tri STRAIGHT TR	e the bas TYPE []8 uck []1 ucr	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over Short Conver	[] [] htional []	1 2 3
	Model Name EDITOR: Code POWER UNIT ' Tractor Straight Tri STRAIGHT TR BODY STYLE:	e the bas TYPE []8 uck []1 UCK	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convert Med. Convent	[] [] htional [] iional []	1 2 3 4
•	Model Name EDITOR: Code POWER UNIT Tractor Straight Tri STRAIGHT TRE BODY STYLE: Van	e the bass TYPE []8 uck []1 UCK []2	e state of c	Company Unoperation	hit Number AB STYLE CAB STYLE Cab Forward Cab Over Short Convent Med. Convent Long Convent	[] [] htional [] fional []	1 2 3 4 5
•	Model Name EDITOR: Code POWER UNIT Tractor Straight Tri STRAIGHT TR BODY STYLE: Van Flatbed	e the bas rype []8 uck []1 vck []2 []2	e state of c	Company Unoperation	hit Number AB STYLE Cab Forward Cab Over Short Convert Med. Convent Long Convent	[] [] htional [] tional [] tional [] 10	1 2 3 4 5
!• !•	Model Name EDITOR: Code POWER UNIT Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker	e the bass rype []8 uck []1 uck []1 uck []2 []2 []3	e state of c	Company Unoperation	hit Number AB STYLE CAB STYLE Cab Forward Cab Over Short Convent Med. Convent Long Convent FUEL	[] [] htional [] donal [] donal [] He	1 2 3 4 5
!• •	Model Name EDITOR: Code POWER UNIT Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig.	e the bas rype []8 uck []1 vck []2 []2 []3 []5	e state of c	Company Unoperation	hit Number IA CAB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Cas	[] [] itional [] ional [] ional [] ional []	1 2 3 4 5
!. !. !.	Model Name EDITOR: Code POWER UNIT ' Tractor Straight Tri STRAIGHT TR BODY STYLE: Van Flatbed Tanker Refrig. Dump	e the bass rype []8 uck []1 vock []2 []2 []3 []5 []6	e state of c	Company Unoperation	hit Number AB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel	[] [] itional [] ional [] ional [] [] []	1 2 3 4 5
•	Model Name EDITOR: Code POWER UNIT ' Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other	e the bass rype []8 uck []1 vock []2 []3 []5 []6 []7	e state of c	Company Unoperation	hit Number AB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other(Spectrum)	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3
•	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tra STRAIGHT TR BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other	e the bass rype []8 uck []1 IS UCK []2 []3 []5 []6 []7 []8 16	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convert Med. Convent Long Convent FUEL Gas Diesel Other (Speced)	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3
•	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify)	e the bass rype []8 uck []1 vck []2 []3 []5 []6 []7 []8 	e state of c	Company Unoperation	hit Number AB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other Fower Unit 1	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 1 2 3 4 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7 7
•	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tra STRAIGHT TR BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A	e the bass rype []8 uck []1 []2 []3 []5 []6 []7 []8 	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convent Med. Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 27 27	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3 4 5
•	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two	e the bass TYPE []8 uck []1 UCK []2 []2 []3 []5 []6 []7 []8 VLES []2	e state of c	Company Unoperation	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convert Med. Convent Long Convent FUEL Gas Diesel Other Fower Unit 1 20 21 22 Diesel	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3 3 HT:
	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify) NUMBER OF A Two Three Four +	e the bass TYPE []8 uck []1 UCK []1 []2 []3 []5 []6 []6 []6 []6 []7 []8 VLES []2 []3 []3 []4	e state of c	Company Un operation 6. 7. 8. 9.	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 22 Power Unit 1	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 5 1 1 2 2 3 4 5 5 1 1 2 2 3 4 5 5 1 1 2 3 4 5 5 1 1 2 3 4 5 5 1 1 1 2 3 4 5 5 5 1 1 1 1 2 1 3 4 5 5 5 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1 1
5.	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify) NUMBER OF A Two Three Four +	e the bass rype []8 uck []1 UCK []2 []3 []5 []6 []7 []8 UCK []2 []2 []3 []4 UCK	e state of c	Company Unoperation 13 6. 7. 9.	hit Number 14 CAB STYLE Cab Forward Cab Forward Cab Over Short Convent Med. Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 22 Power Unit 1 26 27 28	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3 4 5 4 5 4 5 4 5 4 5 4 5 4 5 4 5 4 5 4
5.	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two Three Four + Estimated A	e the bass rype []8 uck []1 []2 []3 []5 []6 []7 []8 	e state of c	Company Unoperation 6. 7. 8. 9. <i>is power</i> unit	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 27 Power Unit 1 24 27 28 29 30 31	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3 4 5 1 2 3 4 7 3
· · · · · · · · · · · · · · · · · · ·	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two Three Four + Estimated A Percent of	e the bass TYPE []8 uck []1 UCK []2 []2 []3 []5 []6 []7 []7 []7 []8 VLES []2 []3 []4 nnual Mil annual Mil	e state of c leage for th ileage for e	Company Unoperation Company U	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convert Med. Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 22 Power Unit 1 24 27 23 5 29 30 31 for this power	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 1 2 3 3 HT: 3
· · ·	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two Three Four + Estimated A Percent of • Local (Pi	e the bass TYPE []8 uck []1 UCK []1 []2 []3 []5 []6 []6 []6 []6 []7 []8 VLES []2 []3 []4 nnual Mil annual mi ckup and	e state of c leage for th ileage for e delivery, w	Company Unoperation Company U	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convert Med. Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 22 Power Unit 1 24 27 28 for this power adius)	[] [] [] [] [] [] [] [] [] []	1 2 3 4 5 5 11 12 13 14 5 14
5. 5.	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two Three Four + Estimated A Percent of Local (Pi Short Hau	e the bass ryPE [] 8 uck [] 1 UCK [] 2 [] 2 [] 3 [] 5 [] 6 [] 7 [] 8 	leage for th ileage for e delivery, w city, one-wa	Company Un operation 3 6. 7. 7. 8. 9. is power unit ach trip type ith 50 mile r y, distance 5	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 22 Power Unit 1 20 21 22 Power Unit 1 26 27 23 5 27 30 31 for this power Gadius) 50-200 miles)	[] [] [] [] [] [] [] [] [] []	$\frac{1}{2}$ $\frac{3}{4}$ $\frac{1}{2}$ $\frac{1}{3}$
· ·	Model Name EDITOR: Code POWER UNIT & Tractor Straight Tri STRAIGHT TRI BODY STYLE: Van Flatbed Tanker Refrig. Dump Refuse Other (Specify NUMBER OF A Two Three Four + Estimated A Percent of • Local (Pi • Short Hau • Long Haul	e the bass TYPE []8 uck []1 UCK []2 []3 []5 []6 []7 []8 	leage for th ileage for e delivery, w city, one-wa city, one-wa	Company Un operation 6. 7. 8. 9. is power unit ach trip type ith 50 mile r y, distance 5 y, distance 2	hit Number 14 CAB STYLE Cab Forward Cab Over Short Convent Long Convent FUEL Gas Diesel Other Power Unit 1 20 21 27 Power Unit 1 24 27 28 : 29 30 31 for this power solutes) :00+ miles)	$\begin{bmatrix} \\ \\ \\ \\ \\ \\ \end{bmatrix}$ ional [] ional [] ional [] ional [] $\begin{bmatrix} \\ \\ \\ \\ \\ \\ \end{bmatrix}$ $\begin{bmatrix} \\ \\ \\ \\ \end{bmatrix}$ EMPTY WEIGI $\boxed{33 24 2}$ LENGTH: 32 33 34 35 34 35 34 35 34 35 34 35 35	$ \frac{1}{2} \\ \frac{3}{4} \\ \frac{1}{5} \\ \frac{1}{2} \\ \frac{1}{3} \\ \frac{1}{4} \\ \frac{37}{4} \\ \frac{1}{40} \\ \frac{1}{5} \\ (Total = 1) $

1.1 1.1
		NT	ris sui	RVEY D	AT TRIP	1		•
1.	. OPERAT Were y	ING AUTHORIT	Y (Private for-hire (Carriers on le.g., on ba	ly) ckhaul)?			
	[]2 10	NO Yes Wasit	as? ICC (Exemp Intra	(common/cont ot (intersta state for-h	ract) te hauling (ire .	[] only) [] []]	2 3 5	
2	. DRIVEF	AGE:	(rs.]. I	ORIVER YEARS	WITH COMPAN	NY: Yr:	š.	
4	. CONFIG	SURATION: A	ny trailers:	? No []2 Yes []2				•
1 2	Type:		Power Unit	Ist Trai Semi	ler 2nd T	<u>railer</u> <u>3</u>	rd Trailer	
-				Utility Other None]]] Utili]]] Other]]5 None	ty []] U []4 O []5 N	tility []3 ther []4 one []5	
	Body:			Van Flatbed Tank Auto C. Dump Other	[]] Van []2 Flatb []3 Tank []4 Auto []6 Dump []8 Other	[]] V ed []2 F []3 T C. []4 A []6 D	Iv an []1 latbed []2 ank []3 uto C. []4 ump []6 tther []8	
۲ د			¥	(Specif	$\frac{1}{2}$	(n ¹) (cify)	(Specify)	
	No. A	xleg Used: _	23			25 -	24	
>	Lengt Empty	Wts (Lbs):	27-29	30-31	<u></u>	<u>। य</u> ्य	36-38	
	5. CARGO	·: _	[]	_[]	[]]	()	
	Cargo	Wt (Lbs): _	65-70 No. 5 - 1 -	71-76				
	Hazar	dous Cargo	No []]	No []	2 No	[]]2	No [12	
	6. GROSS	COMBINATION	N WEIGHT for	r the trip (Lbs):	43-41		[1] 99 00F1-
	1. Start	ting Point				Time:		
	2. End I	Point	(City)					
		<u></u>	(City)		(State	·) ·	AM[] PM[]	
ן ש	J. VIA -	(Describe ro	ute/give ro	ad nos., et)	•		
AG	- A Tota		Trip:		•			
	5. Breal	kdown of Mil	eage:	10-13				
2		LIMITED	ACCESS	US/STATE/MA	JOR ARTERY	OTI	IER	
		Day (6am-9pm)	Night (9pm-6am)	Day (6am-9pm)	Night (9pm-6am)	Day (6am-9pm)	Night (9pm-6am)	
	al: .	14-19	<u>10.</u> 41	22-25	26+74	30-3	3 34-37	
Rur	Urban:							
Rui	1. a		1	46-49	50-53	54-5	58-61	
Rui Sm (Fiz	urban:	38-41	47-45			1		1

.....

r___

17

•••••••••••

APPENDIX III 1986 CATS Commercial Vehicle Survey Form

٠

11.1

t

:

1

s /



and the state of the second

a state of the second second second

CHICAGO AREA TRANSPORTATION STUDY ILLINOIS DEPARTMENT OF TRANSPORTATION

300 WEST ADAMS STREET CHICAGO, ILLINOIS 50608 TELEPHONE: 793-3464

1986 COMMERCIAL VEHICLE SURVEY **Vehicle Trip Record**

THIS TRIP RECORD WILL ACCOMPANY THE ABOVE VEHICLE FOR A PERIOD OF 74 HOURS BEGINNING AT 1201 MI ON TUESDAY, FEBRUARY 25, 1986.

CONFIDENTIAL. The information abtained from this study will be afforded complete confidentiality, and well be used for statistical/analytical purposes only.

A PERSONAL MESSAGE TO THE DRIVER OR DRIVERS

YOUR CHANCE TO GIVE US YOUR TWO CENTS WORTH

ugh the Trip Record before starting out. We need the bad use don' I want at he be difficult for you



The Chicage Area Transportation Study (CATS) of a government againcy that does transportation pranoin for Revitions) Timines . It does studies of inansportation and inaret activity, including tracks and busies CATS are mentant computer programs to project failure traffic conditions

CATS produces two name and dense, which are (1) Long Range Transportation Plans 15-year and 20 year propertions) and (2) the Transportation Improvement Program, which is synchronid specific scheduled improvements such as read surfacing and indemney, signating improvements, grade crassings and Sugar at some , and

About This Study

This Trip Record will give CATS a "Day in the Ldy" look at the region's buck first - what it does not ere if yors. Your brin, in littler and the fectore, and the information we get from if will be unobjaraway Please houp a record of the workday of Tuesday, February 25, 1986.

NOTE . If the mouther on that day is so had that the vehicle will be unable to ramplete a full and rerna lay, then planse here the Trup Arcard on another day later in the same work

but The Besuits.

All information collected team the Survey will be basied in the structure confidence. Note that we are noty information in the TRUCK. We will not use the control to CACY planning users. The information will have an applicit legitime that for topectical forgometer (Programs

The Commercial Valuele Servey has been undersed by the Brimors Free Long Association, and by the Oncase Association of Compariso and Industry

is expectation of getting your help, I thank you.

YOU COUNT TO US. CAN WE COUNT YOU (MIP

Yours appreciatively

F Grand Ray

Chull Freight Pla

We want to give you as idea of what we consider to be a trip, or trips,"for the incurd We will use an example of a school by:

- The i but peers has grange to starting point of scherhold inde-ter but peers to grade scheal at east of route the peers of peers scheal at east of route the peers because the peers the peers the peers because the peers


LICENSE PLATE NUMBER (This will confirm that the Trip Recard is on the vehicle for which it is intended)

....

i trans

12.1

. . .

If the vehicle's sornal work is being conducted by another (that is, different) volucie, then please send the book out with the substitute volucie

LICENSE PLATE OF OTHER VEHICLE	
--------------------------------	--

IF THE LICENSE PLATE HAS RECENTLY BEEN CHANGED, ENTER NEW PLATE NUMBER

HERE: __

STARTING ADDRESS | 12:01 AM, FEBRUARY 25, 1966

AT MONIGHT, THIS VEHICLE WAS (Chack which applies):

CI PARKED ON STREET

[] PARKED ON COMPANY PROPERTY OR IN COMPANY GARAGE

EI PARKED ON PERSONAL PROPERTY OR IN PERSONAL GARAGE

L) IN TRAISIT (That is, at work)

LI OUT OF STATE

IF THE VEHICLE WILL NOT WORK ON FEBRUARY 25, 1986 BECAUSE THERE IS NO WORK

٠.

FOR IT, CHECK THIS BOX []

IF THE VEHICLE WILL NOT WORK ON FEBRUARY 25, 1966 BECAUSE IT IS NOT

OPERATIONAL (For sale; In for service etc.) Check this box [7]

NOTE: This may take you back to the top of the page to onler the plate number of a substitute vehicle.

CONTACT PERSON FOR CATS TO FOLLOW UP WITH:

NAME: _

TELEPHONE :

i proces

TRIP 6

Ο ō

000 α D

D D D

۵ ۵

۵

START TIME:

START ODOMETER READING:

	TRIP 1	TRIP 2	TRIP 3	TRIP 4	TRIP 5	[
P TIME						
» ODOMETER						ł
DN-STREET OFF-STREET	0 0			0		
IVITY AT STOP (Check one) up, Load att, Unicad	0	0	00	0	00	
Un Vehicle	D	ō	Ū	D	ō	[
et other personal	٥	C		۵		
D USE AT STOP (Check and)				_		
dential	0					[
il Analusian	u n					}
nacioning isportation/Utilities/ munications/Warehousing, uting Wholesale	0	ο				
lic & Government, Inct. Jums, Scheets, Libraries	D	σ	D	D		
ce & Services, incl. Finance Estate, indecence, pitats	۵	0	D	۵	D	
n (Arite In)						
CX TYPE (Check one) ight Truck :tor + Semi Vailor :tor + Trailor tail	0000			0000		
nunc avail Warthousing, Iding Wholesale Iding Wholesale Inc & Gowarnson, Incl. Imas, Schoots, Libraries ce & Sarvices, Incl. Finance I Estate, Inserance, pitals r (Write In) CK TYPE (Check one) ight Truck the + Seul bailer the i Trailer teil						

		T R I P	RAME AND ADDRESS OF STOP (Please Include Z19 Code)
TRIP 7	TRIP E		
		1.	
			المیں کے انسان بیان کے اس کا انہ وہ کا میں ان کا اور اور اور اور اور اور اور اور اور او
n			
		2	
		·	
	ū	3	
		4	
	ö		
σ	α	5	
	0		
U I	U U	، ۹	
		1	<u> </u>
α	a		
0			
		•	

.

.

¢

1

}

1

•

١

1

١

1 1

.

1

1

I

١

1

START TIME". START COOMETER READING:			T R NAME AND ADDRESS OF STOP 1 (Pluase Include Zip Code)	THANK YOU FOR TAKING PART IN THIS IMPORTANT STUDY OF COMMERCIAL VEHICLE ACTIVITY.
······	TRIP 9	TRIP 10	9	IN THE SPACE PROVIDED, WHICH FOLLOWS, PLEASE TAKE THE
STOP TIME STOP ODOMETER				
STOP ON-STREET OFF-STREET			10	
ACTIVITY AT STOP (Check one) Prot up, Lead Drop eff, Unland Land and Unland Gas Up Vehicle Weal or ather personal	0000		IF THE VENICLE MARES MORE THAM IO TRIPS FOR THE DAY, WRITE W THE TOTAL NUMBER OF TRIPS HERE:	
LAND USE AT STOP (Chuck one)				
Residential Retail Monufacturing				
Transportation/Utilities/ Communications/Warehousing, including Wholesale	o		FINAL STOP ADDRESS	
Public & Government, incl. Museums, Schools, Libraries	D			
Office & Sarvices, MCL. Financu Reat Estate, Insurance, Hospitals Othur (Write m)	٥	٥		
			FINAL ODOWETER READING:	
TRUCK TYPE (Check one) Siraight Truck Tractor + Semitravier Tractor + Trailor Boblavi			WHAT IS THE MOST COMMON COMMODITY THAT THIS VEHICLE HAULS?	

· 1 - 1

١

BUSINESS REPLY MAIL

PHOT CLARE PERMIT NO. 12218 CHICAGO, RL.

CHICAGO AREA TRANSPORTATION STUDY

300 WEST ADAMS STREET

12.974

.

. . .

· • • • • • •

CHICAGO, ILLINOIS 40606



Planes recal or staple before sailing

- - · ·

÷

WG PORTAGE WECESBAART IF MAALED WI THE WINTER STATES .

1

4

÷.,

· · · · ·

. . .

APPENDIX IV 1991 Phoenix Urban Truck Travel Survey Form

x

ţ



MARICOPA+ASSOCIATION+OF+GOVERNMENTS Transportation & Planning Office 1730 West Jackson Street

Phoenix Arizona 85007 (602) 255-7867

Dear Truck Owner,

Enclosed you will find a Vehicle Trip Record designed to study commercial vehicle The survey will play an important part in transportation and travel activity. planning for future transportation needs here in the Valley. We are interested in learning more about the day-to-day travel behavior of commercial vehicles in Maricopa We need your help. The vehicle with the license plate number listed on the County. label attached to the survey is the vehicle that should have its travel activity recorded on this Vehicle Trip Record. By having the driver of that vehicle fill out travel log for just one day, you will help us learn more about how to help address Arizona's transportation concerns. We need the travel activity of that vehicle for Tuesday, October 3, 1989.

ABOUT MAG

The Maricopa Association of Governments (MAG) is a voluntary association of local governments that does transportation planning for Maricopa county. Research studies about transportation are conducted so that we can learn more about how to solve traffic problems that affect the Valley. We also maintain computer programs that help us to project future traffic patterns.

ABOUT THE STUDY

This Vehicle Trip Record is being filled out by several hundred commercial vehicle drivers in Maricopa county. Vehicles are randomly selected to participate in the study, and yours is one that has been selected. The information about the use of you vehicle will enable MAG to understand "a day in the life" of this county's fleet o commercial vehicles. We want to know what the vehicle is used for and where it goes We are interested in the vehicle, but we need your help. By filling out the Vehicl Trip Record, your driver will be giving us information that we cannot get anywherelse. Through your record, we will learn more about that vehicle and others like it.

ABOUT THE RESULTS

All information gathered for this study is coming from individuals just like you , drivers who are filling out identical Vehicle Trip Records. All of the information will be treated with the utmost confidentiality. We will use the information only for MAG purposes, to help us prepare future transportation improvement programs. Program that we hope will help you.

You really count to us. Can we count on you? In expectation of getting your he with this study, I thank you. If you have any concerns about participating in t research, please call our Survey Information telephone line at 967-4441.

To proceed, the addressee should complete the first page of the Vehicle Trip Reco and then forward the record to the driver of the appropriate vehicle as indicated Page 1.

Yours appreciatively,

Roger Herzog

MAG Transportation & Planning Office Manager

A Voluntary Association of Local Governments in Maricopa County_

	*** A Personal Message To The Driver Or Drivers ***
If you have any	questions, please call our Survey Information telephone line 967-4441.
Who should we	contact if we have questions about your Vehicle Trip Record?
NAME	
TELEPHON	E NUMBER
	· · · · · · · · · · · · · · · · · · ·
X	X
INS	ERT LABEL HERE
x	x
1. What is the a Thursday after section if street	starting address for the vehicle listed on the label above on the survey date (the first Tuesday, Wednes you received this survey)? Please be specific! Indicate St., Ave., North, South, East or West, nearest address is unknown.
Street Address	
City:	Zip Code:
2. Please look the letter next more trailers, 1	at Figure 1 and determine which vehicle looks most like this vehicle. In the space provided below, we to the picture of the vehicle which looks most like this vehicle. (If this vehicle normally operates with write in the letter of the most common tractor/trailer configuration usually used.)
Letter of	Vehicle From Figure 1:
3. If this vehic gross weight.	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4.
3. If this vehic gross weight. Gross w	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. eight
3. If this vehic gross weight. Gross w 4. How will y the vehicle for	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. eight ou be using the vehicle today? Please circle "yes" or "no" for each item below to tell us whether you we that activity today.
3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Tr	 le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. eight ou be using the vehicle today? Please circle "yes" or "no" for each item below to tell us whether you with that activity today. insportation between home and work
 3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Trabe of the b. Ar 	 le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. eight
3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Tr. b. Ar	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight ou be using the vehicle today? Please circle "yes" or "no" for each item below to tell us whether you that activity today. ansportation between home and work
3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Tr b. An Before you g to be recorder semi-trailer to	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. eight
3. If this vehic, gross weight. Gross w 4. How will y the vehicle for a. Tr b. An Before you g to be recorded semi-trailer to Trip 1	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight
3. If this vehic, gross weight. Gross w 4. How will y the vehicle for a. Tr b. An Before you ge to be recorded semi-trailer to Trip 1 Trip 2	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this if the vehicle has six tires or less, go on to Question 4. reight
3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Tr. b. An Before you g to be recorder semi-trailer to Trip 1 Trip 2 Trip 3	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight
3. If this vehic gross weight. Gross w 4. How will y the vehicle for a. Tr. b. An Before you g to be recorder semi-trailer to Trip 1 Trip 2 Trip 3 Trip 4	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight
3. If this vehic, gross weight. Gross w 4. How will y the vehicle for a. Tr b. An Before you ge to be recorded semi-trailer to Trip 1 Trip 2 Trip 3 Trip 4 Trip 5	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight
3. If this vehic, gross weight. Gross w 4. How will y the vehicle for a. Tr b. An Before you ge to be recorded semi-trailer to Trip 1 Trip 2 Trip 3 Trip 4 Trip 5 Trip 6	le, when used as shown in Figure 1, has more than six tires, please write in an estimate of what you this If the vehicle has six tires or less, go on to Question 4. reight

;

etert	time:	<u> </u>	. :	A.N
start	Odom	eter:		
•				

F

يراسي المستحسب المراسم المراسية الراسية المستحد مستحد المستحد مستحد مستحد المستحد مستحد المستحد والمستحد والمستحد والمستحد والمستحد والمستحد والمستحد والمستحد والمستحد والمستحم ا

Ì

A./P.M. TRAVEL DIARY

H

Please record each trip In the order you make it.

trip •	Start Time	Stop Time .u./r.u.)	Stop Odometer	Name & Address of stop Please give EXACT street address, St. vs Ave., etc.	Zipcode of stop	Activity at stop see below.	Stop ON or OFF street	Land Use at stop ere below.	Vehicie Type	Total # axies
11MP 6	E TRI 12:45	1:15	#5 IN PREV 5082.3	1006 EXAMPLE) 1345 N. 10th Street Warehouse #2, Phoenix	85014	·		3	4	3
1.	AN P.N.				••		ON OFF			
2.		44 PM			_		ON OFF			
8.					-		ON OFF			
4.							ON OFF			-
6.							ON OFF			
6.					-		ON OFF			
7.		AU 98				•	ON OFF	-		
8.							ON OF	F		
9.					-		ON OF	F		
10.							ON OF	F		
if you tO tri in the Write What	r vehic pa duri tetat in your was yo	cio mad ng the number final (ur final	e more than day, write of trips her sdometer read stop addres	0: ling here:		TIVITY AT I. Piek op, Lo I. Grop off, W I. Load & Wal I. Load & Wal I. Sorvice as I. Sorvi	STOP ad alead ad iii toto percent	AND USE AT 1. Rosidestial 8. Rotali 5. Marchovelag 4. Transportati 6. Willitian 8. Rommuniat 7. Noopitale 8. Public/Gow 6. Office/Bar	STOP VE 1 . olo. . olol	HICLE Traight Tractor/ Tractor/ Traitor/ Traitor Bobtatt Common
1A -	the er	id of th	a travel day	for this vehicle, please fold and	roturn the	Vehicle T	rip Record	by mail. Po	stage is (pald.
				Pietse	return to:		••			

D'NEIL ASSOCIATES, INC 412 East Southern Avenue Tempe, AZ 85282



APPENDIX V 1987 New York-New Jersey Truck Commodity Survey Form

44

NTERVEW	(A					TF	RUCK C	COM	MODITY	STUDY				PATE		Aet
LANE		<u> </u>						Y					1	WEATHER REMARKS		
OBBER	VATION				DRIVER	TO BE IN	TERVIEW	ED I		ITY OF LO	AD				OBSERVAT	ION
TRUCK Arleo J J O O O	TRACER TVPE LUIDIN D Canishar D C	be åt må Cäy, Team er and Bå	1104 Berough sie	•'	PACEITY TYPE 1 Truck Tame 2 Wordscore 3 Propriet 6 Tac /RE, 6 Diffee 7 Diffee 7 Diffee 9 Diff	(Cay,1e	jila gali 24 gadi Bista	81-83	PACINITY TYPE 1 Trus Turm 2 Trustane 2 Trustane 3 Trustane 6 Tur 6 Tur 6 Tur 7 Tur 7 Tur 8 Diffe 1 Other 1 Other 1 Diffe	PREQUENCY 1 New Then Drive/Day 2 Daily 3 Dai		*	11 -10	LOAD 17.4 9 Part 3 E-ary 2	NAME ON T	IA 4, E 11
								<u> </u>								
				Τ								 \prod	Γ		· · · · · ·	
				-	•											
				Τ		·		Τ				 Π	Τ			
															<u> </u>	
				Τ				Τ				 Π	Т			
			مر المحيو الدين ا				╺┸━╀━╀╴	-					·			
				Т			·	Т				 Π	Т			
						I_	┛┓┨╼╸┨╼╸┨╴	-		·		 ┟╼╻┠╼┙	┶╌┨			
			Π	Τ				Τ					Π			
												 				·
		· ·														

.

APPENDIX VI 1991 New York-New Jersey Truck Commodity Survey Form

۰,

 $\left[\right]$

1991 Truck O - D/Commodity Interview Form

Facili	ty:		•	Date:	Hour Beginning:	Lane		Interviewer:		
OBSERV	E	ASK						ASK FOR OTHER HALF	of Trip	
No. Of Axles:	Troiler Type:	Commodity Load (Type Freight)		Origin What town did you pick up	Destination What town will you deliver	Trailer Width:	Trailer Length;	Water Crossing:	Commodity: (Type Freight)	
	1. Utility 2. Container		1. Full 2. Portial	Ireight?	treight?			1. Outerbridge 2. Goethals Br. 3. Bayonne Br.		
	3, Flatbed 4, Auto		3. Empty	If Empty: Where was your last stop?	lf : Monhattan (latercertion)			4. Holland Tunnel		
	S. Tenker 6. Double Treifer 7. No Treifer 8. Other		If: Nework (Intersection) Jersey Gty (Intersection) Beyonne (Intersection)	Bronx (Intersection) Brooklyn (Intersection) Queens (Intersection) Staten Island (Intersection)			5. Lincoln Tunnel 6. G. Washington Br. 7. Teppon Zee Br. 8. Other (which ?) 9. No onswer			
				•						
								[.]		

Ē

 e



,

i

.

Ì.

[

and the p

997	TRUCK	CORDON	รักษา -	- PHASE 1

Date: November 18, 1902

Facility: NYS Throney Spring Valley Tolt SOUT DOUND (Eastbound)

1

**********	Lane:
iterviewes P	How Beginning.

OSTIVE		ASK	19. 27 TACK 100	Ladr. Souls Back Clever	And the Section	and the contraction is the	Walter The Michigan	2 Becching	Sector State Sector	Constant State States
lo, of Asles	Trader Type:	Commodity	Load:	Pichup:	Facility Type:	Defivery:	Facily Type	Trailer Wichtig	Water Crossing:	Noute 287;
		What we you carrying?	Are you full?	in what hown A sints did	What type of facility	In what hown & state will	What type of facility		If you cross the Hudson	When Noute 237 is completed
		1	1	you PICK LP that tright?	did you get you	you DELIVER that freight?	will you deliver you		what water crossing	would you use that
			1		height hom? .		#+bH 107		will you use?	connection to get to the
) 1	1	1	}]					Tappan Zee Dridge?
	1. Steight		1. 7.4		1. Warehouse		1. Wwehouse		1. Vermanne-Narowa	1. Yes
	2 1111	I 'perhape' ask	2 Partial		2. FactoryPlant		2, Fattory/Plant		2. Holland Turnel	2, No
	1 Contarian	the muse with	3. Erroly Stan	# Emphra aiact	2. Truck Terminal	# Employ a Supple	3. Truck Terminal		1. Lincoln Turnel	3. Maybe
	4 Fished	f.e. "mincipal	8. Does not longer	to be what he will be a take weat the second	4. Port	in what town & state lo as to 3 fer	4. Part		4. George Whish Br.	4. Not growing the Husbon
	A Auto Carter	waster entered	DLANK No Annual	YOU LAST STOPT	S. Areart	YOU NEXT STOP7	1 Areal		& Tacoan Zee Dr.	5. Not lamine with Rt. 267
	A Tanhar	and have been a little			A Robard		6 Balreed		8. Beer Mauria In	B Does not know
	7 Double Trailer			Ash the near set intersection for:	7. Office	Ask the nearest intersection for:	7. Office		7. Newburgh-Baseron	BLANK Ne Acewar
	B No Trailer	(Driver doesn) brow	1	Norma Brooklan	1. Rebl	Neuert Brooker	R. Reinit		8. Galacite States labored	
	. Chan (mar 2.1	Code W		James City During	9. Other Insection	Joner City Queens	B. Other (specify)		1. Other hours M	
			1	Bergran Brons		Devorat Bront			10. Not crossing	
		I FARTY MAY DEANK		Marketten Sinten Jahred		Manhalina States Manif	Don't Know (BLANK)		No Amount (CLANIC)	
	[[
			1							
	1 [1	Į	ļ	[l I			
	1 1	1				1				
		1						•		
			1							
		1	£ .			1				
		4								•
	1 1	1								
		1			·					
	1 1	1								
	l l	1								
		1	1				ł			-
		}	I							
		1	1							
		1								
			•							
							I			
						•				
		1	1			•	1			
		1	1							
		1	1		•	l •				

Page:_____

1 1

1

÷

INE WILL COLOR INTER - FAULT Text Text Text Text Text Text (maxwer 2) Max Bayes Max Bayes Sect Sect Sect Sect (maxwer 2) Max Bayes Max Bayes Sect						, I ,							Y		
Internet in the second of	.		6		1	1992 TRUC	X CORD	ON STUDY - PI	usei T				1	P	
Control of a	Intervie		Hour Beginning:]	Date: November 17, 1992	Bridge I-	-78	ן ר					rage:j	
Control Control <t< th=""><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th></th><th>E</th><th>A S</th><th>T</th><th>B O</th><th><u>U N</u></th><th>D</th><th></th></t<>									E	A S	T	B O	<u>U N</u>	D	
1. Salyi 2. Salyi 3. Control 4. Total 3. Control 4. Total 5. Control 4. Total 5. Control 4. Total 5. Control 4. Struct 5. Struct	CEISER No. of Aules:	VE. <u>Andre</u> Trailer Type:	ASK 30 Marcanig Au Commodity: What are you carrying?	Coad Are you full?	Cright In what been & state did you PICK LP that is sight?	Deuterations In what bown & state will you DELIVER that height?	Trader Webby	ASK FOR TH Destimitor When you mu In the eastho what town an go to? Point	E EASTBOUND) who deliveries who direction, us state do you 1 to the EastJ>	PORTION OF YOUR T Reverse Commodity: What do you carry eastound?	Cright Cright Where do ye up those los	voleta nupicit de?	When Crossing: If you cross the Hudson, what water grassing would you use?	Third Mile 297: Moule 297: When Route 297 is completed would you use that sonnection to get to the Tappan Zee Bridge?	
Description Desc		1. Shight 2. Unity 3. Container 4. Finbed 5. Ado Carrier 8. Tanker 7. Double Traiter 9. Ma Traiter	F 'purbaga' ask for more into (i.e., 'municipal waste' instand of 'parbage').	1. Full 2. Fartual 3. Emptyl 35. Kg 9. Loss not know BLANK No Answer	R Strategy (A) and them a base to pair LAST STORY 2005) Ask Based	U Extension Process Control of Status of the Process Of Status of the Process of Status of Status of Status Netherland Status of Status of Status		I Driver does / Code ** Ask live reares Nevents	not know, st intersection for ; Brooklan	I 'perbage' ask for more toto (i.e., 'muricipal waste' instead of 'perbage'). Orbus doesn't brow	E Di her doss ; Code P Ask the neares Newark	ot know, t intersection for Brooklyn	1. Verareans-Narows 2. Hofand Ternel 3. Lincoln Ternel 4. George Wash. Br. 5. Tappen Zee Br. 6. Bas Mearthin 7. Nenburgh-Bascon 8. Goly to Shalon Mand	1, Yes 2, No 3, Maybe 4, Not enumbing the Hudson 5, Hol lamiliar with RL 287 8, Dees not know BLAHK No Answer	
		Other (specify)	Code T		Jensey City Beyonne	Breaklyn Ousene Brene		Jersey City Bayerne Manfatteri	Outens Branz Staten Island	Code 11 If EMPTY leave BLAN	Jersey City Beyonne Manhattan	Ourens Brone Staten Island	8. Other (specify) 18. Not crossing No Armour (SLANR)		
					·					•					
												•			
														•	

•

.

APPENDIX VIII 1994 El Paso Commercial Truck Travel Survey Form

The Communications Group Indentification Label: a. Record Type 20 b. Month c. Travel Date Please answer the following questions about your truck or commercial vehicle: What is the primary business of the owner of this truck? 1. Who manufactured this truck? 2. What is the model name of this truck? 3. What is the model year of this truck? 4. 5. What type of fuel does this truck use? Gasoline 1 2 Diesel **D**3 Other_ (please specity)

 $i = i = u \{ i \}$

6. How was this truck configured on the travel day:

ģ

		· · · · · · · · · · · · · · · · · · ·		
	NIT S	2 AXLE, 6 - TIRE		1
	BUCK	3 AXLE		2
	SINC	4 OR MORE AXLE		3
S		3 AXLE SINGLE TRAILER	ê Timuli	4
NCX NCX	st	4 AXLE SINGLE TRAILER		5
4	ATION	5 AXLE SINGLE TRAILER		6
	NIBW	6 OR MORE AXLE SINGLE TRAILER		7
	8	5 OR LESS AXLE MULTI-TRAILER		8
		6 AXLE MULTI-TRAILER		٦ ٩
	and the second data where the second data wh			

7. What is the odometer-reading on this truck? _____ miles

Now, please complete the trip diary for each trip (stop) or delivery made today.

he Communications Group
ecord Type 21
ehicle License Number

City, State, Zip Code

1

1

1-

1

T

5

COMMERCIAL VEHICLE TRIP DIARY

INSTRUCTIONS: Please complete the trip diary for each trip made today. You may substitute a manifest if available. Please include each stop as a separate trip, even if only stopping for lunch.

.

DEPART TIME

PM Noon Mdnt

D AM

D Noon

D Mdnt

1

1

41.747

FACILITIES USED

to get to this stop (check all that apply)

2 Border Freeway 3 Montana Avenue 4 Trans-Mountain Drive 5 Mesa Street

8 North Loop 9 1 375

8 North Loop 9 1 375

1 D I-10 2 D Border Freeway

3 CI Montana Avenue

4 Trans-Mountain Drive

1 6 1 1

101.10

Cargo at Departure 7 Paisano Drive

Cargo at Departure 7 D Paisano Drive

TRAVEL DAY:			, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,, ,	
At 4 AM, this vehicle was at: 1 Your place of business 2				
Name of Place	Kind of Place			
Address or Intersecting Streets				
City State) Zij	o Code		
Departure Time: AM	l Noon I Mdnt			
Beginning Odometer Reading:	Miles			*
Cargo at Departure				
WHERE was this vehicle taken next?	KIND OF PLACE (for example, apparel manufacturer)	PURPOSE of trip (Check one)	ARRIVAL TIME	ODOMETER READING (In Miles)
Name of Place		1 Return to base location 2 Delivery 3 Pick-up 4 Maintenance, fuel, oll, 5 Driver needs (lunch,) 6 Other	☐ AM :_ PM ☐ Noon ' ☐ Mdrt	
City, State, Zip Code	Activity Type			-
Name of Place		1 Return to base location 2 Delivery 3 Pick-up 4 Maintenance, fuel, oil, 5 Driver needs (lunch,) 6 Other	I AM PM Noon Mdnt	
	Activity Type			

	KIND OF PLACE (for example,	PURPOSE of Inp	ARRIVAL TIME	ODOMETER READING	DEPART TIME	FACILITIES USED to get to this stop (check
HERE was this vehicle taken next	apparal manufacturer)	(CHOCK OND)		(in Miles)		
me of Place		1 Return to base location 2 Delivery 3 Pick-up 4 Maintenance, fuel, oil,	PM PM Noon Mdnt		: PM PM Noon . Mdnt	2 Border Freeway 3 Montana Avenue 4 Trans-Mountain Drive
dress or Intersecting Streets		5 Driver needs (lunch,) 6 Other			Cargo at Departure	6 Alameda Avenue 7 Paisano Drive
State Zip Code	Activity Type					9 1 375
y, State, Zip Cooe		1 Return to base location			D AM :_D PM D Noon	1 🖬 I-10 2 🗖 Border Freeway 3 🗖 Montana Avenue
me of Place		3 D Pick-up 4 D Maintenance, fuel, oil, 5 D Driver needs (lunch,)	D Mont		C Mdnt	4 Trans-Mountain Drive 5 Mesa Street 6 Alameda Avenue
dress or Intersecting Streets	Toingity Lung	6 🗖 Other			Cargo at Departure	7 Delisano Drive 8 North Loop
y, State, Zip Code	Activity Type		FT AM		D AM	101-10
me of Place		1 Delivery 3 Delivery 4 DMaintenance, fuel, oil,	:PM Noon Mdnt		:_DPM ' D Noon D Mdnt	2 Border Freeway 3 Montana Avenue 4 Trans-Mountain Drive 5 Mesa Street
dress or Intersecting Streets	Activity Lyne	5 D Driver needs (lunch,) 6 D Other			Cargo at Departure	6 Alameda Avenue 7 Paisano Drive 8 North Loop 9 975
v. State, Zip Code	Activity Type		- H AM		MA D	10 -10
me of Place	-	2 Delivery 3 Pick-up	: PM Noon Mdnt		D PM	2 Border Freeway 3 Montana Avenue 4 Trans-Mountain Drive
dress or Intersecting Streets	•	5 Driver needs (lunch,) 6 Other			Cargo at Departure	5 U Mesa Street 6 U Alameda Avenue 7 U Paisano Drive
	Activity Type	1		•		8 INorth Loop 9 II 375
y, State, Zip Code	1	1 T Return to base location	AM 🗆		D AM	101-10
ime of Place		2 Delivery 3 Delivery 4 Maintenance, fuel, oil,	DPM Noon Mdnt		Noon	3 Montana Avenue 4 Trans-Mountain Drive 5 Mesa Street
dress or Intersecting Streets	- Activity Lyna	6 🖸 Other			Cargo al Departure	7 Palsano Drive 8 North Loop
v. State, Zip Code						

.

APPENDIX IX 1994 North Carolina-Triad Area Truck Survey Form 1__

(1. ...

L

l_

Ł_

[___

ŧ.

TRIAD AREA, NORTH CAROLINA TRUCK SURVEY - ADMINISTRATION DATA Pre Interview On-Site Interview

and the second second

1

Call Date:/	Vehicle #
. Sample Number	License #
. Travel Date	Contact with Driver or Owner
	Date Time By Result
Day Mon _ The _ The _ The _ The	
weets Name	
	·
Registration Address	Remarks
STREET NUMBER STREET NAME	
aty	
	6. Total Trips Reported
BTATE DF CODE	7. Hours of Vehicle Operation
Contact Name	8. Interview Type: (Check one below)
for vehicle information	Trips Possible
Gelephone Number:	Partial Interview
	Refused
Coroning Address (f Diferent)	Cannot Find Owner
•. Caraging Audress ("Durken)	Cannot Find Address
	Garaged Outside Area
STREET NUMBER	No Trips Possible
	Out of Area on Travel Day
	Vehicle Sold or Junked
STATE 2P CODE	9. Interviewer Identification and Certification:
Number of Trucks here:	I hereby certify that the information on this form has been obtained by me from the respondent and is accurate and
Driver/Dispatcher:	complexe.
	OFFERVIEWER'S MONATURE OATE
	anterviewer & Mumber

Part 1 Vehicle Information

Your Travel Day is:

1

ł

Ŧ

1

I

ł

1

T

1

The label above includes some of the information you provided when our interviewer called. Included are your address, type of company, and the number of persons employed in your company. If any of these items are incorrect, please write the correct information directly on the label.

Please fill out the information below about the vehicles owned by or garaged at your company.

				Туре с	Турі	e of Fue	1	Number of Axles	Weight (ibs)				
Vehicle Number	License Plate #	Year	Single Unit Truck	Combination Truck	Pick-up or Panel	Van	Car	Diesei	Gasoline	Other	2, 3, 4, or 5+	Empty	Loaded
1					Make	Make	Make						
2					Make	Make	Make						
3					Make Model	Make Model	Make						
á					Make Model	Make	Make						
5					Meke	Make Model	Make Model						
6					Make Model	Make	Make						
7					Model	Make	Make						
8					Make	Make	Make Model		Ū.				
9					Make Model	Make Model	Make Model						
10					Make	Make Model	Meke Model	: □					

	,) ,	1.])	·	er-man.	
Travel Day: Mon Tues We For Truck Number (Write in n	ed [] Thurs number from co nis a deliver	Fri an Impany Informa Van. T Ye	ad Date:/_ ation form) == NoOT p	assen	 ger van? Y== No			Pa Ti	art rav	2 vel Diary
On this day, was this truck used?	Yes - Conti No - Return What time (Check AM or P or Location (Ind	did you fear M) icate address k	Zip Code		Mileage for the tr Begin Odometer Numb End Odometer Numbe Instructions: Please carry this diary to e Record each trip y e Include the specifi e If you are leaving to for the trip you with	avel day: er with you througho ou make in the on c data requested i he Triad area and 'make and return	ut the der yc or ead will nu this fo	travel day shown above. nu make it. ch trip. ot come back today, pleas rm to the person who gave	e fill ou e it to y	A the information
Location Address	When did you get here/leave here?	Did you turn the engine off at this stop?	Activity at St (check one for eac	top ch trip)	Land Use at Stop (check one for each trip)	If auto or van, # of persons in vehicle including driver.		، if truck, what a (Type c (check one)	re you / Goods for eed	carrying? a) h trip)
First, 1 went to: Name of Place Address or nearest Intersection CityStateZp	Arrive	Yes No	1 Pick-up Load 2 Drop-off Load 3 Service Unit 4 Fuel Unit 5 Personal Busine 6 Return to base 7 Other (Please Sp	ese pecity)	1 Office Bldg. Commercial 2 Retail/Restaurant/Gas Station 3 Warehouse/Manufacturing 4 Residential 5 Port/Transportation Hub 6 Utilities 7 Construction/Gravel/Landfill 8 Other (Piesse Specify)	# of People	[]1 []2 []3 []4	Food or Kindred Products Tabacco, Textiles, Apparel Containers returned empty Mail or Express Traffic/ Smail Packaged Freight Printed Matter Clay, Concrete Glass or Stone Products/Furniture or Fabricated Metal Products/Lumber, Pulp, Paper or Allied Products		Metallic Ores, Coal, Petroleum Natural Gas Farm, Forest or Marine Products Machinery Transportation Equipment Equipment or Supplies Waste or Scrap Material Hazardous Material Fak Other
2 Then, I went to: Name of Place Address or nearest Intersection	Arrive : Depart	. Yee	1 Pick-up Load 2 Drop-off Load 3 Service Unit 4 Fuel Unit 5 Personal Busin 6 Return to base 7 Other	958 056	1 Office Bidg. Commercial 2 Retail/Restaurant/Gas Station 3 Warehouse/Manufacturing 4 Residential 5 Port/Transportation Hub 6 Utilities 7 Construction/Gravel/Landfill 6 Other	# of People	□ 1 □ 2 □ 3 □ 4	Food or Kindred Products Tebecco, Textiles, Apparel Containers returned errpty Mail or Express Trailic/ Smail Packaged Freight Printed Matter Cley, Concrete Glass or Stone Producta/Furniture or Fabricated Metai Brockuted Justice: Data	□7 □8 □9 □10 □11	Metallic Ores, Coal, Petroleum Natural Gas Farm, Forest or Marine Products Machinery Transportation Equipment Equipment or Supplies Waste or Scrap Material Hazardous Material
City State LI	am. p.m.				(Please Specify)			Paper or Altied Products		Other

-

1

.

APPENDIX X 1994 Houston-Galveston Area Council Commercial Vehicle Survey Form

ŧ

Plate Number	License	[-] [-] [-							npany ne: ress: ention:	Con Nan Add Atte	le	elephon Number	Т
-Up ey)	nd Follow Before Sun	Seco (Day			Up vev)	t Follow-	Firs			all	eening C	Scr	
	Respondent Name Respondent Name								ent Name	sponde			
Int Result	Time	Date	Call	Result	Int	Time	Date	Call	Result	Int	Time	Date	Call_
			1		÷	•		1					1
		-	2					2					2. [.]
			3					3					3
2nd Follow-Up ? <u>Comments</u>	mailed back No	DNSES uestionnaire s [] 1	Respo 1. Was qu D Ye	v-Up mments	1st Follov Co	received? No vey mailed? No	DINSES Jestionnaire s second sum s	Respo 1. Was qu 2. Was a Yes	ments	S <u>Co</u> п	esponse: le? No	R to Participat s	Willing Yes

Ē.

~



Houston-Galveston Area Council

Office of the Executive Director

PO Box 22777 • 3555 Timmons • Houston, Texas 77227-2777 • 713/627-3200

Dear Commercial Vehicle Survey Participant:

Thank you for agreeing to participate in the Houston-Galveston area one-day travel survey. The Houston-Galveston Area Council, assisted by Wilbur Smith Associates, is conducting this survey to help plan for future transportation needs in our area. Your vehicle(s) was randomly selected in the sample of commercial vehicles registered in the Houston-Galveston area, and we need your assistance.

We are enclosing a copy of the one-day travel survey form for your or your truck driver's use in identifying trip information. We would like the truck driver to complete the form for all travel performed by the selected vehicle on the designated date. It is really quite simple and will assist us in studying the way commercial vehicles get around the Houston-Galveston area.

- i -

It is important that you complete and return this form even if no trips were made in the Houston-Galveston area on the selected date, as we need to know this information to estimate overall travel characteristics. Please note that this information will be treated confidentially, and will only be used in summary form.

Your cooperation and assistance in providing information for this important survey is greatly appreciated. If you have any questions, please call Wayne Holcombe at 465-7800. Thank you.

Jack Steele

RECEIVED

NOV 1 1 1994

COMMERCIAL PASSENGER CARRIER SURVE

ATTENTION PASSENGER CARRIER DRIVER!

Here is your chance to participate in decisions about highway and transportation improvements ir the Houston-Galveston area. Roadway modifications will be based upon your survey answers. This survey applies to every trip made in your passenger carrier on the designated travel day.

Thank you for your participation!

PART 1 VEHICLE INFORMATION

1) Record Type: <u>23</u>	- ·	· •
2) Day of Travel:	Month	Day
3) Company Name:		
4) Address:		
0FFICE USE ONLY 5) Census Tract Nu	mber:	6):Traffic:Serial:Zone Number:
7) License Plate Numb 8) Make of Vehicle:	er:	9) Model of Vehicle:
10) Vehicle Year:		
11) Vehicle Fuel Type:	1. Unleaded Gasoline	4. Propane
	2. Unleaded Gasoline3. Diesel	5. Compressed Natrual Gas 6. Other
12) Maximum number of passengers vehicle can carry:	· · · · · · · · · · · · · · · · · · ·	

The next part of the survey asks that you record the following information after each trip made by this passenger carrier vehicle on the designated travel day. A trip is any time the vehicle stops and you or someone else gets into or out of the vehicle.

COMMERCIAL PASSENGER CARRIER SURVEY

OFFICE USE ONLY

ŗ

March Million of the	2 Sa 2 A
Record Type:	
Constantina in the	
LICENSE Plate:	the second s

PART 2 TRIP DIARY

00 My Trip Began At: Location: Place, Address, or nearest intersection, - City/State/ZIP 01 Then I Went To: Location: Place, Address, or nearest intersection, City/State/ZIP	CERSUS TRACT	Ocometer Reading Declarure Time am pm Colometer Reading Arrival Time am pm Declarure Time am	Number of Passangers	Please indicate the purpose of this tro; 1. Base Locason / Return to Base Location 2. Delivery 3. Pick-L0 4. Vehicle Maintenance (kuel, ed., etc.) 5. Driver Needs Sunch, etc.) 6. Other Please inscale the purpose of this tro; 1. Base Lacation / Return to Base Lacation 2. Delivery 3. Pick-L0 4. Vehicle Maintenance (tuel, etc.) 5. Driver Needs (kinch, etc.) 5. Driver Needs (kinch, etc.) 5. Driver Needs (kinch, etc.)	Please destrice """ 1. Residential 2. Educational 2. Educational 3. Industrial 4. Government
02 Then I Went To: Locaton: Place. Address, or nearest intersection. City/State/ZIP	OFRCE USE ONLY CENSUS TRACT	Arrival Time_ Depanure Time_ am pm Depanure Time_ am pm	Number of Passengers	Please macate the purpose of the tro: 1. Base Location / Return to Base Location 2. Delivery 3. Pick-Up 4. Vehicle Maintenance (kuel, od. etc.) 5. Driver Needs (kunch, etc.) 6. Other	Please emprove
03 Then I Went To: Location: Place, Address, or nearest intersection, City/State/ZIP	OFFICE USE ONLY CENSUS TRACT	Odometer Reading Arrival Time am pm Departure Time am pm	Number of Passengers	Please inocate the purpose of this tro; 1. Base Location / Return to Base Location 2. Detreny 3. Pick-Uo 4. Vehicle Maintenance (fuer, oil, etc.) 5. Driver Needs (funch, etc.) 6. Other	Please sectors It is location:], Resperiul 2. Educational], Resperiul 3. Industinal 4. Government 5. Meocat 6. Ottice], Retari 4. Other
04 Then I Went To: Location: Place, Address, or nearest intersection, City/Smle/ZIP	CENSUS TRACT	Odometer Reading Arrival Time am pm Departure Time am pm	Number of Passengers	Please indicate the purpose of this ino: 1. Base Location / Return to Base Location 2. Delivery 3. Pick-Up 4. Vehicle Mainténanos (fuel, oil, elc.) 5. Driver Needs (funch, etc.) 6. Other	Please describe We location. 1. Respectual 2. Educational 3. Industriat 4. Government 5. Metocal 6. Office 7. Retail 8. Office
05 Then I Went To: Location: Place, Address, or nearest intersection, City/State/ZIP	CERSUS TRACT	Odometer Reading Arrival Time am pm Decarture Time am pm	Number of Passengers	Please indicate the purpose of this tro: 1. Base Location / Return to Base Location 2. Delivery 3. Pick-Up 4. Vehicle Maintenance (luel, oil, etc.) 5. Driver Needs (lunch, etc.)	Please descrice this location: 1. Residential 2. Educational 3. Industrial 4. Government 5. Medica 6. Office 7. Retai 8. Other

APPENDIX XI 1988 Ontario Commercial Vehicle Survey Form

 $\left[\right]$

-

Ţ

1

1

: ! 1~

r [

QUESTIONNAIRE FOR 1993 CVS SURVEY

COMBINATION CONFIGURATION & AXLE SETTINGS				
Image: Stranger in the image: Strange				
COMPANY NAME(S) }				
VEHICLE & COMPANY INFORMATION Reg. Gross Weight: IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII				
Dispetch Is In:City/Town Prov/State Freight Terminal There?				
For Hire This Trip? Same For All Trips? Company Speed Limit? Km/hr Is This Truck Assigned Exclusively To You? Do You Have A Co-Driver On This Trip?				
Where Did You Start This Shift?ON ON				
Where Will You End This Shift? Same Place: orON				
FREIGHT DATA units				
How Many Shipments On-Board?				
How Full (volume)? [Mill 124] [74] [74] [74] [74] [74] [74] [74] [7				
City/Town Prov/State Where Did You Receive That Cargo?				
City/Town Prov/State What is That Cargo? Mixed Freight Auto Parts Vehicles Food Household Goods Wood Products Machinery Textiles Livestock Metais/Minerais Manufactured Goods Chemicais Petroleum Waste Grain/Feed Aggregates Equipment Furniture Other				

-

.

.

MINISTRY OF TRANSPORTATION, ONTARIO

QUESTIONNAIRE FOR 1993 CVS SURVEY (Continued)

J.

ſ

[...]

(,---,1

L, --- , J

[]

Ū

Π

÷

· · · · · · ·

1

VEHICLE TRIP Is this trip: Locat: Peddle Run: Line Haul:
IF TRACTOR/TRAILER: Where did you pick up this trailer? ON ON Prov/State
Where will you drop this trailer? City/Town Prov/State
If Empty, go to next section is this the truck/trailer the cargo was loaded onto when picked up?
If not, where was the cargo loaded on this truck/trailer? ON ON Prov/State
Akport 🔲 Railhead 🔲 Port 💭 Terminal 💭 Factory 💭 Producer 🗋 Warehouse 🛄 Store 🗔 Other 🗔
Will this truck/trailer carry the cargo to its final destination?
If not, where will the cargo be unloaded from this truck/trailer?ONCity/Town Prov/State
Airport 🗌 Rallhead 🗋 Port 🗋 Terminal 🗌 Factory 🗋 Producer 🗋 Warehouse 🗋 Store 🗖 Other 🗖
DRIVER TRIP
Where Did You Start This Trip ON ON
Where Will You End This Trip? Same Place: Or ON ON Prov/State
Furthest Point Travelled To During Trip: ON Prov/State
Where did you first turn on to this highway on this trip?
Where will you finally turn off this highway on this trip?
DRIVER PROFILE Do you have automatic A or manual M slack adjusters? Are you allowed to adjust them?
When was the last time you attended a driver safety meeting or training course?
What was the subject discussed?
What are are your instructions in the event of a split?
Are you a: Company driver?
Owner-Operator: L.J. If Yes, are you contracted to a carrier? L.J
If No, did you obtain this load through a load broker?
Do you belong to a: Union:
How many years have you been driving trucks?
Are you exempt from keeping a log?
How many on duty hours so far this shift?
How many on duty hours in past 7 days?
Does the truck have seatbelts?
Driver Ucense Number?
THANK YOU FOR YOUR HELP TODAY
Male Female Age Range: 20's 30's 40's 50's 60's

MINISTRY OF TRANSPORTATION, ONTARIO

EXIT RAMP SURVEY QUESTIONNAIRE

]

J

]

]

]

COMBINATION & AXLE SETTINGS Power Doity Trader Doity Trader STRAIGHT A.C TRAILER #1 A.B.C TRAILER #2 ITRACTOR FULL TRAILER IFULL TRAILER ITRAILER AXLES ITRACTOR FULL TRAILER IFULL TRAILER UFT7 III IIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIIII				
COMPANY NAME(S) }				
ON DRIVER DOOH } FREIGHT DATA Where Is Your Next Delivery? ON City/Town Prov/State Where Did You Receive That Cargo? ON City/Town Prov/State What Is That Cargo? Mixed Freight Hoursehold Goods Wood Parts Wood Participation Testilize				
Metals/Minerals Manufactured Goods Chemicals Petroleum Waste Grain/Feed Aggregates Equipment Furniture Other Where was the cargo loaded on this truck/trailer? ON City/Town Prov/State Where will the cargo be unloaded from this truck/trailer? ON ON				
DRIVER TRIP Where Did You Start This Trip ON ON City/Town Prov/State				
Where Will You End This Trip? Same Place: orON				
City/Town Prov/State Where did you first turn on to this highway on this trip?				
THANK YOU FOR YOUR HELP TODAY				
Male Female Age Range: 20's 30's 40's 50's 60's				

MINISTRY OF TRANSPORTATION, ONTARIO

ţ

-

APPENDIX XII 1986 Caltrans Bay Bridge Truck Travel Survey Questionnaires

- 04

 $\left(\right)$

ſ

C

[_____]

Ĺ

``
SAN FRANCISCO-OAKLAND BAY BRIDGE TRUCK TRAFFIC SURVEY

The California Department of Transportation is conducting a study to evaluate the possibility of improving truck access to the San Francisco-Oakland Bay Bridge. Please answer the questions below. Indicate answers on the detachable card at right. Thank you for your cooperation.

١

1

1

1

1

I

1. Where did your trip start? (last pickup or delivery)

2. What route did you use to reach the San Francisco-Oakland Bay Bridge?

- 3. Where will this trip end? (next pickup or delivery)
- 4. Will you use the San Francisco-Oakland Bay Bridge to return?
- 5. How many axles on truck or truck/trailer combination?
- 6. How often do you cross the San Francisco-Oakland Bay Bridge?

1

I

7. Do you use a credit card for payment of tolls?

8. Truck Body type?

9. Home base:

. .

10. Truck ownership:

11. Any comments?

ons.		SAN FRANCISCO-OAKLAND BAY TRUCK TRAFFIC SURVE	BRIDGE , Y							
	i	Print answers on this card. Drop card in any U.S. mail box. (No postage required).								
	i	1. Indicate zone on map	~~~~~~~~~~~~							
		 2. Route 80 (Eastshore Freeway) Route 880(17) (Nimitz Freeway) Route 580 (MacArthur Freeway) Route 24 (Grove-Shafter Freeway) 	· · · · · · · · · · · · · · · · · · ·							
	I I I	3. Indicate zone on map	•							
		4. □ Yes □ No								
		5·								
	ch and Ma	6. Less than once per month 1 to 3 times per month 1 or 2 times per week 3 or more times per week								
'!	(Deta	7. □Yes □No								
• • •		8. Ocean container Full Empty Container chassis only Flatbed Tank/Bulk liquid Van Tractor only Other, specify:								
		9. California								
		10. Operator owned Firm owned	;							
		11	·							
	i									
	I.									

PORT	DF_	5/1	FR	Anci	500
TRUCH	5 7	RAFF	Ю	SUR	VEY

ŗ

,

e4

i e a

	where did your trip start? (lest pickup or delivery) Zone :
• 1.	what mute did you use to reach the Port of San Francisco?
Ζ.	In Route 80 (Bay Bridge)
3.	How many axles on truck or truck/trailer combination ?
4.	Truck Body type : 0 Ocean container 0 Full 0 Empty 0 Container chassis only 0 Flatbed 0 Yan 0 Tractor only 0 Other, specify :
5	. Destination after leaving Port? (next pickup or delivery) Zone :
6	. What route will you use after leaving the Port of San Francisco?
	D Route 80 (Bay Bridge)
-	7. How many axles on truck or truck/trailer combination ?
1	B. Truck Body type : Ocean container DFull DEmpty D Container chassis only DFlatbed DYan DTractor only Other, specify :
	9. How often do you travel to the Port of San Francisco?
	liess than once per month 1 or 2 times per week 1 to 3 times per month 1 to 3 times per month
	10. Home base : 🛛 California 🛛 Out of state
	11. Truck ownership : 🛛 Operator owned 🛛 Firm owned
	Comments :
	Date : PORT QUESTIONNAIRE

Bother Cocon Participation of the source of	Retmond Here and a service of the s	All BOY SUISUN SUISU	ETTERS HIDICATE ZONES EAVY LINES INDICATE DNE BOUNDARIES
	د منه من بين من رمان من من من من من من من من من رم د		د می مو می بعد دو مد ^م در می شد می مد می مد می مد می مد مد مد مد م
			NO POSTAGE NECESSARY IF MAILED IN THE UNITED STATES
BUSINESS REPLY MAIL FIRST CLASS-PERMIT NO. 15205-SAN F	RANCISCO, CA		
POSTAGE WILL BE PAID BY ADDRESSEE			
State of California DEPARTMENT OF TRANSPOR Highway Operations Branch District 4 P.O. Box 7310 San Francisco, CA 94120	TATION		

APPENDIX XIII 1991 Caltrans-Alameda County Truck Intercept Survey and Classification Count Forms

÷

ſ.

Γ

ſ

[

nđ

1.1

.

Truck FMPTY or FULL (obride and)	What city are you COMING FROM? (decis if the la where the truck is GARAGED, was LART LOADED or was LART UN OADED)	What city are you GOING TO? (of de if this is where the trick is OARAGED, will be NEXT LOADED & MEXT UNCOADED)	Where le this truck QABAGED7 (this only if reaclourly answer)	What TYPE OF GOODS are you hauting? (rinde if EMPTY)	Number of AXLES (circle energy)
1 1 Empty (2. Juli	Sur- Jr Jr City (1.) Gereged 3. Lest Unloaded	Scale projecto Criv 1. Garagad State 3. Next Unloaded	King Oren	Computer Chips Inputer Chips Emply	2 6 3 7 4 8 (5) 9
2 1 Empty (2.) uil	(1) Garagad Stata 3. Last Unloaded	Livermore City 1. Garagant 3. Noxt Unloaded	Cilcoy Civ	Jarlee Internity	2 6 (3) 7 4 8 5 9
3 (T)Empty 2. Full	(1.) Garaged State 2. Last Londed	City City NV 1. Gereged 3. Next Unloaded	Migutan City Store	(epecify)	2 6 3 <u>(</u>) 4 8 5 9
/ 4 1. Empty (2) Full	Stere 1. Gereged (2) Lest Loeded 3. Leet Unloeded	<u>SI Lorun</u> City <u>MU</u> 1. Geraged 2. Next Loaded 3. Next Unloaded	Monderie, Jeity (A Bioto	Auglanc suits 10000141 Emply	2 6 3 7 4 8 5 (5)

Ŀ	a manager the and the		• ~•≯.	AST A MARKET AND AND TRUCK CLASSIFICATION FORM				a the form of the second s		
	Time Period	2	3	4	5	6	7	8		TOTAL
ļ	<u>xo - 16 .</u>									
	15 • 30									
	90 · 45	····								
Ļ	15 - 00	<u> </u>				en an				
	0 - 15									
	5 - 30									
<u>_</u>	0 - 45					<u></u>				
Ŀ	5 • 00					a de la companya de				
•	0 - 15									
	5 · 30			-						D <u></u>
	0 • 45						[
	5 • 00		- 0							
, x	D - 15									
15	i - 30							·		

-

-

``

.

APPENDIX XIV PDA Comparisons and Tandy Zoomer [.]

	Brand of Personal Digital Assistant						
Criteria	Tandy Zoomer	HP 95LX	Apple Newton	Sharp Wizard			
Approx. Price	\$599	\$580	\$650	\$480			
Form Factor	Single Piece	Clamshell	Single Plece	Clamshell			
Touch screen	Yës	No	Yes	Yes			
RS 232	Yesi	Yest	Yes	Yes (Option)			
Character Recognition	Yes	No	Yes	No			
PC Compatible Processor	Yes	Yes	No	No ·			
Operating System	Geos	MS:DOS	Subset of Mac System	Proprietary			
SDK Price	\$394 (withDoc)	Free	Unknown	No			
Ease of Programming	Moderate	Easy	Difficult	Difficult			
Emulation Available	Yes	Yes	No	No			
Display Character Size	Big	Small	Big	Small			
Keyboard	Νο	Yəs	No	Yes			
Expansion	PCMCIA	PCMCIA	PEMCIA	IC Card			
Preference	1	2 .	3	4			

1 |<u>|</u> | | | + +

ŧ

STREET

DEVELOPMENT OF NOTEBOOK COMPUTER SOFTWARE



Project:

Development of Notebook Computer Software to Collect Trip Generation Data at Intermodal Facilities.

Client: Federal Highway Administration

Description

of Services: Evaluated existing technology to select most appropriate equipment for data collection use. Developed data collection methodolgy including software for data collection, data aggregation and data processing. Used handheld computer data collection methodology in three case studies to collect truck origin/destination and commodity information.

Project: The overall project was developed to test the feasibility of using computerized data collection for collecting trip generation data at intermodal facilities.

_STREET === === SMARTς

APPENDIX XV Report Contact List

- H

.

f**• **j**

ſ.

CONTACT LIST

ł

.uì

•4

American Trucking Association Dan Murray

(612) 641-6162

Barton Aschman Associates, Inc.

Nancy McGuckin, Senior Associate 1133 15th St. NW Washington, D.C. 20005-2701 (202) 775-6051

California Public Utilities Commission

Jack Fulcher, Public Utility Regulatory Program Specialist Transportation Division 505 Van Ness Ave. San Francisco, CA 94102 (415) 703-1890

Cambridge Systematics, Inc.

Earl Ruiter, Principal 150 Cambridge Park Dr. #4000 Cambridge, MA 02140 (617) 354-0167

Chicago Area Transportation Study

Gerald Rawling, Director of Operations Analysis 300 West Adams St. Chicago, IL 60606-5101 (312) 793-3467

Houston-Galveston Area Council

Keith Hall P.O. Box 22777 Houston, TX 77227-2777 (713) 993-2441

Intermodal Association of North America

Michael Arendes (301) 864-4160

Maricopa Association of Governments

Cathy Arthur or Mark Schlappi Transportation and Planning Office 2901 W. Durango St. Phoenix, AZ 85009 (602) 506-4117

Ontario Ministry of Transportation

Murray McLeod or Greg Little (416) 235-4077 or 235-3617

Port Authority of New York and New Jersey

Ron Taste Office of Transportation Planning One World Trade Center 64E New York, NY 10048-0001 (212) 435-4535

South East Michigan Council of Governments (SEMCOG)

лł

Carmine Palombo or Adiele Nwankwo 660 Plaza Dr. Ste. 1900 Detroit, MI 48226 (313) 961-4266

Street Smarts

Marsha Anderson, President 3400 McClure Bridge Road Building G, Ste. C Duluth, GA 30136-3281 (404) 813-0882

Texas Department of Transportation

Zac Graham (512) 467-3812

Texas Transportation Institute

David Pearson (409) 845-9933

Transmode Consultants, Inc.

Paul Roberts 3400 International Dr. N.W. Ste. 2K Washington, D.C. 20008 (202) 363-2954

Trucking Research Institute

Clyde Woodle, Executive Director 2200 Mill Road Alexandria, VA 22314-4677

University of Michigan Transportation Research Institute (UMTRI)

Daniel Blower 2901 Baxter Road Ann Arbor, MI 48109-2150 (313) 764-0248

Greater Vancouver Regional District

Karoly Krajczar Strategic Planning (605) 432-6367 (605) 436-6970 FAX Krajczar@sp.gvrd.bc.ca